# Removal of Imperfections in Digital Scans using Generative Adversarial Networks

Bachelor's Thesis of

Mohamad Algoabra

At the Department of Informatics and Electrical Engineering University of Rostock

Matrikel-Nr.:219204154Birthday:18. October 1998First reviewer:Dr.-Ing. Hannes GrunertSecond reviewer:Dr.-Ing. Holger Meyer

24. October 2022 - 13. March 2023

Faculty of Computer Science and Electrical Engineering Institute of Computer Science University of Rostock 18059 Rostock

I declare that I have developed and written the enclosed thesis completely by myself, and have not used sources or means without declaration in the text.

. . . . . . . . .

Rostock, 13.03.2023

. .

 $\mathbb{A}$ Ý

(Mohamad Algoabra)

# Abstract

Document digitization is becoming increasingly important as organizations and institutions move towards a more digital workflow. However, scanned document images often suffer from damage caused by different real-life scenarios, which makes them difficult to read and use. In this thesis, we explore the use of generative adversarial networks (GANs) to enhance scanned images and remove imperfections, such as coffee stains and other distortion factors. The problem is formulated as an image-to-image translation task between two domains, and we compare the performance of two GAN types: Pix2Pix, a supervised image-to-image translation model that uses paired data, and CycleGAN, an unsupervised image-to-image translation model that uses unpaired data.

To address this problem, we developed a data pipeline to generate appropriate data to train the aforementioned models. Furthermore, we developed a prototype that allows users to easily test out these models. The effectiveness of the proposed methods was evaluated in detail using various criteria, such as Peak Signal-to-Noise Ratio (PSNR), Structural Similarity Index Measure (SSIM), and Fréchet Inception Distance (FID). We also assessed the impact of these approaches on improving Optical Character Recognition (OCR) efficiency. The results showed that Pix2Pix can significantly improve the quality of scanned images and remove defects such as coffee stains, while CycleGAN performed averagely.

Overall, this study provides a perspective on improving the digitization process by using GANs to address image imperfections. The data pipeline and prototype developed in this work can be used to improve the quality of scanned images and facilitate the transition to a more digital workflow. Future research could explore the possibility of further improving the performance of this method by incorporating other types of GANs or alternative approaches.

**Keywords:** Document enhancement, Degraded document binarization, Scanned images Cleaning, Generative Adversarial Networks, Pix2Pix, CycleGAN

# Acknowledgment

I would like to express my appreciation to my parents for their unwavering support, encouragement, and patience during this period. Their love, understanding, and moral support were indispensable to keeping me motivated and focused on my goals. Their belief in me never wavered, and I am grateful for the sacrifices they made for me.

I would also like to extend my sincere gratitude to my supervisor Dr. Ing. Hannes Grunert, for his invaluable guidance, support, and cooperation throughout my thesis writing journey. His expertise and willingness to share his knowledge have been instrumental in shaping my research and helping me achieve my academic goal. I am also deeply grateful to Daniel Pokrandt for providing me with the necessary equipment and resources to conduct my experiments, which greatly contributed to my research.

Once again, thank you to my supervisor and my parents for their invaluable contributions to my thesis writing.

# Contents

Ab	Abstract i									
1.	Intro	oduction	1							
	1.1.	Motivation	1							
	1.2.	Problem Formulation	2							
	1.3.	Objectives of the Thesis	2							
	1.4.	Structure of the Thesis	3							
2.	Theo	Theoretical Foundation 5								
	2.1.	Artificial Neural Network	5							
		2.1.1. Activation Functions	6							
		2.1.2. Loss Functions	7							
		2.1.3. Optimizers	8							
		2.1.4. Normalization	9							
	2.2.	Convolutional Neural Network	10							
		2.2.1. VGG and VGG Loss	11							
		2.2.2. ResNet and Residual Blocks	12							
	2.3.	Autoencoders	13							
		2.3.1. Convolutional Autoencoders	14							
		2.3.2. U-Net	14							
		2.3.3. Variational Autoencoders	15							
	2.4.	Generative Adversarial Network	16							
		2.4.1. Generative Adversarial Network Training	17							
		2.4.2. GAN Loss Functions	17							
		2.4.3. Conditional Generative Adversarial Network	20							
	2.5.	Image-to-Image Translation	21							
3.	l iter	rature Review	23							
	3 1	Search Process	23							
	3.2	The Results	25							
	3.3.	Selected Approaches	29							
Д	Moth	ods	21							
-10	4 1	General remarks	31							
	1.1.	4.1.1 Hardware environment	31							
		4.1.2 Software environment	33							
	42	Piy2Piy	35							
	1.4.	4.2.1. Generator architecture	35							

		4.2.2.	Discriminator architecture		36
		4.2.3.	Loss function	3	38
		4.2.4.	Implementation	3	38
	4.3.	Cycle	GAN	4	44
		4.3.1.	Generator architecture	4	45
		4.3.2.	Discriminator architecture	4	45
		4.3.3.	Loss function	4	46
		4.3.4.	Implementation	4	47
	4.4.	Data P	Preparation	5	53
		4.4.1.	Raw Data Creation:	5	53
		4.4.2.	Data Pipeline	5	54
		4.4.3.	Data Preprocessing	5	57
	4.5.	Prototy	ype	5	58
		4.5.1.	The Sidebar	5	58
		4.5.2.	The Main Viewing Area	6	50
	4.6.	System	n Architecture	6	51
5.	Eval	uation s	setup	e	63
	5.1.	Testing	g Dataset	6	63
	5.2.	Ouanti	titative Evaluation Metrics		64
	5.3.	ÕCR E	Evaluation Metrics	6	66
	5.4.	Models	ls and Training details	6	67
6	Fyne	riments	s and Results	ç	69
0.	6 1	Experi	iments scenario	é	69
	6.2	Results	۲۹		69
	0.2.	Result			57
7.	Cond	lusion a	and Future work	7	79
	7.1.	Conclu	usion	7	79
	7.2.	Future	e work	8	80
A.	Test	Cases Aj	Appendix	8	39
в.	Resu	ılts Appe	endix	12	29
c.	Code	e Appeno	dix	14	49

# **List of Figures**

1.1.	Examples of the documents	1
2.1.	The different elements of the neuron	5
2.2.	A multi-layer network that has an input layer, two hidden layers, and an output	
	layer	6
2.3.	How the convolution layers work	10
2.4.	The architecture of the Residual Block	12
2.5.	The architecture of Autoencoders	13
2.6.	The architecture of Unet	14
2.7.	The architecture of GAN	16
2.8.	The architecture of Conditional GAN	20
2.9.	Some examples for I2I translation applications	21
4.1.	The architecture of Pix2Pix	35
4.2.	Pix2Pix U-Net Generator	36
4.3.	Pix2Pix Discriminator (PatchGAN)	37
4.4.	The architecture of CycleGAN	44
4.5.	CycleGAN Generator	45
4.6.	Cycle consistency loss	46
4.7.	Examples for the raw data	53
4.8.	The default data pipeline	54
4.9.	The default data pipeline	55
4.10.	Examples for Scanning Effects	55
4.11.	Examples for degraded paper textures and coffee stains	56
4.12.	Examples for the default pipeline outputs	56
4.13.	Prototype Interface	58
4.14.	Prototype Sidebar File uploader and Model selector	59
4.15.	Prototype Sidebar Automatic adjustments	59
4.16.	Prototype Sidebar Manual adjustments	60
4.17.	Prototype Main Viewing Area	60
4.18.	System Architecture	61
61	PSNR line chart (higher is better) for test case 1	71
6.2	SSIM line chart (higher is better) for test case 1	72
63	FID line chart (lower is better) for test case 1	73
6.4.	Test case 1	73
Λ 1	DSNID line abort (higher is better) for test acces	00
A.I.	r SINK line chart (higher is better) for test case 2	90
A.2.	Souve fine chart (fingher is better) for test case 2	91

FID line chart (lower is better) for test case 2	91
Test case 2	92
PSNR line charts (higher is better) for test case 3	98
SSIM line charts (higher is better) for test case 3	99
FID line charts (lower is better) for test case 3	100
Test case 3	100
PSNR line charts (higher is better) for test case 4	108
SSIM line charts (higher is better) for test case 4	108
FID line charts (higher is better) for test case 4	109
Test case 4	110
PSNR line charts (higher is better) for test case 5	116
SSIM line charts (higher is better) for test case 5	116
FID line charts (higher is better) for test case 5	117
Test case 5	118
PSNR line charts (higher is better) for test case 6	124
SSIM line charts (higher is better) for test case 6	124
FID line charts (higher is better) for test case 6	124
Test case 6	125
	FID line chart (lower is better) for test case 2

# 1. Introduction

This chapter will reveal the motives and objectives of the project along with the formulation of the main problem that will be addressed in this thesis.

## 1.1. Motivation

Digitization has become an intertwined part of all aspects of modern life, especially in industry, because the outcome of digitalization is higher process efficiency, lower transaction costs, and better control [1]. Today, all sectors depend on automating most data-related processes, such as financial market or patient data. However, these data are not always present in digital form and can be easily processed because documents and paper transactions still play a major role in various fields. There are also a huge number of ancient documents that must be archived and preserved for future generations. These documents are often damaged for many reasons, such as scratches, wrinkles, dust, coffee stains, moisture, and other real-life scenarios. In addition, damage can appear in digital scans due to poor digitization conditions, such as using smartphone cameras that usually show several types of blur or shadow. Furthermore, some digital scans may contain watermarks, stamps, or other annotations [2] that cause misreading or difficulty in reading their contents, whether on humans or optical character recognition (OCR) systems that play an important role in digitizing the world. Therefore, the automatic processing of digital scans to convert them into a form that can be understood by the OCR system or by humans plays a very important role in the digital archiving of documents in all fields.



Figure 1.1.: Examples of the documents

## **1.2.** Problem Formulation

Once we recognize the significance of properly archiving documents for future generations and expediting the processing of digital scans, which constitutes a significant portion of our daily activities, it becomes necessary to consider methods for enhancing the quality of these documents to avoid potential issues. However, addressing this matter manually can be impractical in many cases given the vast number of documents and scanned images involved. Therefore, we must seek an automated solution that is highly efficient. In this thesis, we endeavored to find such a solution, which we formulated as an Image-to-Image Translation problem, where the objective is to take images from one domain and convert them in such a way that they have the style and characteristics of images from another domain. Specifically, we aim to transform the damaged scanned images, defined as the source domain, into clean images, defined as the target domain. To achieve this goal, we propose the use of Generative Adversarial Networks (GANs) as the core technology for the solution. GANs are a type of deep neural networks that can learn to generate realistic images from a given input distribution.

### **Research question**

The research question we seek to answer in this thesis is:

#### **Research** question

What is the potential effectiveness of using Generative Adversarial Networks (GANs) to enhance the quality of scanned documents that have been damaged or contain imperfections while preserving their content? Can this methodology be considered a practical preprocessing technique for preparing documents for digital archiving?

To address this question, we will utilize and implement several advanced GAN-based image-toimage translation models, and evaluate their efficacy by means of several quantitative metrics. Furthermore, we will investigate the potential of these models in enhancing the accuracy of text extraction from OCR models. We also intend to conduct several test cases utilizing different hyperparameters to compare the efficiency of these models and determine the one that yields the best results.

# 1.3. Objectives of the Thesis

The aim of this study is to investigate the potential of Generative Adversarial Networks (GANs) for restoring damaged digitally scanned documents to their original state or, more precisely, enhance their quality for digital preservation. To accomplish this objective, the following steps were taken:

• Development of a scalable pipeline to generate appropriate datasets for training, validation, and testing of the models. The pipeline will produce both paired and unpaired datasets with varying types of damage and imperfections to ensure that the models are robust enough to handle a wide range of scenarios.

- Development of a framework for restoring damaged documents utilizing state-of-the-art GAN architectures specifically designed for image-to-image translation problems.
- Evaluation and comparison of the performance of multiple GAN architectures to enhance the quality of digital scanned documents in various scenarios. We will use various metrics to measure the performance of these architectures, which will enable us to identify the most effective ones for different types of digital scanned documents and scenarios.

# 1.4. Structure of the Thesis

This section aims to provide readers with a concise overview of the organization and content of each chapter in this thesis. By presenting a comprehensive summary of the structure and substance of each chapter, we hope to enhance readers' comprehension of the research material and facilitate easier navigation through it. Ultimately, our goal is to enable readers to better comprehend the thesis as a whole.

**Chapter 2** which is the Theoretical Foundation, will offer a brief overview of the theoretical background related to this work. Specifically, this chapter will focus on deep learning concepts, generative models, and the image-to-image translation problem. We will provide a detailed explanation of generative models, particularly GANs, and how they can be applied to solve image-to-image translation problems.

**Chapter 3** which is the Literature Review, will offer a comprehensive overview of the research methodology we used, the relevant work and state of the art in the image-to-image translation field. We will conduct a thorough survey of recent research in this field, examining the various techniques and algorithms that have been proposed. Furthermore, we will provide a brief explanation of some of the most promising solutions, outlining their respective strengths and weaknesses.

**Chapter 4** which is Methods, we will present the models and methods that we used in this work. We will provide a detailed explanation of how these methods work and all the technical matters related to the development of the solutions. This chapter will be particularly useful for readers who are interested in understanding the technical details of the models.

**Chapter 5** which is Evaluation Setup, will provide a detailed description of how we evaluated the models. We will explain the testing datasets, evaluation matrices, and model hyperparameters used in the evaluation process. We will also provide an explanation of the rationale behind our evaluation methodology.

**Chapter 6** which is Experiments and Results, will provide a comprehensive comparison and evaluation of all the results obtained for each model. We will present the results of our experiments in a clear and concise manner, making it easy for readers to understand the relative performance of each model.

Finally, in **Chapter 7** which is Conclusion, we will provide a summary and conclusion of the thesis. We will discuss what we achieved in this work, highlight the contributions of our research, and provide an analysis of the effective and ineffective solutions to the research problem.

# 2. Theoretical Foundation

This chapter gives a brief overview of the basics and background concepts used in this work. Other aspects and additional information can be found in the resources.

## 2.1. Artificial Neural Network

Artificial neural networks are one of the most important concepts in deep learning today, because they are so versatile and have impressive results, which makes them ideal for handling large and complex machine learning tasks such as classifying lots of images, running speech recognition services, and recommending the best videos to watch to hundreds of millions of users or even to produce new data that is very similar to real data [3].

Inspired by biological brains, these models are designed to abstractly simulate the functions of interconnected biological neurons by passing input features through several layers that transform the input using a different set of operations as shown in the following figure:



Figure 2.1.: The different elements of the neuron

Figure 2.1 describes the architecture of one simple neuron that consists of the main operations: scalar multiplication, a summation and a special function called activation function.

The neuron is defined mathematically as follows:

$$y = F(\sum_{i} (x_i \cdot w_i) + b)$$
(2.1)

So that:

- 1. In the beginning each feature  $x_i$  is multiplied by a certain weight  $w_i$ , where :
  - $x_i$  is either numerical value that represent the input data, or the output of another neuron.
  - $w_i$  is numerical value that represent either the importance of the input or the importance of the connections between the neurons.
- 2. Then we compute the weighted sum from the previous step and summed with the bias value and this produces a single output value.
- 3. Then, we apply the result of the weighted sum as an input to the activation function *F*, which in most cases is a non-linear function.

After we have seen how each neuron works, we can define a simple neural network as a combination of 3 sections: first the input layer that contains the properties of the input, then a multiple layers created by combining many neurons in an organized manner into multiple interconnected layers called the hidden layers and finally the output layer that specifies the output [4, 3, 5]. This can be seen more clearly in Figure 2.2.



Figure 2.2.: A multi-layer network that has an input layer, two hidden layers, and an output layer.

### 2.1.1. Activation Functions

The Activation Functions are formulas applied to the output of a neural network's neurons to determine whether they should be activated or not, and these functions can be categorized into two primary types:

- 1. Linear Activation Function.
- 2. Non-linear Activation Functions.

But as we mentioned earlier, non-linear functions are used most of the time, because the data is not usually found in a linear form. Therefore, the use of non-linear functions allows neural networks to learn a better generalization of the data [4, 6].

These are the most common activation functions:

• Identity function:

$$F(x) = x \tag{2.2}$$

• Sigmoid:

$$F(x) = \frac{1}{1 + e^{-x}} \tag{2.3}$$

• Tanh:

$$F(x) = \frac{e^x - e^{-x}}{e^x + e^{-x}}$$
(2.4)

• ReLU:

$$F(x) = max(0, x) \tag{2.5}$$

• Leaky ReLU:

$$F(x) = max(\alpha x, x) \tag{2.6}$$

where:

-  $\alpha$  is a small constant, that provides a small slope to allow for the learning of negative input values instead of a flat slope in ReLU.

Each activation function has its own unique properties and can be used for specific tasks and scenarios in neural network models.

### 2.1.2. Loss Functions

The loss function is crucial in training a neural network, as it evaluates the quality of the model by comparing predicted and target output values. Its primary objective is to minimize the value of the loss function using gradient descent, which ultimately leads to a better model. The lower the loss function value, the more accurate the model is in predicting the target output. Hence, reducing the loss function is essential in achieving the best possible performance of the model.

These are the most common loss functions:

#### • Mean Squared Error (MSE)/L2 loss:

MSE is a loss function that used to minimize the sum of the all the squared differences between the true value y and the predicted value  $\hat{y}$ .

$$MSE = \frac{1}{n} \sum_{i=1}^{n} (y_i - \hat{y}_i)^2$$
(2.7)

#### • Mean Absolute Error (MAE)/L1 loss:

MAE is a loss function that used to minimize the sum of the all the absolute differences between the true value y and the predicted value  $\hat{y}$ .

$$MAE = \frac{1}{n} \sum_{i=1}^{n} |y_i - \hat{y}_i|$$
(2.8)

#### • Binary Cross Entropy (BCE):

BCE is a loss function used in binary classification tasks that compares each of the predicted probabilities to the actual category output which can be either 0 or 1 i.e. one of two predefined categories.

$$BCE = -\frac{1}{n} \sum_{i=1}^{n} y_i \cdot \log(p(\hat{y}_i)) + (1 - y_i) \cdot \log(1 - p(\hat{y}_i))$$
(2.9)

#### Categorical Cross Entropy (CCE):

CCE is a loss function used in multi-class classification tasks that compares each of the predicted probabilities to the actual category output which can one of several predefined categories.

$$CCE = -\frac{1}{n} \sum_{i=1}^{n} \sum_{j=1}^{m} y_{ij} \cdot log(p(\hat{y}_{ij}))$$
(2.10)

#### 2.1.3. Optimizers

Optimizers are algorithms used to adjust the parameters in the neural network such as weights and learning rate during the training process in order to minimize the losses that calculated by the loss functions [7], which are described in the previous subsection 2.1.2. Some common optimizers are:

- **Gradient Descent:** This is the simplest optimizer that relies on the first-order derivative (gradient) of a loss function to determine the optimal adjustments to the model weights. The purpose of this algorithm is to minimize the loss function by iteratively updating the weights in the direction of the negative gradient [7].
- Stochastic Gradient Descent (SGD): It's a variation of Gradient Descent that updates the model parameters with more frequency. Unlike Gradient Descent, which updates the parameters after computing the loss on the entire dataset, SGD updates the parameters after computing the loss on each training sample. So, in a single pass through a dataset of 1000 rows, SGD will update the model parameters 1000 times, compared to just once with Gradient Descent [7].
- **Momentum:** It is a variation of SGD that considers the previous gradients to accelerate the optimization process. By using an exponentially weighted moving average of the gradients, the optimizer is able to smooth out the fluctuations and adjust the step size in the direction of the gradient. The Momentum optimizer has two main hyperparameters: the learning rate and the beta, which controls the decay rate of the momentum [8]. This helps the optimizer to escape from shallow local minima and converge faster to the global minimum.
- AdaGrad (Adaptive Gradient): is an adaptive learning optimization algorithm that can automatically adjust the learning rate for each parameter based on its occurrence frequency, making it suitable for sparse data since it can adaptively adjust the learning rate for each parameter based on the available information [9].

• Adam (Adaptive Moment Estimation): is an optimization algorithm that uses a combination of momentum and adaptive learning rates to optimize the weights of a neural network. The momentum term helps the optimizer to continue to move in the same direction as previous steps, while the adaptive learning rate adjusts the step size based on the gradient of the loss function. The algorithm uses two moving averages of the gradient and its square, to estimate the first and second moments of the gradient. These are used to calculate the learning rate for each parameter, which is then scaled by the momentum term [10, 11].

## 2.1.4. Normalization

Normalization techniques in neural networks are used with the purpose of standardizing and transforming inputs to a uniform scale. Typically, the goal is to center and rescale the data between 0 and 1 or -1 and 1, depending on the data type. To achieve this, a prevalent approach is to compute the mean and standard deviation of the data and normalize each sample by subtracting the mean and dividing by the standard deviation. Normalization aids in stabilizing gradient descent, facilitating the use of larger learning rates, and faster convergence of the models [12]. There are several types of normalization techniques:

- **Batch Normalization:** is a normalization technique introduced in [13] that is used to normalize the activations of each batch during training. This involves gathering all data points in a batch and normalizing them with the same mean and standard deviation for each dimension of the input [14]. It helps to reduce the internal covariate shift, which refers to the change in the distribution of activations due to the update of the parameters. This makes the training of deep networks more stable and faster.
- **Instance Normalization:** as presented in [15], is a normalization technique that normalizes the activations of each instance in the batch independently during training, instead of normalizing across the entire batch. This means that the only difference from batch normalization is that the mean and variance are not computed over the entire batch but rather only over the same data point. It is particularly useful for style transfer and image generation tasks, as it helps to reduce the style discrepancy between the input and output images [14].
- Layer Normalization: is a normalization technique introduced in [16] that is employs a similar technique to batch normalization, but instead of normalizing input features across the batch dimension, it normalizes the input across the features. In essence, layer normalization performs normalization on a per-data point basis. Additionally, the mean and variance are shared across all hidden units (i.e. neurons) in the layer [14].
- **Group Normalization:** is a normalization technique introduced in [17] that is normalizes across groups of channels for each training data point, making it a balance between LayerNorm and InstanceNorm. While Layer Normalization takes the channel dimension into account during computation and Instance Normalization doesn't, GroupNorm groups channels into clusters and normalizes within each group [14].

## 2.2. Convolutional Neural Network

While the concept of *regular neural networks* was presented in the previous section, there are different types of neural networks, which are used for different use cases and data types, and one of the most important of these types is the convolutional neural network or CNN for short, which has become famous in the field of computer vision for its superiority over the rest of the types in image processing tasks such as image classification and object recognition.

CNNs have three main types of layers that are different from a regular neural network:

1. Convolutional layers:

The main difference between a regular neural network and a CNN is the convolution layers which are the most important building block of a CNN. A convolution layer consists of a set of filters defined by a set of learnable weights [4]. Each filter is applied across all receptive areas of the input data and learn multiple features in parallel for a given input by reducing the input to a form that is easy to process, without losing features that are critical to a good prediction.



Figure 2.3.: How the convolution layers work

As shown in the diagram above, the convolutional layers are calculated as follows:

$$Output[0][0] = (2 * 0) + (0 * 3) + (1 * 4)$$
$$+(1 * 2) + (5 * 1) + (3 * 2)$$
$$+(9 * 0) + (0 * 5) + (2 * 0) = 17$$

Then the filter will move and the second cell will be calculated until the entire input is covered.

There are three parameters that affect the size of the output and the way these layers work [18]:

- a) The number of filters: affects the depth of output.
- b) Stride: is the number of pixels or the step size, that the filter sliding over between each convolution.
- c) Padding: is usually used when the filters do not fit the input image. This sets all elements that fall outside of the input matrix to zero, producing a larger or equally sized output.
- 2. Pooling layers:

Pooling layers work similarly to convolutional layers and they also reduce the dimensions of the input. They moving a filter across the entire input, but the difference is that this filter doesn't have any weights. Instead, it applies an pooling operation to the values inside the receptive area.

There are two main types of the pooling operation:

- a) Max pooling: While the filter is working on the input data, it selects the pixel with the maximum value and sends it to the output.
- b) Average pooling: While the filter is working on the input data, it calculates the average value within the receptive area and sends it to the output.

#### 2.2.1. VGG and VGG Loss

VGG is a Convolutional Neural Network architecture proposed in [19]. It is a multi-layer deep neural network architecture where each block of this network consists of 2 or 3 convolutional layers and a pooling layer, and finally 2 fully-connected Layer and then the output layer and it has several variants, the most important are VGG-16 and VGG-19, which consists of 16 and 19 convolutional layers [19, 3].

VGG Loss is a content loss introduced in [20] that is based on the ReLU activation layers of the pre-trained 19-layer VGG network [19, 20].

$$\mathcal{L}_{VGG/i,j} = \frac{1}{W_{i,j}H_{i,j}} \sum_{x=1}^{W_{i,j}} \sum_{y=1}^{H_{i,j}} (\phi_{i,j}(I^{HR})_{x,y} - \phi_{i,j}(G_{\theta_G}(I^{LR}))_{x,y})^2$$
(2.11)

where:

- $\phi_{i,j}$  indicate the feature map obtained by the *j*-th convolution layer (after activation) before the *i*-th max pooling layer within the given VGG19 network.
- $G_{\theta_G}(I^{LR})$  is the feature representations of a reconstructed image.
- $I^{HR}$  is the feature representations of the reference image.
- $W_{i,j}$  and  $H_{i,j}$  are the dimensions of the respective feature maps within the VGG network.

#### 2.2.2. ResNet and Residual Blocks

Residual Networks, or ResNets for short, are a Convolutional Neural Network architecture proposed in [21]. This architecture tries to solve the gradient vanishing problem and degradation problem that occurs in deep convolutional networks. The gradient vanishing occurs during backpropagation when the neural network training algorithm tries to find the weights that make the loss function reach the minimal value until the gradient becomes too small and then disappears, and the optimization cannot continue [22]. The degradation problem is that the deeper layers in the networks get a higher error rate compared to the first layers.

This approach relies on the Residual Blocks or Identity Blocks shown in Fig. 2.4, which are skip-connection blocks that learn residual functions with reference to the layer inputs, instead of learning unreferenced functions [21]. This makes gradients flow through network layers easier and ensures that important features continue to be learned through to the final layers, which allows us to build deep CNNs without the gradient vanishing problem or the degradation problem to occurs.



Figure 2.4.: The architecture of the Residual Block

As shown in Fig. 2.4, we can see that, in addition to the normal connections, there is a skip connection that skips some layers in the model. With the skip connection The original mapping is recast into H(x) = F(x) + x, because when we add the input x to the network output via the skip-connection, the neural network will be forced to learn F(x) = H(x) - x rather than F(x) = H(x) and this called residual learning. where:

- *x* is the input to the Residual block.
- F(x) is a small neural network with several convolution blocks.
- H(x) is the target function that the neural network will learn.

## 2.3. Autoencoders

Before we discuss how *Generative adversarial networks* work, we will first start with autoencoders, which have been first introduced in [23] as a unique neural network architecture consisting of two neural networks concatenated together to reconstruct the input:

- 1. **Encoder:** This is the first network, which compresses the input into a lower-dimensional latent code.
- 2. **Decoder:** This is the second network, which reconstructs the output from the lowerdimensional representation.



Figure 2.5.: The architecture of Autoencoders

As shown in Fig. 2.5, we can see both the encoder and the decoder are connected to create the autoencoder so that the encoder receives a d-dimensional input and encodes it into a latent code with lower dimensions. This means that the encoder will learn a function that reduces the size of the input and creates a more abstract representation of it. On the other hand, the decoder receives the latent code that has been generated by the encoder and reconstructs the original input from this low-dimensional representation, which means that it learns a function that returns the latent code to its original form before the encoding.

There are four parameters that play a role in the training of the autoencoder [24]:

- 1. Latent code size.
- 2. Number of layers, in general, both networks should have the same number of layers.
- 3. Number of nodes per layer.
- 4. Loss function, in general, MSE or BCE is used depending on the situation, for more information see this section 2.1.2.

#### 2.3.1. Convolutional Autoencoders

As we have already seen in the *Convolutional Neural Networks section*, CNNs are the best way to process images, so if we want to build an autoencoder for image processing, we have to use the features of CNNs in it. As we've seen so far, encoders have two main networks, one to compress the input into a latent code, and the other to reconstruct that code back to the original input. From here we note that we can use both *convolutional layers* and *pooling layers* in the encoder because they reduce the dimensions of the input as required, but on the contrary, the decoder rebuilds the input that has been reduced and return it to the original size and for this we use what is called Transposed Convolutional Layers.

Transposed Convolutional Layers, in contrast to the convolutional layers, are a upsampling techniques that increases the dimensions of the input to match the output dimensions.

#### 2.3.2. U-Net

U-Net is an architecture for semantic segmentation introduced in [25] and can be considered as a special *convolutional autoencoder* architecture using skip connections so that the structure of the network looks like the English letter "U", hence the name U-Nets. As we said earlier, this network is an improvement over the convolutional autoencoder, and therefore it contains two main networks, the encoder and decoder, which were explained in the previous section. What distinguishes this structure from its previous one is what is called skip connections, which connect each layer of the encoder with its decoder counterpart, look at Fig. 2.6. These connections pass spatial information to the decoder so that no noise occurs in the output after the reconstruction of the output because normally a lot of information will be lost during the encoding process and this improves the network output.



Figure 2.6.: The architecture of U-Net [25]

## 2.3.3. Variational Autoencoders

So far it seems that the purpose of the autoencoder is to only repeat its inputs. But this is not true, the most important feature in these models is the abstract representation of the data so that the most important properties are encoded in the latent code layer and then decoding them so that they will produce an output that carries the basic characteristics and this is very useful in denoising problems etc. But with that, the result will always the same for specific input, or in other words, there will always be a single reconstruction of one input sample. To change this, the variational autoencoders where proposed in [26]. Variational autoencoders are similar to the *autoencoders*, they consists of an encoder and a decoder, but the main difference is that instead of the encoder learning the latent code of the input, it learns the probability distribution, in other words instead of outputting the discrete values, we will have a probability distribution for each latent feature, which makes the latent space continuous. Then, the decoder can randomly select a sample of a latent variable from its distribution range and produce the output from this sample [4]. Thus, we can get many different results for a single input and for this reason, we can consider VAE as a generative model.

## 2.4. Generative Adversarial Network

Generative adversarial networks, or GANs for short, are deep generative models consisting of two or more neural networks. This architecture was first introduced in [27] and was built from two different neural networks each with a specific task. These two neural networks are:

- 1. **Generator:** This is the model that is responsible for generating new data samples by learning the distribution over the data, which is synthesized from the random vector [27]. In other words, it takes a random vector (typically Gaussian distribution) as an input and produces an output similar to the data it was trained on.
- 2. **Discriminator:** This is the Model that is responsible for determine whether its inputs are coming from the real database, or fake and generated by the generator.

During the training process both models are trained together as one system but the generator and the discriminator have two opposite objectives. On the one hand, the discriminator is trying to get better at distinguishing between real and fake data. On the other hand, the generator tries to generate more accurate data so that the discriminator cannot distinguish it from the real one. Thus, both models are constantly trying to outsmart each other, hence the name "adversarial" [4, 3].



Figure 2.7.: The architecture of GAN

As shown in Fig. 2.7, we can see how the structure of the GANs is interconnected so that the generator receives a random vector and from this input it generates a data sample that flows to the discriminator and there it takes two alternating inputs the real sample and the fake sample and tries to determine these inputs. Ultimately, the goal of this architecture is to make the generator so good that the discernible cannot differentiate between its output and real data, hence the name "generative".

### 2.4.1. Generative Adversarial Network Training

As mentioned earlier, GANs consists of two different models with opposite goals, and for this reason, these models cannot be trained like other neural networks. The GAN training can be considered as a sequential minimax zero-sum game of two players [4] where the training proceeds in alternating phases in each iteration:

- **Phase 1:** Train the discriminator and freeze the generator, this means that at this point the backpropagation is only improving the discriminator. This phase can be divided into several steps:
  - 1. A number of samples are taken from the real data and set as real. In other words, the label for these samples is set to 1.
  - 2. With the same number of samples taken from the real data, fake samples are generated by the generator and set as fake i.e. setting label 0 for these samples.
  - 3. Then the discriminator starts training on this set of data for one step using the *BCE loss function*, because the discriminator considered as a binary classifier whose goal is to distinguish between real and fake data.
- **Phase 2:** Train the generator and freeze the discriminator, this means that at this point the backpropagation is only improving the generator. This phase can be divided into several steps:
  - 1. First, we generate a set of fake data via the generator, and they are all set as real i.e. labels 1 (this time, we don't add real data in the set).
  - 2. Then the discriminator is used to find out whether the data is fake or real.
  - 3. Then the generator starts training to generate data based on the feedback coming from the discriminator, and because the weights of the discriminator are frozen at this phase, we can improve the performance of the generator, and this is what maximizes the loss of the discriminator by making the generator better without making the discriminator worse.

As we can see, the process of training this adversarial system constitutes a min-max game, so that the goal is to minimize the generator loss and maximize the loss of the discriminator at the same time, and for this, distinctive loss functions were created to train this type of neural network architecture.

### 2.4.2. GAN Loss Functions

Here we will show some loss functions that are designed to train GANs:

#### • Min-Max GAN Loss:

The Min-Max loss function is the standard loss function for GANs which was introduced in [27], the original paper that introduced this type of neural network.

This loss function is used so that the generator tries to minimize it while the discriminator tries to maximize it.

$$\mathcal{L}_{GAN}(G,D) = \min_{G} \max_{D} \mathbb{E}_{x}[log(D(x))] + \mathbb{E}_{z}[log(1 - D(G(z)))]$$
(2.12)

where:

- $\mathbb{E}_x$ : is the expected value over all real data samples.
- $\mathbb{E}_{z}$ : is the expected value over all random inputs to the generator.
- D(x): is the discriminator's estimate of the probability that a sample of real data x is real.
- G(z): is the generator's output when given noise z.
- D(G(z)): is the discriminator's estimate of the probability that a fake instance is real.

#### Non-Saturating GAN Loss:

Another alternative to the standard loss function was introduced in the original paper is the Non-Saturating GAN Loss, which modifies the generator section of the previous equation 2.4.2, to overcome the saturation problem that appears in the early stages of training. When the generator is weak, the discriminant is more efficient because the generated data is significantly different from the real data. This leads to saturation in log(1 - D(G(z))). To solve this problem, instead of having the generator to minimize the loss in log(1 - D(G(z))), we can train it to maximize log(D(G(z))). In other words, the generator will aim to maximize the probability of the generated samples being real and not to minimize the probability of images being predicted as fake [27, 28].

#### • Least Squares GAN Loss:

Least Squares GAN or LSGAN loss is a type of loss function used in generative adversarial networks, which was introduced in [29]. Unlike traditional GANs that use binary crossentropy (see equation 2.1.2) loss for the discriminator, LSGAN adopts the least squares loss function. The LSGAN loss function calculates the difference between the output of the discriminator network and the target output, which is 1 for real images and 0 for generated images [30]. The goal of the discriminator network is to output high scores for real data and low scores for fake data, while the generator network is trained to produce data that results in high scores from the discriminator. By utilizing this loss function, the model can produce more stable and high-quality results compared to traditional GANs. LSGAN loss functions are defined as follows:

$$\min_{D} \mathcal{V}_{LSGAN}(D) = \frac{1}{2} \mathbb{E}_{x} [(D(x) - b)^{2}] + \frac{1}{2} \mathbb{E}_{z} [(D(G(z)) - a)^{2}]$$

$$\min_{G} \mathcal{V}_{LSGAN}(G) = \frac{1}{2} \mathbb{E}_{z} [(D(G(z)) - c)^{2}]$$
(2.13)

where:

- *a* is the label for fake data, which is 0
- *b* is the label for real data, which is 1
- *c* is the value that *G* wants *D* to believe for fake data, which is 1.

#### Wasserstein GAN Loss:

The Wasserstein or WGAN loss was proposed in [31]. This method is considered one of the best alternatives to the standard loss function for GANs because it improves the stability of learning during training and solves common problems that occur with GANs such as Mode Collapse and Vanishing Gradient problems. This method works in a slightly different way to the regular method for GANs in that instead of using a discriminator to classify generated data as real or fake, WGAN changes the functionality of the discriminator so that it outputs a scores that represent the realness or fakeness of a particular data instance [31, 28]. In other words, discriminator does not actually classify the inputs, instead for each sample it outputs a number between 0 and 1.

Because the model can't really distinguish between the real and the fake, the WGAN discriminator is actually called a "critic" rather than a "discriminator" [32].

Loss functions are defined as follows:

- Critic Loss: D(x) D(G(z)) so that it tries to maximize this function, which means that it tries to increase the difference between real samples and fake samples [32].
- Generator Loss: D(G(z)) so that it tries to maximize this function, which means that it tries to maximize the discriminator's output for its fake samples [32].

#### 2.4.3. Conditional Generative Adversarial Network

A conditional generative adversarial network, or cGAN for short, was introduced in [33]. It is a conditional type of GANs intended to control the generation of data, where both the Generator and discriminator receive some additional conditioning input information Look at Fig. 2.8. This information could be the class of the input or some other property that is correlated with the output we want, for example images from another domain for image-to-image translation. This improvement comes with faster, more stable training and better-quality results.



Figure 2.8.: The architecture of Conditional GAN

Compared to regular GANs, conditional GANs make a small modification to the standard GAN loss function to make it possible to include additional information by replacing the real data x with x|y and the generated data z with z|y, which means that instead of modeling the joint probability, the conditional GAN is modeling the conditional probability.

$$\mathcal{L}_{cGAN}(G,D) = \min_{G} \max_{D} \mathbb{E}_{x}[log(D(x|y))] + \mathbb{E}_{z}[log(1 - D(G(z|y)))]$$
(2.14)

where:

- *y*: represents the additional informations.
- D(x|y): is the discriminator's estimate of the probability that a sample of real data x is real under the condition of y.
- G(z|y): is the generator's output when given noise z under the condition of y.
- D(G(z|y)): is the discriminator's estimate of the probability that a fake instance is real under the condition of y.

For more details look at the Min-Max loss function in equation 2.4.2

# 2.5. Image-to-Image Translation

Image-to-Image Translation is a class of computer vision and machine learning problems that focuses to transform images from one domain into another so that they have the style and properties of images from another domain, i.e. to learn the mapping between the input image and the output image. This type of problem exists in many fields and has wide uses, for example, problems of improving medical images taken with a microscope or improving images of the ocean depths which are usually blurred and many other problems can be included in this category of problems, see Fig. 2.9. In other words, the main goal of Image-to-Image Translation is to generate new images that are similar to the target domain while preserving the structure and content of the original image. In this thesis we have formulated the problem of improving scanned images as a problem of this kind, so that the damaged scanned images are the source field and we want to convert them into clean images.



Figure 2.9.: Some examples for I2I translation applications [34]

This type of problem can be divided into two main types:

## 1. Supervised Image-to-Image Translation:

Supervised Image-to-Image Translation aims to translate the source images to the target domain through paired datasets consisting of compatible pairs, each in a specific domain. Which means that the dataset consists of images from two different domains, each image in one domain has a corresponding image in the other domain.

## 2. Unsupervised Image-to-Image Translation:

Unsupervised Image-to-Image Translation aims to translate the source images to the target domain through two large datasets of unrelated training images, which means that there is no matching pair for each image in the other domain. In general, most of the problems with translating images are of this type, and the reason is that in real life, pairs of images for both domains cannot always exist.

# 3. Literature Review

This chapter reviews the related work to the problem we are facing. It will present how the papers were collected, then briefly explain the promising results, compare them with each other, and finally choose some approaches to try in solving the problem.

## 3.1. Search Process

This section of the thesis provides a detailed review of the literature search process used to identify sources and research relevant to the problem of improving and removing imperfections from digitally scanned images. The literature search method has been structured in a clear and organized manner, making it easy to access the relevant works that have been identified.

#### 1. Problem Formulation:

Clearly defining the problem is a crucial initial step in the research as it establishes the foundation for the study and serves as a guide for the selection of appropriate methods and techniques. In this thesis, the problem of improving and removing imperfections from digitally scanned images was formally formulated as an image-to-image translation problem, for more details see Section 2.5. Formulating the problem of improving and removing imperfections from digitally scanned images as an image-to-image translation problem is a natural choice because it aligns with the task at hand. Image-to-image translation refers to the process of converting an image from one representation to another, such as from a low-quality scanned image to a high-quality version. In this case, the input image is a low-quality scanned image with imperfections, and the goal is to generate a high-quality version of the image that is clean. By framing the problem in this way, it allows for the use of techniques and methods developed for image-to-image translation, such as GANs, which have been shown to be effective in improving the quality of images. Additionally, this formulation also allows for the use of evaluation metrics commonly used in image-to-image translation tasks, such as peak signal-to-noise ratio (PSNR) and structural similarity index (SSIM), to measure the performance of the proposed approaches.

### 2. Define Search Queries and Search Engines:

After the problem is appropriately defined, it is imperative to begin the process of identifying suitable search queries that can efficiently locate relevant literature and research material pertaining to the given problem. Consequently, it is important to carefully choose the most appropriate search queries that accurately correspond to the problem and its scope to ensure precision and pertinence. Considering these factors, we established the following search queries:

- "Image To Image Translation Using GANs"
- "Improve Scanned Documents with GANs"
- "GANs for Image Enhancement"
- "Removing Imperfections from Scanned Images using GANs"
- "Image Denoising with GANs"
- "Image Deblurring with GANs"
- "Supervised/Unsupervised Image To Image Translation"
- "Image Inpainting with GANs"

These search queries are relevant to the research as they focus on the use of GANs for image-to-image translation and image enhancement, which aligns with the problem of improving and removing imperfections from digitally scanned images. Additionally, these queries also include specific terms such as " improve scanned image" and "removing imperfections" which are relevant to the research problem.

After determining the search queries that will be used, it is important to identify which search engines to use to find relevant literature and research for this problem. Choosing the right search engines is essential as it can greatly impact the quality and quantity of the results obtained, so we have defined the following search engines:

Search Engine	Link
Google Scholar	https://scholar.google.com/
ACM digital library	https://dl.acm.org/search/advanced
arXiv	https://arxiv.org/search/cs
Papers with Code	https://paperswithcode.com/
IEEE Xplore	https://ieeexplore.ieee.org/

These search engines are relevant to the computer science and machine learning research as they provide access to a wide range of scholarly literature that is relevant to our problem.

#### 3. Quick filtering:

After the completion of the previous steps, we commence with Quick filtering, a crucial phase in the search process as it allows us to quickly and efficiently sift through the large number of results obtained from the search engines. The goal of this step is to determine the relevance of the literature and research to the problem at hand by reading the title and abstract of the papers and articles. During this step, we have look for papers and articles that are directly related to our problem and present a promising approach to solving the problem.

This step starts by reading the title and abstract of each paper or article. The title should be informative and give a good indication of what the paper or article is about. The abstract should provide a brief summary of the paper or article and should provide enough information to determine the relevance of the work to the research problem. After reading
the title and abstract, you will then look at the applied examples of the approach that the work presents. This will give you a better understanding of how the approach is used and how well it performs. Papers and articles that do not fit the problem criteria will be excluded from further consideration.

### 4. Promising Results selecting:

After excluding most of the papers that do not fit the criteria of the objective of this thesis, we start selecting the Promising Results by reading the remaining papers more carefully and select the most promising results from them. The goal is to identify the most relevant and promising research that aligns with the objectives of the thesis. This process involves reading the entire paper or article, including the methodology, results, and conclusions. We will then evaluate the strengths and weaknesses of the paper, as well as how well we expect it to solve the research problem. The selected papers or articles will be further analyzed and discussed in the next section.

# 3.2. The Results

In this section, we will discuss the promising results achieved by applying the research steps outlined in the previous section. The results we obtained are as follows:

### • Pix2Pix:

Pix2Pix is a GAN-based model for supervised image-to-image translation introduced in [34]. It uses a conditional GAN to learn mapping from input images to output images with corresponding features, allowing it to produce realistic results. The model was trained with a paired dataset of input and target images, enabling it to perform tasks such as image colorization, style transfer, and edge-to-image conversion. It has been used in various applications, demonstrating its effectiveness in generating accurate and visually pleasing output images.

Pros:

- Easy to train and efficient
- Can be used for various image-to-image translation tasks

Cons:

- Requires paired training data, meaning that there needs to be a corresponding output image for each input image used during training.
- It can handle images with a size of 256 × 256 pixels, which means that it has difficulty creating high-resolution images while preserving detail and output quality.
- May produce low-quality results if the training data is not representative of the target domain.

### • Pix2PixHD:

Pix2PixHD introduced in [35] is a variant of pix2pix that designed to generates highresolution images using a multi-scale generator and discriminator architecture, where the generator generates images at multiple resolutions and the discriminator evaluates the images at each resolution.

Pros:

- Generates high-resolution images
- Can be used for various image-to-image translation tasks

Cons:

- Requires paired training data like Pix2Pix
- Computationally expensive, as it requires more resources and processing power compared to the original pix2pix model.
- May produce low-quality results if the training data is not representative of the target domain.

#### • BicycleGAN:

BicycleGAN is a GAN-based multi-modal for supervised image-to-image translation introduced in [36]. It is designed to handle the problem of multimodal image-to-image translation, where multiple possible outputs exist for a given input. It extends Pix2Pix by learning a stochastic mapping from the source to the target, and shows interesting diversity in the generated samples.

Pros:

- Generates multiple possible variations for a given input.

Cons:

- It can be difficult to control the specific output that the model produces.
- Requires paired training data.
- The model may not generalize well to unseen data.
- It may not be suitable for all types of image-to-image translation tasks.

#### • CycleGAN:

CycleGAN is a GAN-based model for unsupervised image-to-image translation introduced in [37], which uses a cycle consistency loss to ensure that the generated images are consistent with the input images. The model consists of two generators and two discriminators, one for each domain (e.g. source and target).

Pros:

- Can be used for unsupervised image-to-image translation, meaning that it does not require paired training data.
- Generates good images that are consistent with the input images.
- The cycle consistency loss helps to ensure that the generated images are realistic.

Cons:

- The model may not work well when the two domains have a large domain gap.

 May produce low-quality results if the training data is not representative of the target domain.

### • DiscoGAN:

DiscoGAN is a GAN-based model for unsupervised image-to-image translation introduced in [38]. It uses two generators and two discriminators, one for each domain, and two reconstruction loss, one for both the domain to improve the quality of the generated images. DiscoGAN aims to learn the cross-domain relations between the two domains and to generate images that are consistent with the input images.

Pros:

- Can be used for unsupervised image-to-image translation, meaning that it does not require paired training data.
- DiscoGAN is able to discover the underlying structures of the two domains, in order to translate the images from one domain to another.

Cons:

- Difficulty handling large domain gaps.
- It can handle images with a size of 64 × 64 pixels, which means that it has difficulty creating high-resolution images while preserving detail and output quality.
- May produce low-quality results if the training data is not representative of the target domain.

### • UNIT:

UNIT is a GAN-based model for unsupervised image-to-image translation introduced in [39] that utilizes a shared-latent space to align the distributions of two domains. The main idea behind UNIT is to use two autoencoders, two generators and two discriminators for both domains, and use an adversarial loss and a cycle-consistency loss to align the distributions and preserve the structure of the images during translation.

Pros:

- Can be used for unsupervised image-to-image translation, meaning that it does not require paired training data.
- It can handle multiple domains and multiple image modalities, making it a versatile method for image-to-image translation.
- The shared-latent space allows to align multiple domains and generate new samples with structure and quality.

Cons:

- The results may not be as good as supervised methods when the domains have a high degree of dissimilarity.
- It can be sensitive to the choice of the architecture and hyperparameters of the generator and discriminators.

- May produce low-quality results if the training data is not representative of the target domain.

### • CUT:

CUT (Contrastive Unpaired Translation) a GAN-based model for unsupervised image-toimage translation introduced in [40] with a loss function called patch-wise contrastive loss to transfer images from a source domain to a target domain while preserving the structure of the input.

Pros:

- Can be used for unsupervised image-to-image translation, meaning that it does not require paired training data.
- Faster and more efficient in terms of memory.

Cons:

- Model may not be able to learn the nuances of the images as well as a supervised model trained on paired data, which could lead to lower translation quality.
- May produce low-quality results if the training data is not representative of the target domain.

Model	Data Type	Networks	Image Size	Idea
			(in pixel)	
Pix2Pix	Paired data	1G and 1D	256 × 256	conditional GAN
pix2pixHD	Paired data	2G and 3D	$2048 \times 1024$	multi-scale architectures
BicycleGAN	Paired data	2AE, 2G and 2D	$128 \times 128$	cVAE-GAN and cLR-GAN
CycleGAN	Unpaired data	2G and 2D	$256 \times 256$	cycle consistency loss
DiscoGAN	Unpaired data	2G and 2D	64 × 64	two reconstruction losses
UNIT	Unpaired data	2VAE, 2G and 2D	640  imes 480	shared latent space
CUT	Unpaired data	1G and 1D	$256 \times 256$	contrastive learning

Table 3.1.: This table shows which GAN methods were found, with relevant information to compare them.

# 3.3. Selected Approaches

In this section, we will introduce the approaches that will be utilized in this thesis. The selection is based on the different approaches presented in the previous section. We will divide the chosen methods into two categories, corresponding to the two formulations of our problem (see Section 2.5): one for Supervised Image-to-Image Translation and another for Unsupervised Image-to-Image Translation.

For supervised image-to-image translation, we have selected Pix2Pix for our problem of improving and denoising/deblurring scanned documents. This model has been shown to be very effective in a variety of image-to-image translation tasks, and so we like to test it on our problem. Pix2PixHD, which is another variant of Pix2Pix for handling high-resolution images, was initially considered, but was ultimately excluded due to its computationally expensive nature compared to Pix2Pix. As for the BicycleGAN, it is the other modification of Pix2Pix to generate several variations for each input, and this matter is not important in our case, so it was excluded.

For unsupervised image-to-image translation, we have selected CycleGAN. It is based on the idea of using a cycle-consistency loss to ensure that the translated images can be translated back to the original images, and this gives us two generators, one of which is to transfer the image from the first domain to the second, and the other does the opposite, which helps us in data augmentation for other training later. The reason for choosing CycleGAN is that it is a well-established model and has been shown to be effective in a variety of image-to-image translation tasks, so we like to test if it will be a good fit for our problem.

These methods will be used in the following chapter to put the insights gained from the literature review into practice. Therefore, the following chapter will provide a detailed description of these methods, and the evaluation chapter will outline the steps taken to ensure the validity and reliability of the results through the evaluation setup.

# 4. Methods

In this chapter, we present the methods and approaches used in our study to evaluate the effectiveness of GAN-based models for improving scanned images. The primary objective of this study is to test and compare the performance of different GAN architectures for this task. We have selected a combination of supervised and unsupervised approaches including Pix2Pix and CycleGAN to achieve this objective. The chapter starts by providing a detailed description of the methods, including their architectures, training procedures and implementation. We also will delve into the technical aspects of the proposed solutions, including the tools, methodologies, and techniques that will be used to bring the solution to fruition. The main goal of this chapter is to provide a clear and thorough understanding of the methods and the underlying rationale behind them.

# 4.1. General remarks

Initially, we will provide general remarks about the software and hardware environment used in this work. This will include the choice of programming language, libraries, and tools used for developing and training the system, as well as the specifications of the computer system used for running the model. By providing this information, we aim to provide a clear understanding of the environment used in this work.

Pix2Pix<sup>1</sup> and CycleGAN<sup>2</sup> were implemented in accordance with the references [34, 37] and their open source implementation.

All the visual aids presented in this work were created using Diagrams.net<sup>3</sup>

## 4.1.1. Hardware environment

### Training hardware environment:

The hardware environment used for the training in this work consisted of a computer from the university with an AMD EPYC 7413 24-core processor, 16 GB of RAM, an RTX A6000 graphics card with 48GB of GDDR6 memory, and Google Colab Pro+. The AMD EPYC 7413 processor is a powerful server-grade processor with 24 cores and 48 threads. The high core count makes it ideal for running complex computations and multi-tasking. This processor has a base clock speed of 2.8 GHz and a boost clock speed of 3.9 GHz, which makes it efficient

<sup>&</sup>lt;sup>1</sup>https://github.com/phillipi/pix2pix, accessed 24. Feb. 2023

<sup>&</sup>lt;sup>2</sup>https://github.com/junyanz/CycleGAN, accessed 24. Feb. 2023

<sup>&</sup>lt;sup>3</sup>https://www.diagrams.net/

for high-performance computing tasks like deep learning. The RTX A6000 graphics card is a high-end GPU designed for accelerating the training of deep neural networks. It features 10752 CUDA cores and 48 GB of GDDR6 memory, making it ideal for processing large batches of data during training. The large amount of memory available on this graphics card is particularly useful for processing large datasets and models. 16 GB of RAM is a relatively small amount of memory for training large machine learning models, but it may have been sufficient for the specific dataset and model used in this work. However, depending on the size of the dataset and the complexity of the model, more RAM may have been required for optimal performance. Using Google Colab Pro+ along with the university's computer hardware environment allows for a powerful and efficient hardware environment for training machine learning models. The cloud-based platform provides access to high-end hardware that can speed up the training process and reduce the time required for large-scale machine learning tasks.

The use of university hardware and Google Colab Pro+ highlights the importance of access to powerful computing resources for machine learning research. While not all researchers may have access to similar hardware environments, the use of cloud-based platforms like Google Colab Pro+ can provide a cost-effective way to access high-end hardware resources for machine learning tasks.

### Testing hardware environment:

The hardware environment used for testing in this work consisted of a computer with an AMD Ryzen 9 5900HS processor, 16 GB of RAM, an RTX 3080 graphics card with 8GB of GDDR6 memory. The AMD Ryzen 9 5900HS processor is a powerful processor designed for high-end computing tasks. It features 8 cores and 16 threads. This makes it good for tasks such as testing machine learning models, as it can efficiently process large amounts of data and perform complex computations. The RTX 3080 graphics card is a high-end GPU that is good for testing of machine learning models. It features 8704 CUDA cores and 8 GB of GDDR6 memory, making it good for processing large amounts of data and performing complex calculations during testing. 16 GB of RAM is a moderate amount of memory for testing machine learning models, and it have been sufficient for the specific models and datasets used in this work. However, larger datasets or more complex models may have required additional RAM for optimal performance.

Overall, the hardware environment used for testing in this work is well-suited for machine learning tasks. The combination of the AMD Ryzen 9 5900HS processor and the RTX 3080 graphics card provides a good computing resources for efficiently processing large amounts of data and performing complex calculations. While the 16 GB of RAM is a moderate amount of memory, it may be sufficient for many machine learning tasks and datasets.

It is worth noting that the specific hardware environment used for training or testing may vary depending on the size and complexity of the models and datasets used in different machine learning projects. However, the combination of a powerful processor, high-end graphics card, and sufficient RAM is generally a good starting point for testing machine learning models.

# 4.1.2. Software environment

The software environment used in this work consisted of several libraries and tools that were essential for developing and testing the deep learning model. The main programming language used was Python 3.10, and several libraries including PyTorch (1.13), CUDA (11.7), NumPy (1.24.1), OpenCV (4.7.0.68), PIL (9.3.0), Seaborn (0.11.2), Plotly (5.8.0), Streamlit (1.17.0), augraphy (8.0.1), Albumentations (1.3.0), PyMuPDF (1.20.1), and pdf2image (1.16.0).

**Python** is a popular programming language in the field of deep learning and machine learning due to its flexibility, readability, and a wide range of available libraries and frameworks. Python 3.10 was used as the main programming language in this work because it was compatible with the required libraries, some of which did not support the latest version 3.11 at the time of this writing, so Python 3.10 was the best option to ensure compatibility with all of the libraries used.

**PyTorch + CUDA** were used in this work to take advantage of the powerful parallel computing capabilities of NVIDIA GPUs, which can significantly speed up deep learning training and inference. PyTorch is an open-source machine learning library based on the Torch library, which provides fast, flexible, and easy-to-use tools for building deep neural networks. PyTorch is also widely used in the deep learning community and has a large and active developer community that provides frequent updates, bug fixes, and new features. Additionally, CUDA is a parallel computing platform and programming model that enables developers to use the power of NVIDIA GPUs for high-performance computing tasks. Together, PyTorch and CUDA made it possible to train and test the deep learning model in this work efficiently and effectively.

**NumPy** is used in this work to perform numerical computations and operations that involve arrays and matrices, which is a common requirement in many machine learning applications, including image processing. OpenCV uses Numpy arrays to store and manipulate image data. Similarly, PyTorch relies heavily on Numpy arrays for data and matrices manipulation, as it provides efficient operations for linear algebra, Fourier analysis, and random number generation, among others. Therefore, it is an essential library for numerical computing, and its integration with other libraries enhances the efficiency and usability of the overall software environment.

**OpenCV and PIL** were used in this work to manipulate and process the image data for deep learning training and inference. OpenCV is an open-source computer vision library that provides tools for image and video analysis, manipulation, and processing. It offers a wide range of functions for various image processing tasks, such as resizing, cropping, and filtering images. Additionally, PIL (Python Imaging Library) is a library that adds support for opening, manipulating, and saving many different image file formats. PIL is useful for image preprocessing, which is a critical step in any image-based deep learning task, as it can improve the quality and consistency of the data. PIL provides tools for resizing, cropping, and transforming images, as well as for applying various image filters and enhancements. Together, OpenCV and PIL provided powerful tools for the image processing and preparation required for training and testing the deep learning models in this work. **Pandas** was used in this work to collect and manipulate data during the evaluation process. Specifically, it was used to read in the results of the models, which were output as a CSV file, and manipulate the data into an appropriate form for analysis and visualization. Pandas is a Python library that provides easy-to-use data structures and data analysis tools, making it ideal for working with tabular data.

**Seaborn and Plotly** were used to visualize the results because they offer powerful and flexible tools for creating high-quality and interactive visualizations in Python. Seaborn is a Python data visualization library that is built on top of Matplotlib and provides a higher-level interface for creating informative and attractive statistical graphics. Plotly, on the other hand, is an interactive visualization library that allows users to create and share interactive, web-based visualizations in Python.

**Streamlit** was used to make a prototype for using pre-trained models because it provides an easy-to-use framework for creating web applications with Python. Streamlit allows users to build interactive and responsive web applications without requiring extensive knowledge of web development technologies such as HTML, CSS, or JavaScript. This makes it an ideal choice for creating quick prototypes of machine learning models, where the focus is on showcasing the functionality of the models rather than the design of the application.

**Augraphy and Albumentations** were used to make image augmentations and other transformations because they are powerful and efficient libraries for creating custom data augmentations for image processing tasks. Augraphy provides scanned image effects, which can generate numerous variations of an original document by simulating a variety of paper-oriented process distortions, including effects from scanning, printing, and faxing. Albumentations, on the other hand, is a fast and flexible image augmentation library that is designed for use with deep learning frameworks such as PyTorch. It offers a wide range of transformations that can be applied to images, such as rotations, flips, and color adjustments.

**PyMuPDF and pdf2image** were used to handle PDF files because they provide efficient tools for working with PDF documents like converting PDF files to various image formats and resolutions, which is useful in machine learning projects that involve processing document images.

# 4.2. Pix2Pix

Pix2Pix is a generative adversarial network (see section 2.4) introduced in [34] with the aim of being an approach to solve the supervised image-to-image translation problem (see section 2.5). It is based on the conditional generative adversarial network that has been introduced in section 2.4.3, where we condition an input image in the source domain and generate a corresponding output image in the target domain. This approach not only learns the mapping from the input image to the output image, but also it is learning the loss function to train this mapping.



Figure 4.1.: The architecture of Pix2Pix

This allows for flexibility in applying the same approach to different image-to-image translation problems that require different loss functions [34]. As we mentioned earlier, this approach is supervised approach, which means that the training data for this model should be consist of pairs of images from different domains, with a correspondence between each pixel in the pair. The generator and discriminator are designed differently, with the generator using a U-Net architecture (see section 2.3.2) and the discriminator using a patch-based approach to give more detailed feedback on different parts of the image. Pix2Pix uses an input image, such as a damaged scanned image, as the conditioning factor for the generator to produce a realistic clean output image. The discriminator receives the input image along with either the real target output or the generated output and must determine if the image is real or fake, see Fig 4.1.

### 4.2.1. Generator architecture

The generator architecture in Pix2Pix is based on the U-Net architecture [34] that has been introduced in section 2.3.2. The U-Net architecture utilizes a specific type of structure known as an encoder-decoder with skip connections, in which the network's architecture is divided into two distinct portions. The first portion, referred to as the encoder, contains convolutional layers and max pool layers to progressively down-sample the input image. The second portion,

referred to as the decoder, contains layers such as convolutional transpose to upscale the image, thus increasing the resolution of the output image. The skip connections connect the corresponding blocks that have the same sizes between the two portions before going into each block in the decoder [34, 25]. This structure enables the preservation of low-level features while generating outputs. However, there are some significant distinctions between the Pix2Pix generator and the original U-Net. The encoder in Pix2Pix contains downsample-blocks that include a convolutional layer, a BatchNorm layer, and a LeakyReLU activation function. Additionally, the decoder in Pix2Pix uses upsample-blocks that are contains a transposed convolution, a BatchNorm layer, and a ReLU activation function. Moreover, dropout layers were added to the first three blocks of the decoder [34].



Figure 4.2.: Pix2Pix U-Net Generator

As we can see in the Fig 4.2 the encoder takes in an input image of size  $256 \times 256$  pixels with 3 channels, and through eight encoder blocks, it compresses the input while decreasing the spatial size by a factor of two in each block. The final output of the encoder has a spatial size of  $1 \times 1$  pixel and 512 channels. Meanwhile, on the decoder side, it starts with an input of  $1 \times 1$  pixel and uses eight decoder blocks to increase the spatial size back to the original size of  $256 \times 256$  pixels with 3 channels, resulting in the generated image as output.

### 4.2.2. Discriminator architecture

The discriminator architecture in Pix2Pix is patch-based architecture called Markovian discriminator or PatchGAN [34]. Instead of evaluating the full image, it evaluates details on a  $N \times N$  patch level, determining if each  $N \times N$  patch is real or fake. This results in a feature map representing the loss, with each patch in the original image corresponding to a single pixel in the loss map with value between 0 and 1 where 0 is fake and 1 is real. The primary advantages of the PatchGAN discriminator come from the facts that they have fewer training parameters, run faster, it can be applied to arbitrarily large images and It also evaluates the quality of the input image according to the quality of the local patches, rather than their global property [34, 41].



Architecture: C64-C128-C256-C512

Figure 4.3.: Pix2Pix Discriminator (PatchGAN)

As we can see in the Fig 4.3 A pair of samples (one from each domain) are concatenated along the depth channel, and this 6 channels  $256 \times 256$  pixels image is treated as the actual input of the PatchGAN discriminator. The discriminator network maps the input to a 1 channel  $30 \times 30$  pixels image, which is the loss map that used to calculate the loss. It consists of five blocks, each containing a convolution layer with a kernel size of 4 and a stride of 2 (except for the 4th and 5th blocks, which have a stride of 1), a BatchNorm layer, and a LeakyReLU activation function (excluding the 5th block).

The patch size that has been used in PatchGAN is  $70 \times 70$  and it has calculated by considering the different convolution layers in the discriminator network. For each layer, the size of the input patch corresponding to a single pixel in the output feature map is calculated using the formula:

Thus, the input patch size per pixel in the output feature map for each layer in the discriminator network can be computed as follows:

- 5th layer: k=4, s=1. Input patch size is 4 (equal to kernel size).
- 4th layer: k=4, s=1. Input patch size is 4 + 1 \* (4-1) = 7.
- 3rd layer: k=4, s=2. Input patch size is 4 + 2 \* (7-1) = 16.
- 2nd layer: k=4, s=2. Input patch size is 4 + 2 \* (16-1) = 34.
- 1st layer: k=4, s=2. Input patch size is 4 + 2 \* (34-1) = 70.

Consequently, the end result is a patch size of  $70 \times 70$ , which are transformed into separate pixels in the  $30 \times 30$  loss map. Any pixel outside the  $70 \times 70$  patch will not have an impact on the corresponding pixel in the loss map [41].

### 4.2.3. Loss function

Pix2Pix model trained similar to conditional GAN, which we introduced in section 2.4.3, but with a slight modification. It uses a combination of adversarial loss (see section 2.4.2) and reconstruction loss. The adversarial loss is computed using a discriminator network that tries to distinguish the generated image from the target image, while the generator network tries to produce an image that can fool the discriminator under the condition of the input. The reconstruction loss is the L1 loss (see Equation 2.8) between the generated image and the target image, which helps to ensure that the generated image is visually similar to the target image. So, the loss function is as follows:

$$\mathcal{L}_{Pix2Pix}(G,D) = \min_{G} \max_{D} \mathcal{L}_{cGAN}(G,D) + \lambda \mathcal{L}_{L1}(G)$$
(4.1)

where:

- $\mathcal{L}_{cGAN}(G, D)$  is the conditional GAN loss function, see Equation 2.14
- $\mathcal{L}_{L1}(G)$  is the L1/MAE loss function between the generated data and the ground truth data, see Equation 2.8
- $\lambda$  controls the relative importance for  $\mathcal{L}_{L1}(G)$ ,

For more details on the GANs training process see Section 2.4.1.

#### 4.2.4. Implementation

After having reviewed the key elements of Pix2Pix, it is now appropriate to delve into its implementation with greater detail. This section will guide you through all the necessary steps for creating a Pix2Pix model, which includes constructing the generator and discriminator networks, configuring hyperparameters, and ultimately, training the model. Before delving into Pix2Pix's implementation, it is important to first introduce the libraries that will be used:

```
1 import torch
```

```
2 import torch.nn as nn
```

Listing 4.1: Libraries used in CycleGAN

where:

- **torch** is the main PyTorch package, which provides tensor computation with strong support for GPU acceleration, automatic differentiation, and various other functionalities that make it a popular choice for deep learning research and applications.
- torch.nn is a subpackage within PyTorch that provides various building blocks for constructing neural networks. It includes various pre-built layers (such as convolutional layers, fully-connected layers, activation functions, etc.) and tools for defining and training neural networks.

#### 4.2.4.1. Generator network

As mentioned earlier in section 4.2.1, the construction of the generator's network involves utilizing a U-Net architecture that comprises an encoder-decoder with skip connections. The encoder part of the network consists of 8 downsample blocks that serve to reduce the dimensionality of the input image. The encoder is constructed in the following manner:

```
encoder_layers = [
1
     # The first downsample-block with the specified parameters
2
3
     nn.Conv2d(in_channels=3,out_channels=64,kernel_size=4,stride=2,padding=1),
4
     nn.LeakyReLU(0.2),
     # The second downsample-block with the specified parameters
5
     nn.Conv2d(in_channels=64,out_channels=128,kernel_size=4,stride=2,padding=1),
6
7
     nn.BatchNorm2d(128), nn.LeakyReLU(0.2),
     # The third downsample-block with the specified parameters
8
     nn.Conv2d(in_channels=128,out_channels=256,kernel_size=4,stride=2,padding=1),
9
     nn.BatchNorm2d(256), nn.LeakyReLU(0.2),
10
     # The fourth downsample-block with the specified parameters
11
     nn.Conv2d(in_channels=256,out_channels=512,kernel_size=4,stride=2,padding=1),
12
     nn.BatchNorm2d(512), nn.LeakyReLU(0.2),
13
14
     # The fifth downsample-block with the specified parameters
     nn.Conv2d(in_channels=512,out_channels=512,kernel_size=4,stride=2,padding=1),
15
     nn.BatchNorm2d(512), nn.LeakyReLU(0.2),
16
     # The sixth downsample-block with the specified parameters
17
18
     nn.Conv2d(in_channels=512,out_channels=512,kernel_size=4,stride=2,padding=1),
19
     nn.BatchNorm2d(512), nn.LeakyReLU(0.2),
     # The seventh downsample-block with the specified parameters
20
     nn.Conv2d(in_channels=512,out_channels=512,kernel_size=4,stride=2,padding=1),
21
     nn.BatchNorm2d(512), nn.LeakyReLU(0.2),
22
23
     # The eighth downsample-block with the specified parameters
     nn.Conv2d(in_channels=512,out_channels=512,kernel_size=4,stride=2,padding=1),
24
25
     nn.LeakyReLU(0.2)]
```

#### Listing 4.2: Pix2Pix encoder

As we see, this code snippet defines the encoder part of the U-Net architecture used in the Pix2Pix model. Each downsample block is defined as a series of operations on the input tensor. The operations include a 2D convolutional layer with specified parameters such as input and output channels, kernel size, stride, and padding, followed by a batch normalization layer and a leaky rectified linear unit (LeakyReLU) activation function with a negative slope of 0.2. The first downsample block takes a 3-channel input image and outputs a 64-channel feature map with a kernel size of 4, a stride of 2, and padding of 1. Subsequent downsample blocks double the number of channels and halve the spatial dimensions of the feature maps until the eighth downsample block does not apply batch normalization, which is a deviation from the standard U-Net architecture.

After the encoder, the generator's network includes a decoder that applies 8 upsample blocks with skip connections to reconstruct the output image from the encoded feature map. The decoder is constructed in the following manner:

```
1
   decoder_layers = [
     # The first upsample-block with the specified parameters
2
3
     nn.ConvTranspose2d(in_channels=512,out_channels=512,kernel_size=4,stride=2,
     padding=1), nn.BatchNorm2d(512), nn.ReLU(True), nn.Dropout2d(0.5),
4
     # The second upsample-block with the specified parameters
5
     nn.ConvTranspose2d(in_channels=1024,out_channels=512,kernel_size=4,stride=2,
6
     padding=1), nn.BatchNorm2d(512), nn.ReLU(True), nn.Dropout2d(0.5),
7
8
     # The third upsample-block with the specified parameters
     nn.ConvTranspose2d(in_channels=1024,out_channels=512,kernel_size=4,stride=2,
9
     padding=1), nn.BatchNorm2d(512), nn.ReLU(True), nn.Dropout2d(0.5),
10
     # The fourth upsample-block with the specified parameters
11
     nn.ConvTranspose2d(in_channels=1024,out_channels=512,kernel_size=4,stride=2,
12
     padding=1), nn.BatchNorm2d(512), nn.ReLU(True),
13
     # The fifth upsample-block with the specified parameters
14
15
     nn.ConvTranspose2d(in_channels=1024, out_channels=256,kernel_size=4,stride=2,
     padding=1), nn.BatchNorm2d(256), nn.ReLU(True),
16
     # The sixth upsample-block with the specified parameters
17
     nn.ConvTranspose2d(in_channels=512,out_channels=128,kernel_size=4,stride=2,
18
     padding=1), nn.BatchNorm2d(128), nn.ReLU(True),
19
     # The seventh upsample-block with the specified parameters
20
     nn.ConvTranspose2d(in_channels=256,out_channels=64,kernel_size=4,stride=2,
21
     padding=1), nn.BatchNorm2d(64), nn.ReLU(True),
22
23
     # The eighth upsample-block with the specified parameters
     nn.ConvTranspose2d(in_channels=64, out_channels=3,kernel_size=4,stride=2,
24
     padding=1), nn.Tanh()]
25
```

### Listing 4.3: Pix2Pix decoder

As we see, this code defines the decoder part of the U-Net architecture used in the Pix2Pix model. Each upsample block is composed of a transpose convolutional layer with specified parameters, followed by batch normalization, a ReLU activation function, and dropout regularization with a rate of 0.5 in the first three upsample blocks. The kernel size for all the upsample blocks is 4, stride is 2, and padding is 1.

The first upsample block takes a 512-channel encoded feature map and outputs a 512-channel feature map. Subsequent upsample blocks halve the number of channels and double the spatial dimensions of the feature maps until the eighth upsample block outputs a 3-channel image with the same spatial dimensions as the input image. The decoder applies skip connections to combine features from the corresponding downsample block in the encoder and the upsample block in the decoder. The final activation function used in the last upsample block is a hyperbolic tangent (Tanh) function, which maps the pixel values of the output image to the range [-1, 1]. You can find the complete generator in the appendix C.3

#### 4.2.4.2. Discriminator network

As previously noted in section 4.2.2, the implementation of the discriminator's network consists of a series of blocks including Convolutional layer, Batch Normalization layer, and LeakyReLU activation layer as follows:

```
1 discriminator_layers = [
2
     # The first block with the specified parameters
3
     nn.Conv2d(in_channels=6,out_channels=64,kernel_size=4,stride=2,
     padding=1), nn.LeakyReLU(0.2, True),
4
5
     # The second block with the specified parameters
     nn.Conv2d(in_channels=64,out_channels=128,kernel_size=4,stride=2,
6
     padding=1), nn.BatchNorm2d(128), nn.LeakyReLU(0.2, True),
7
8
     # The thierd block with the specified parameters
9
     nn.Conv2d(in_channels=128,out_channels=256,kernel_size=4,stride=2,
     padding=1), nn.BatchNorm2d(256), nn.LeakyReLU(0.2, True),
10
11
     # The fourth block with the specified parameters
     nn.Conv2d(in_channels=256,out_channels=512,kernel_size=4,stride=1,
12
     padding=1), nn.BatchNorm2d(512), nn.LeakyReLU(0.2, True),
13
     # The final block with the specified parameters
14
     nn.Conv2d(in_channels=512,out_channels=1,kernel_size=4,stride=1,
15
     padding=1)]
16
```

Listing 4.4: Pix2Pix discriminator layers

This code snippet 4.4 is composed of five blocks that define the discriminator network architecture. The first block takes a 6-channel input tensor (combining the source and target channels) and performs a 2D convolutional filter with 64 output channels, using a kernel size of 4, a stride of 2, and a padding of 1. The output then undergoes a LeakyReLU activation function with a slope of 0.2. In the second block, the output from the first block (which has 64 channels) is subjected to a 2D convolutional filter with 128 output channels, using a kernel size of 4, a stride of 2, and a padding of 1. The output is normalized via batch normalization and then passed through a LeakyReLU activation function. The third block takes the output from the second block (which has 128 channels) and applies a 2D convolutional filter with 256 output channels, using a kernel size of 4, a stride of 2, and a padding of 1. The output is normalized via batch normalization and then passed through a LeakyReLU activation function. In the fourth block, the output from the third block (which has 256 channels) is subjected to a 2D convolutional filter with 512 output channels, using a kernel size of 4, a stride of 1, and a padding of 1. The output is normalized via batch normalization and then passed through a LeakyReLU activation function. Finally, the fifth block takes the output from the fourth block (which has 512 channels) and performs a 2D convolutional filter with 1 output channel, using a kernel size of 4, a stride of 1, and a padding of 1. The resulting output is a  $30 \times 30$  tensor with 1 channel, which is used by the adversarial loss to improve the GAN model.

You can find the complete discriminator in the appendix C.4

### 4.2.4.3. Hyperparameters

Hyperparameters are essential in determining the performance of any model and can be adjusted to optimize its training process. In the case of Pix2Pix, the following hyperparameters are utilized by default:

- **Learning rate:** The learning rate determines how quickly the model updates its internal parameters during training. A lower learning rate can lead to more precise updates, but can also increase the amount of time it takes for the model to converge on a solution. In the case of Pix2Pix, the default learning rate is set to 0.0002.
- **Batch size:** Batch size refers to the number of samples that are processed by the model during each iteration of training. A larger batch size can lead to faster training times, but may also require more memory and can make it difficult for the model to converge on a solution. In the case of Pix2Pix, the default batch size is set to 1, meaning that the model processes one image at a time during training.
- L1 lambda: L1 lambda is a hyperparameter that balances the weight of the L1 loss term in the generator's loss function. The L1 loss measures the difference between the generated image and the target image in terms of absolute pixel differences. A higher L1 lambda value puts more emphasis on the L1 loss term and results in sharper, more accurate images, while a lower value gives less weight to the L1 loss and can produce smoother but less accurate images. The default value for L1 lambda in Pix2Pix is set to 100.
- **Optimizer:** The optimizer is a function that determines how the model updates its internal parameters during training. Adam is a popular optimizer that is commonly used in machine learning models. In the case of Pix2Pix, Adam is utilized as the default optimizer.
- Adam beta 1 and beta 2: Adam beta 1 and beta 2 are hyperparameters that determine the decay rates for the first and second moments of the gradients during training, respectively. These hyperparameters impact the speed and stability of the model's training process. In the case of Pix2Pix, the default values for Adam beta 1 and beta 2 are set to 0.5 and 0.999, respectively.

### 4.2.4.4. Training algorithm

The training algorithm for Pix2Pix follows a similar approach to regular GANs (refer to section 2.4.1) but with some differences in the loss function (see section 4.2.3).

During training, the discriminator is updated using only adversarial loss to ensure that the generated images are realistic and indistinguishable from the real images. On the other hand, the generator is updated using both adversarial loss and L1 loss to ensure that the generated images have a similar pixel-level appearance to the ground truth images. By combining these two losses, the generator can produce high-quality translations between the two domains.

Algorithm 1 Pix2Pix training algorithm				
1: procedure Pix2Pix				
// Initialize discriminators networks				
2: $D \leftarrow discriminator\_network()$				
// Initialize generators networks				
3: $G \leftarrow generator\_network()$				
4: <b>for each</b> epoch <b>do</b>				
for each batch of images from domain A and domain B do				
// Train the discriminator				
6: $generated_B \leftarrow G(real_A)$				
7: $loss_D \leftarrow adversarial\_loss(real_B, target\_value = 1.0) +$				
8: <i>adversarial_loss(generated_B, target_value = 0.0)</i>				
9: update_weights(loss_D, D.weights)				
// Train the generator				
10: $generated_B \leftarrow G(real_A)$				
11: $loss_G \leftarrow adversarial_loss(generated_B, target_value = 1.0) +$				
12: lambda_weight * L1_loss(real_B, generated_B)				
13: update_weights(loss_G,G.weights)				
14: end for				
15: end for				
16: end procedure				

# 4.3. CycleGAN

CycleGAN is an approach based on generative adversarial networks introduced in [37] for solving the problem of unsupervised image-to-image translation (see section 2.5), meaning that it can be trained without the need for paired datasets. It is a bidirectional generative model based on the concept of cycle consistency. It consists of four neural networks, two generators and two discriminators. The reason for that because CycleGAN addresses the problem of image-to-image translation by treating it as an image reconstruction problem, where the input image is transformed into another domain using the first generator, and then transformed back into the original domain using the second generator. The accuracy of the reconstruction is measured by calculating the mean squared error (see section 2.1.2) between the original image and the reconstructed image. This helps to ensure that the content and structure of the original image is preserved during the translation process.



Figure 4.4.: The architecture of CycleGAN

As we can see in the Fig. 4.4, the process starts by feeding a sample from domain A into generator A2B, producing a translated image in domain B. This translated image then passed through the discriminator B and with real B samples to calculate the adversarial loss. At the same time, the translated image is also processed through generator B2A to generate a reconstructed image to calculate the cycle consistency loss between the reconstructed and original image. The adversarial loss is used to update the discriminator weights, and both the adversarial and cycle consistency losses are used to update the generator weights. This process repeats with generator B2A and discriminator A until generated domain A and B images resembling the real domains.

### 4.3.1. Generator architecture

The generator in CycleGAN is based on an encoder-decoder architecture, which is augmented with residual blocks from the ResNet architecture [37] that has been introduced in section 2.3 and 2.2.2. The encoder part of the generator compresses the input image into a lower-dimensional representation using downsample-blocks that include a convolutional layer, a instance normalization layer, and a ReLU activation function, while the decoder part expands the compressed representation back to the original image size using upsample-blocks that are contains a transposed convolution layer, a instance normalization layer, and a ReLU activation function. The residual blocks in the generator allow for the learning of residual functions that help improve the quality of the generated images.



Figure 4.5.: CycleGAN Generator

As we can see in the Fig. 4.5, the generator first compresses the input image, which has dimensions of  $256 \times 256$  pixels with 3 channels, to a lower-dimensional representation of size  $64 \times 64$  pixels with 256 channels through a series of three encoder blocks. This representation is then processed by a series of residual blocks, which interpret the encoding through the use of skip connections to stabilize the model and preserve the features of the original image. Finally, the generator utilizes a series of upsampling layers through the decoder part to increase the spatial resolution of the representation and produce the final output image, which has the same dimensions as the input image.

### 4.3.2. Discriminator architecture

The discriminator in CycleGAN is patch-based architecture called Markovian discriminator or PatchGAN [34, 37]. It's the same discriminator architecture used in the Pix2Pix method, except that the input image has a depth channel of 3, instead of 6, and it uses instance normalization layer instead of batch normalization layer. Refer to section 4.2.2 for further details.

# 4.3.3. Loss function

CycleGAN is trained using combination of adversarial losses for each generator and discriminator pair (see section 2.4.2), and the concept of cycle consistency loss [37], which aims to keep the output of the GAN generator consistent with the original input. The first generator converts an image from one domain to another, and the cycle consistency loss requires that when the generated image is reconstructed back to the original domain using the second generator, it should be similar to the original image.



Figure 4.6.: Cycle consistency loss [37]

As we can see in the Fig. 4.6, cycle consistency loss is comprised of two components for each direction, forward cycle consistency and backward cycle consistency [37].

- 1. Forward cycle consistency ensures that for each image *a* from domain *A*, it can be transformed back to its original form through the image translation cycle. This means that starting with an image from domain *A*, after transforming it through the first generator *A2B* and then transforming the result through the second generator *B2A*, the final image should closely resemble the original image, i.e.,  $a \rightarrow A2B(a) \rightarrow B2A(A2B(a)) \approx a$
- 2. Backward cycle consistency ensures that for each image *b* from domain *B*, it can be transformed back to its original form through the image translation cycle. This means that starting with an image from domain *B*, after transforming it through the second generator *B*2*A* and then transforming the result through the first generator *A*2*B*, the final image should closely resemble the original image, i.e.,  $b \rightarrow B2A(b) \rightarrow A2B(B2A(b)) \approx b$ .

The combination of these two constraints forms the cycle consistency loss, which serves as a crucial component in maintaining the authenticity and integrity of the generated images. By ensuring that the generated images retain the semantic and structural information of the original images, the cycle consistency loss reduces the possibility of generating distorted or unrealistic images and leads to more coherent and recognizable results from the model. The cycle consistency loss is calculated as follows:

$$\mathcal{L}_{cyc}(A2B, B2A) = \mathbb{E}_{a}[||B2A(A2B(a)) - a||_{1}] + \mathbb{E}_{b}[||A2B(B2A(b)) - b||_{1}]$$
(4.2)

where:

- a and b are samples from domain A and B
- A2B is the first generator to convert images from domain A to domain B
- B2A is the second generator to convert images from domain B to domain A
- $||...||_1$  denotes the mean absolute error, or MAE see equation 2.8
- $\mathbb{E}_{a}[||B2A(A2B(a)) a||_{1}]$  is the forward cycle consistency
- $\mathbb{E}_{b}[||A2B(B2A(b)) b||_{1}]$  is the backward cycle consistency

As mentioned earlier, the loss function in CycleGAN is a combination of adversarial losses for each generator-discriminator pair, and the cycle consistency loss. Therefore, the following form is the final loss function:

$$\mathcal{L}_{\text{CycleGAN}}(A2B, B2A, D_A, D_B) = \min_{A2B, B2A} \max_{D_A, D_B} \mathcal{L}_{\text{GAN}}(A2B, D_B, A, B) + \mathcal{L}_{\text{GAN}}(B2A, D_A, B, A) + \lambda \mathcal{L}_{\text{cyc}}(A2B, B2A)$$
(4.3)

where:

- $\mathcal{L}_{GAN}(A2B, D_B)$  and  $\mathcal{L}_{GAN}(B2A, D_A)$  are the adversarial losses (GAN losses), see section 2.4.2
- $\mathcal{L}_{cyc}(A2B, A2B)$  is the cycle consistency loss
- $\lambda$  controls the relative importance of the cycle consistency loss

This loss helps to maintain stability during the training process and ensures that the generated images have a coherent structure that resembles the original images, not just random noise.

## 4.3.4. Implementation

Now that we've seen the main components of CycleGAN, it's time to explore its implementation in more detail. This section covers all the steps necessary to implement a CycleGAN model, including the construction of the generator and discriminator networks, setting up the hyperparameters, and finally, the training process. Before we delve deeper into the implementation process of CycleGAN, let's start with the used libraries, which are the same as those used to implement pix2pix:

```
1 import torch
```

```
2 import torch.nn as nn
```

### 4.3.4.1. Generator network

As previously noted in section 4.3.1, the implementation of the generator's network involves the utilization of an encoder-decoder structure surrounding a several of residual blocks. The encoder network implements one initial block and two downsample blocks, which are used to reduce the input image to a lower-dimensional representation. The encoder is implemented as follows:

```
1
   encoder_layers = [
     # The initial block with the specified parameters
2
3
     nn.ReflectionPad2d(3),
     nn.Conv2d(in_channels=3,out_channels=64,kernel_size=7,stride=1,padding=0),
4
     nn.InstanceNorm2d(64), nn.ReLU(True),
5
     # The first downsample-block with the specified parameters
6
     nn.Conv2d(in_channels=64,out_channels=128,kernel_size=3,stride=2,padding=1),
7
     nn.InstanceNorm2d(128), nn.ReLU(True),
8
     # The second downsample-block with the specified parameters
9
10
     nn.Conv2d(in_channels=128,out_channels=256,kernel_size=3,stride=2,padding=1),
     nn.InstanceNorm2d(256), nn.ReLU(True)]
11
```

### Listing 4.6: CycleGAN encoder

As we see this code snippet 4.6 defines three encoder layers, including one initial block and two downsample blocks. The first block is an initial block that takes a 3-channel input tensor (representing the RGB channels of an image) and applies a reflection padding of size three to the input tensor. This is followed by a 2D convolutional layer with a kernel size of  $7 \times 7$ , 64 output channels, and strides of 1 and 0 padding. Then, an instance normalization layer is applied to the output tensor, followed by a rectified linear unit (ReLU) activation function to produce a 64-channel tensor. The second block is a downsample block that takes the 64-channel output tensor from the first block and applies a 2D convolutional layer with a kernel size of  $3 \times 3$ , 128 output channels, a stride of 2, and padding of 1. An instance normalization layer is then applied to the output tensor, followed by a ReLU activation function, producing a 128-channel tensor. The third block is also a downsample block that takes the 128-channel output tensor from the second block and applies a 2D convolutional layer with a kernel size of  $3 \times 3$ , 256 output channels, a stride of 2, and padding of 1. An instance normalization layer is then applied to the output tensor, followed by a ReLU activation function, producing a 128-channel tensor from the second block and applies a 2D convolutional layer with a kernel size of  $3 \times 3$ , 256 output channels, a stride of 2, and padding of 1. An instance normalization layer is then applied to the output tensor, followed by a ReLU activation function is a stride of 3 × 3, 256 output channels, a stride of 2, and padding of 1. An instance normalization layer is then applied to the output tensor, followed by a ReLU activation function resulting in a 256-channel tensor.

After the encoder, the output is then processed through several residual blocks. Each residual block consists of two  $3 \times 3$  convolutional layers with the same number of filters as the input tensor, two instance normalization layers, and ReLU activation function, for further details refer to section 2.2.2. Each residual block is implemented as follows:

```
1 class ResidualBlock(nn.Module):
```

```
2 def __init__(self, channels):
```

```
3 super().__init__()
```

```
4 # Define the block componants
```

```
5 block = [nn.Conv2d(channels, channels, 3), # first convolutional layer
```

```
nn.InstanceNorm2d(channels),
6
                                                     # instance normalization
7
                nn.ReLU(True),
                                                     # ReLU activation
                nn.Conv2d(channels, channels, 3), # second convolutional layer
8
                nn.InstanceNorm2d(channels)]
                                                     # instance normalization
9
       self.block = nn.Sequential(*block)
10
11
12
     def forward(self, x):
       """Forward function (with skip connections)"""
13
       out = x + self.block(x) # add skip connections to the output
14
       return out
15
```

Listing 4.7: Residual Block

After the residual blocks, the output is then processed through the decoder, which consists of two upsample blocks and one final block. Each upsample block takes the output from the previous layer and increases its spatial resolution through transposed convolutional layers, instance normalization and ReLU activation function. While the final block consists of convolutional layer followed by Tanh activation function. The decoder is implemented as follows:

```
decoder_layers = [
1
2
     # The first upsample-block with the specified parameters
3
     nn.ConvTranspose2d(in_channels=256,out_channels=128,kernel_size=3,stride=2,
4
     padding=1), nn.InstanceNorm2d(128), nn.ReLU(True),
5
     # The second upsample-block with the specified parameters
     nn.ConvTranspose2d(in_channels=128,out_channels=64,kernel_size=3,stride=2,
6
7
     padding=1), nn.InstanceNorm2d(64), nn.ReLU(True),
     # The final block with the specified parameters
8
9
     nn.ReflectionPad2d(3),
     nn.Conv2d(in_channels=64,out_channels=3,kernel_size=7,stride=1,padding=0),
10
     nn.Tanh()]
11
```

### Listing 4.8: CycleGAN decoder

This code snippet 4.8 defines three decoder layers, including two upsample blocks and one final block. The first block is an upsample block that takes a 256-channel input tensor and applies a 2D transpose convolutional layer with a kernel size of  $3 \times 3$ , 128 output channels, stride of 2, and padding of 1. An instance normalization layer was then applied to the output tensor, followed by a ReLU activation function that produced a 128-channel tensor. The second block is also an upsample block that takes the 128-channel output tensor from the first block and applies a 2D transpose convolutional layer with a kernel size of  $3 \times 3$ , 64 output channels, stride of 2, and padding of 1. An instance normalization layer was then applied to the output tensor, followed by a ReLU activation function, resulting in a 64-channel tensor. The third block is the final block that takes the 64-channel output tensor from the second block and applies a reflection padding of size 3 to the input tensor. This is followed by a 2D convolutional layer with a kernel size of  $7 \times 7$ , three output channels (representing the RGB channels of an image), a stride of one, and no padding. Finally, the Tanh activation function was applied to the output tensor.

Now that we have all the components, we can put everything together to build the CycleGAN generator model. You can find the complete generator in the appendix C

### 4.3.4.2. Discriminator network

As previously noted in section 4.3.2 and section 4.2.2, the implementation of the discriminator's network consists of a series of blocks including Convolutional layer, Instance Normalization layer, and LeakyReLU activation layer as follows:

```
discriminator_layers = [
1
     # The first block with the specified parameters
2
     nn.Conv2d(in_channels=3,out_channels=64,kernel_size=4,stride=2,
3
     padding=1), nn.LeakyReLU(0.2, True),
4
     # The second block with the specified parameters
5
     nn.Conv2d(in_channels=64,out_channels=128,kernel_size=4,stride=2,
6
7
     padding=1), nn.InstanceNorm2d(128), nn.LeakyReLU(0.2, True),
8
     # The thierd block with the specified parameters
     nn.Conv2d(in_channels=128,out_channels=256,kernel_size=4,stride=2,
9
     padding=1), nn.InstanceNorm2d(256), nn.LeakyReLU(0.2, True),
10
     # The fourth block with the specified parameters
11
12
     nn.Conv2d(in_channels=256,out_channels=512,kernel_size=4,stride=1,
     padding=1), nn.InstanceNorm2d(512), nn.LeakyReLU(0.2, True),
13
14
     # The final block with the specified parameters
15
     nn.Conv2d(in_channels=512,out_channels=1,kernel_size=4,stride=1,
     padding=1)]
16
```

### Listing 4.9: CycleGAN discriminator layers

As we see this code snippet 4.9 defines five blocks. The first block of the network takes a 3-channel input tensor and applies a 2D convolutional filter with 64 output channels, a kernel size of 4, stride of 2, and padding of 1. The resulting output is then passed through a LeakyReLU activation function with slope of 0.2. The second block takes the output of the first block, which has 64 channels, and applies a 2D convolutional filter with 128 output channels, a kernel size of 4, stride of 2, and padding of 1. The output is then normalized using instance normalization and passed through a LeakyReLU activation function. The third block takes the output of the second block, which has 128 channels, and applies a 2D convolutional filter with 256 output channels, a kernel size of 4, stride of 2, and padding of 1. The output is then normalized using instance normalization and passed through a LeakyReLU activation function. The fourth block takes the output of the third block, which has 256 channels, and applies a 2D convolutional filter with 512 output channels, a kernel size of 4, stride of 1, and padding of 1. The output is then normalized using instance normalization and passed through a LeakyReLU activation function. Finally, the fifth block takes the output of the fourth block, which has 512 channels, and applies a 2D convolutional filter with 1 output channel, a kernel size of 4, stride of 1, and padding of 1. The resulting output is a  $30 \times 30$  with 1-channel tensor, which will used by the adversarial loss to improve the GAN model. You can find the complete discriminator in the appendix C

### 4.3.4.3. Hyperparameters

Hyperparameters are essential in determining the performance of any model and can be adjusted to optimize its training process. In the case of CycleGAN, the following hyperparameters are utilized by default:

- **Learning rate:** The learning rate determines how quickly the model updates its internal parameters during training. A lower learning rate can lead to more precise updates, but can also increase the amount of time it takes for the model to converge on a solution. In the case of CycleGAN, the default learning rate is set to 0.0002.
- **Batch size:** Batch size refers to the number of samples that are processed by the model during each iteration of training. A larger batch size can lead to faster training times, but may also require more memory and can make it difficult for the model to converge on a solution. In the case of CycleGAN, the default batch size is set to 1, meaning that the model processes one image at a time during training.
- **Cycle lambda:** Cycle lambda is a hyperparameter that is specific to CycleGAN. It determines the weight given to the cycle consistency loss during training, which is a measure of how well the model is able to map images between the two sets in both directions. A higher cycle lambda value can lead to a more accurate mapping between image sets, but can also make it more difficult for the model to converge on a solution. In the case of CycleGAN, the default cycle lambda value is set to 10.
- **Optimizer:** The optimizer is a function that determines how the model updates its internal parameters during training. Adam is a popular optimizer that is commonly used in machine learning models. In the case of CycleGAN, Adam is utilized as the default optimizer.
- Adam beta 1 and beta 2: Adam beta 1 and beta 2 are hyperparameters that determine the decay rates for the first and second moments of the gradients during training, respectively. These hyperparameters impact the speed and stability of the model's training process. In the case of CycleGAN, the default values for Adam beta 1 and beta 2 are set to 0.5 and 0.999, respectively.

### 4.3.4.4. Training algorithm

The training algorithm for CycleGAN follows the same training approach as for regular GANs (see section 2.4.1) but with the difference in the loss function and the number of neural networks. In CycleGAN, two generators and two discriminators are used. The generators are trained to translate images from one domain to another, while the discriminators are trained to distinguish between real and fake images.

During training, the generators and discriminators are updated iteratively using adversarial loss and cycle-consistency loss. Adversarial loss ensures that the generated images are indistinguishable from the real images, while cycle-consistency loss ensures that the generated images can be translated back to their original domain without losing information.

Algorithm 2 CycleGAN training algorithm			
1: procedure CycleGAN			
// Initialize discriminators networks			
2: $D_A \leftarrow discriminator_network()$			
3: $D_B \leftarrow discriminator_network()$			
// Initialize generators networks			
4: $G\_A2B \leftarrow generator\_network()$			
5: $G_B2A \leftarrow generator_network()$			
6: <b>for each</b> epoch <b>do</b>			
7: <b>for each</b> batch of images from domain A and domain B <b>do</b>			
// Train discriminator A			
8: $D_A_real \leftarrow D_A(real_A)$			
9: $D_A_fake \leftarrow D_A(G_B2A(real_B))$			
10: $loss_D_A \leftarrow adversarial_loss(D_A_real, target_value = 1.0) +$			
11: $adversarial\_loss(D\_A\_fake, target\_value = 0.0)$			
12: <i>update_weights(loss_D_A, D_A.weights)</i>			
// Train discriminator B			
13: $D_B_{real} \leftarrow D_B(real_B)$			
14: $D_B_fake \leftarrow D_B(G_A2B(real_A))$			
15: $loss_D_B \leftarrow adversarial_loss(D_B_real, target_value = 1.0) +$			
16: $adversarial\_loss(D\_B\_fake, target\_value = 0.0)$			
17: <i>update_weights(loss_D_B, D_B.weights)</i>			
// Train generator A2B			
18: $generated_B \leftarrow G_A 2B(real_A)$			
19: $reconstructed_A \leftarrow G_B2A(generated_B)$			
20: $loss\_cycle\_consistency\_A \leftarrow cycle\_consistency\_loss(real\_A, reconstructed\_A)$			
21: $D_B_fake \leftarrow D_B(generated_B)$			
22: $loss_G_A2B \leftarrow adversarial_loss(D_B_fake, target_value = 1.0) +$			
23: lambaa_weight * loss_cycle_consistency_A			
24: update_weights(loss_G_A2D, G_A2D.weights)			
$\frac{1}{17} \text{ fram generator B2A}$			
25: $generated_A \leftarrow G_DZA(real_D)$			
20: $Ieconstructed_D \leftarrow O_AZD(generated_A)$ 27: $loss cucle consistency B \leftarrow cucle consistency loss(real B reconstructed B)$			
27. $D = A  fake \leftarrow D  A(aenerated A)$			
$D_{1} ukc \leftarrow D_{1} (generation Loss(D, A, fake target value = 1.0) + 10000 + 10000 + 10000 + 10000 + 10000 + 10000 + 10000 + 10000 + 10000 + $			
30: 1000 1000 1000 1000 1000 1000 1000 100			
31: update weights(loss G B2A G B2A weights)			
32: end for			
33: end for			
34: end procedure			

# 4.4. Data Preparation

Dataset preparation is a key aspect of our research on improving of scanned images using GANs. In this section the focus will be on discussing the different techniques used for preparing the dataset for training advanced generative models such as Pix2Pix and CycleGAN. These models rely on large amounts of suitable data for training, and the quality of the final output is directly influenced by the quality of the input data. The process of dataset preparation involves tasks such as data collection, cleaning, augmentation, and normalization, that aim to make the data suitable for training the models.

# 4.4.1. Raw Data Creation:

In order to train our Pix2Pix and CycleGAN models, we created a varied raw dataset of text images in a clear digital PDF format. This dataset contained a mix of automatically generated text and text from uncopyrighted books, presented in different font styles, Languages and sizes. To expand the dataset and diversify it even further, we incorporated old newspaper images sourced from Wikimedia Commons<sup>4</sup>, which are licensed under Creative Commons CC0 and come with OCR digital copies. We also incorporated data from the publicly available Wossidlo Digital Archive<sup>5</sup> as a source for images of old documents. This website provides a collection of historical document images, although this posed some challenges since the data does not have clean pairs. As a result, we only utilized this data for CycleGAN training, as it does not require paired data.



Figure 4.7.: Examples for the raw data

To preprocess the raw data and create the training dataset for the GAN models, we devised a data pipeline with several stages, each serving a distinct purpose in processing the data. Further details regarding this pipeline will be provided in the following section.

<sup>&</sup>lt;sup>4</sup>https://commons.wikimedia.org/wiki/Main\_Page, accessed 18. Feb. 2023

<sup>&</sup>lt;sup>5</sup>https://apps.wossidia.de/webapp/run, accessed 18. Feb. 2023

# 4.4.2. Data Pipeline

An image processing data pipeline is essential for creating suitable training data for models that aim to enhance and clean scanned images. With this in mind, we have developed a flexible and scalable data pipeline that can be easily modified as necessary. Our default pipeline consists of several steps, each serving a distinct purpose in processing the data. To begin, we start with a clean PDF file as the raw data source, from which we extract the images. We use a configuration file to specify various parameters of the pipeline, including the desired image dimensions, as well as the types of augmentations and transformations to be applied. Once the PDF file and configuration file are in place, the data pipeline begins its processing.



Figure 4.8.: The default data pipeline

As shown in Fig. 4.8, the default pipeline composed of four processing steps, which are outlined as follows:

## The First Stage: PDF to Images

The main purpose of this stage is to convert a PDF file into high-quality images that can be utilized as the groundwork for various tasks in the data pipeline. To initiate the conversion process, the desired PDF document is selected, and the configuration file specifies the necessary parameters, such as the number of pages, image resolution, and format. *PyMuPDF* library is used to execute the conversion task, providing various functionalities that include the extraction of pages and converting them to high-resolution image formats, among other functionalities. Converting the PDF into images improves the handling and processing of data, making it more accessible for subsequent stages in the data pipeline.

### The Second Stage: Random Crops

The second stage of the pipeline takes advantage of the high-quality images generated in the first stage and crops them randomly to generate a unified dataset that can be used to train the models later. The number of crops is specified in the configuration file, and these crops serve as ground truth for later stages in the data pipeline. To perform this cropping process, we use the *Albumentations* library, which provides efficient and flexible image augmentation capabilities. Along with cropping, we apply scaling (zoom) to the images, while maintaining the resolution and dimensions of the cropped images. By doing so, we obtain a variety of font sizes in each cropped image, which adds to the diversity of the dataset. To add more realism to the dataset, we apply some Elastic random distortion to the images that have been cropped. This helps

to simulate real-world conditions and ensures that the models can handle various distortions and still produce high-quality images. These cropped images act as the input for subsequent stages in the data pipeline, where they will be further processed to simulate degraded image conditions.



Figure 4.9.: The default data pipeline

### The Third Stage Scanning Effects

The third stage of the pipeline continues the process of simulating degradation and distortion that scanned document images typically undergo, building on the previous stages. To accomplish this, the *Augraphy* library is utilized, which offers a variety of augmentations and transformations that simulate the appearance of scanned images. These effects are specified in the configuration file. The library provides a series of distortion and transformation effects to convert clean source documents into scanned-like images. Initially, the source document is separated into two layers - an ink-layer representing text and graphics, and a paper-layer that can be either white or have a randomly chosen texture. Both layers are then further processed to imitate the effects of paper printing, faxing, scanning, and copying, generating multiple variations from a single source document, resulting in numerous realistic and degraded document images. Once both the ink and paper layers have been completed, the ink is applied to the paper with the desired effects. The resulting merged document image is then subject to additional distortions, such as the addition of folds or other physical deformations that are determined by the interplay between the paper and ink layers.



Figure 4.10.: Examples for Scanning Effects

### The Fourth Stage: Imperfections Effects

In the fourth stage of the document image degradation pipeline, the aim is to introduce additional degradation and imperfections to the scan-like images generated in the previous stages. To achieve this, various degraded paper textures and coffee and ink stains are added, which are specified in the configuration file and randomly applied to each image. These textures and stains are from Creative Commons sources, generated using modern text to image artificial intelligence techniques like DALL·E 2 and stable diffusion, or created manually.



Figure 4.11.: Examples for degraded paper textures and coffee stains

To ensure that the models can better learn and generalize in real-world situations, blending techniques such as multiply blend mode and darken mode with various opacities are utilized to smoothly merge defects with the third stage layers. This stage adds a variety of textures and stains that mimic the natural aging and deterioration of documents over time, resulting in a more diverse and realistic set of degraded document images. As a result, the models can better handle a wider range of real-world scenarios, including images that have been impacted by environmental factors or natural aging. Furthermore, the blending techniques used in this stage produce a more cohesive image, where the degradation and distortion blend seamlessly with the overall picture. This enhances the models' ability to perceive and distinguish between the genuine text and the degradation present in the image.



Figure 4.12.: Examples for the default pipeline outputs

Once the data is prepared, it is split into two distinct datasets, training and testing datasets. The training dataset is used for training the models, while the test dataset is used to evaluate the performance of the models and fine-tune them as necessary, for further details about testing datasets and evaluation refer to chapter 5. Additionally, the data pipeline can produce both paired and unpaired datasets for a variety of computer vision tasks, such as supervised and unsupervised image-to-image translation (refer to section 2.5). This data can be used to train various machine learning models, such as Generative Adversarial Networks (GANs) to learn the correlation between degraded and clean images and generate high-quality images from degraded inputs.

## 4.4.3. Data Preprocessing

Data preprocessing is a critical step in any machine learning project, especially when dealing with image data. In this section, we will discuss the preprocessing steps employed to prepare the data for training and testing/using the Pix2Pix and CycleGAN models. The pre-processing steps for both models consist of image resizing, cropping, and normalization. These steps are necessary to ensure that the images have a uniform size, are centered around the area of interest, and are normalized to a common scale. The goal of this process is to standardize the data and eliminate any unnecessary variations that may hinder the performance of the models.

In order to prepare the data for training in Pix2Pix and CycleGAN, various preprocessing steps are employed for both image domains. The initial step involves resizing and cropping both images to a fixed size of  $256 \times 256$  pixels. Afterward, the pixel values are normalized to ensures that the output is mapped to a value range of [-1, 1]. This step is critical in improving neural network convergence by giving all features an equal range of values and preventing some from dominating others due to wide value ranges. In addition to normalization, another preprocessing technique used involves applying random horizontal or vertical flipping to the images. This technique creates a new set of images that improve the model's ability to generalize to unseen data. Overall, these preprocessing steps are critical in standardizing the data and eliminating any unnecessary variations that may hinder the performance of the models.

It should be noted that pre-processing consists of two types: one for training data and one for test/user data. The standardized pre-processing process is applied to the training data, treating each sample as a patch and using it to train the models. However, it is not possible to pre-process the test/user data as one patch for the entire sample, as real samples have varying dimensions, not just  $256 \times 256$ . To maintain data integrity and adhere to the same principles of training data pre-processing, the test/user data is divided into patches that are proportional in size to the training data. These patches are processed separately and then fed into the model one by one. Once all patches are processed, they are combined again. This ensures that the test/user data undergoes the same pre-processing steps as the training data and that its dimensions remain unchanged.

# 4.5. Prototype

In this section, we will present a prototype that demonstrates the usage of the GAN models we have trained to remove imperfections and improve scanned images. The prototype offers a simple and intuitive user interface that enables users to effortlessly upload their scanned images and apply pre-trained GAN models on them. It also provides several other useful features, including the option to compare the original and enhanced images side-by-side and the ability to save the improved images. Furthermore, we will delve into other features of the prototype in more detail later in this section. The prototype was developed using *Streamlit*, a Python-based framework for building interactive web applications for machine learning and data science.



Figure 4.13.: Prototype Interface

As we can see in Fig. 4.13, The prototype interface is partitioned into two primary components to enhance the user's experience. The first component is the Sidebar that contains a range of functionalities that the user can access. On the other hand, the second component is the main viewing area, which is designed to show the input and processed images. This area is the primary focus of the prototype interface as it allows the user to view the images they are working on and monitor the progress of their work. By partitioning the prototype interface into these two primary components, the user can easily navigate and use the various features of the interface. The sidebar provides quick access to all the functionalities, while the main display area serves as the central workspace.

# 4.5.1. The Sidebar

As mentioned, the sidebar houses all available functionalities to interact with the software. These functionalities could be in the form of buttons, drop-down menus, or any other graphical representation that allows users to interact with the software. Typically, the sidebar is located on the left side of the interface, and it's designed to be easily accessible to the user.

The available functionalities are as follows:

- 1. **File uploader:** With this feature, users can upload their own files to the program for processing, with each file limited to a size of 200MB. The files can be selected by clicking on the "browse files" button or by dragging and dropping them to the designated location. Multiple files can be uploaded simultaneously, and the feature supports a variety of file extensions such as PDF, PNG, JPEG, BMP, and TIFF.
- 2. **Model selector:** This feature allows users to select a pre-trained GAN model that they wish to apply to the uploaded files from a drop-down list. To make the pre-trained models available in the list, they must be added to the "pre-trained models" folder within the project.

File uploader	Model selector	
Upload files	Select a pre-trained model	
Drag and drop files here Limit 200MB per file • PDF, PNG, JPG, JPEG, BMP, TIFF	Model 0  Model 1	
Browse files	Model 2	
	Model 3	
(a) File uploader	(b) Model selector	

Figure 4.14.: Prototype Sidebar File uploader and Model selector

3. **Automatic adjustments:** This feature enables the user to automatically adjust the uploaded images by applying certain modifications. These adjustments are converting the image to grayscale or applying the CLAHE algorithm, which can help to avoid over-amplification of contrast. The purpose of these automatic adjustments is to assess whether they enhance the quality of the output generated by the model. However, the effectiveness of these adjustments may vary depending on the condition of the image that was entered. In some cases, the automatic adjustments may result in improved results, while in other cases it may lead to worse results.

Automatic adjustments			
🗹 Black and white			
Auto Enhance			

Figure 4.15.: Prototype Sidebar Automatic adjustments

4. **Manual adjustments:** By means of this feature, users can manually fine-tune the uploaded images by manipulating different variables, such as Sharpness, Color, Brightness, and Contrast. The user can adjust these variables using a slider that can be dragged left

or right until they achieve the desired result. In addition, the user can rotate the image in multiple angels (0, 90, 180, and 270 degrees) using a drop-down list. The purpose of making these modifications is to identify the optimal combination of values that can improve the quality of the outputs produced by the model.

Manual adjustments	
Rotate Image	
0 degrees 🗸	Brightness
Sharpness 0.00	0.00 10.00
0.00 10.00	1.00
Color 1.00	0.00 10.00
0.00 10.00	Reset sittings

Figure 4.16.: Prototype Sidebar Manual adjustments

5. **Download:** This feature allows the user to download all the processed and enhanced images in the form of a zip file.

## 4.5.2. The Main Viewing Area

The primary section of the prototype's interface is the main viewing area, where the user can observe the images they have uploaded and the outcomes of any processing that has been carried out on these images. This area is the most significant element of the interface and is composed of two tabs. The first tab, titled "Main", exhibits the uploaded images on the left side and the transformed images on the right side. The second tab, called "Comparison," enables the user to scrutinize the images more closely by using a slider that can be shifted to the left to view the output, or to the right to view the input, enabling the user to compare the results effectively.



Main tab

Comparison tab

Figure 4.17.: Prototype Main Viewing Area
# 4.6. System Architecture

In this section, we will present the architecture of the system and the workflow between its main components. The system comprises four distinct parts: Data Preparation, Training, Testing, and the Interface. Each section serves a particular purpose in accomplishing the system's ultimate objective. We will discuss how data flows within the system and the processes involved in each section.



Figure 4.18.: System Architecture

As we see in Fig. 4.18 the first component of the system is the data preparation section, which begins with processing the raw data using the data pipeline to generate data in the required format for the task (see section 4.4). Once the data is processed, it is divided into two parts: the first part is used for training, while the second part is reserved for testing. The training data is then directed to the training section, where it undergoes preliminary processing to prepare it for training (see section 4.4.3). Subsequently, the model trains on this processed data. After completing the training process, the trained model flows into the testing section for evaluation, and the Interface section where users can leverage the pre-trained model to immediately use the results.

# 5. Evaluation setup

In this chapter, we will outline the evaluation setup for the models. This procedure aims to determine the effectiveness of these methods in improving scanned images. When evaluating the improvement of scanned images, it is important to take into consideration factors such as image resolution and the preservation of fine details in the translated images. Therefore, the evaluation will consider three key elements: testing datasets, evaluation metrics, models and training details. The testing datasets will be carefully selected to reflect a diverse range of scanning scenarios and ensure that the evaluation results are representative of real-world situations. The evaluation metrics used will quantify the effectiveness of the model in terms of visual quality, preservation of important details, and overall improvement of the scanned images. Finally, we will describe the models and training details, including the dataset used for training, the optimization algorithm, and the hyperparameters used in the experiments.

## 5.1. Testing Dataset

Testing datasets are essential component of evaluating the performance of our study. A paired dataset is required for the evaluation as it allows us to assess the performance of the model objectively and draw meaningful conclusions about the quality of the generated images by comparing them to real images, thus providing a basis for determining the quality of the generated images and measures the model's ability to maintain important features like color, texture, and structure.

In our evaluation process, we will generate two types of test datasets, each containing 200 paired images of various languages, fonts, and sizes, for each scenario in our evaluation process (see section 6.1). These datasets will be produced using a data pipeline and supplemented with images from the kaggle Denoising Dirty Documents dataset<sup>1</sup>. This approach will enable us to evaluate our model's performance in handling various types of input data, such as documents with different languages, styles, and layouts. Additionally, the use of such a diverse dataset will allow us to assess the model's generalizability and efficiency in processing different types of documents.

By evaluating the model's performance under such diverse conditions, we will gain a more comprehensive understanding of its capabilities and limitations, which will help us make informed decisions about the quality of the generated images. Overall, the use of a diverse test dataset is essential for obtaining a more accurate assessment of our model's performance and generalizability.

<sup>&</sup>lt;sup>1</sup>https://www.kaggle.com/c/denoising-dirty-documents, accessed in 26. Feb. 2023

## 5.2. Quantitative Evaluation Metrics

Evaluating the performance of Generative Adversarial Networks (GANs) for image-to-image translation tasks can be challenging, as the success of these models is often measured by subjective quality of the generated images. To address this issue, several quantitative evaluation metrics have been proposed. Some of the most commonly used metrics in this context include:

#### • Peak Signal-to-Noise Ratio (PSNR):

PSNR is a commonly used image quality metric that measures the difference between a reference image (ground truth) and the generated image. By expressing this difference in decibels (dB), PSNR aims to quantify the amount of noise and degradation in image quality as a result of lossy compression or other processes that result in a reduction of the image's structural and visual information [42]. PSNR is calculated as follows:

$$PSNR(x, y) = 20 \times \log_{10}\left(\frac{MAX_x}{\sqrt{MSE(x, y)}}\right)$$
(5.1)

where:

- $MAX_x$  is the maximum possible pixel value that exists in our original image.
- *MSE* is the Mean Squared Error between the reference image and the reconstructed image see equation 2.1.2

It's important to note that while PSNR can be a useful metric for assessing image quality in some cases, it has several limitations, including being sensitive to only the magnitude of the error and not to its distribution. Additionally, high PSNR values do not necessarily guarantee good image quality in a perceptual sense [43]. Therefore, PSNR should be used in combination with other metrics.

#### Structural Similarity Method (SSIM):

SSIM is another widely used image quality metric that measures the structural similarity between two images. It provides a perception-based measure of the similarity between two images by considering the structural information, luminance information, and contrast information [44]. SSIM is calculated as follows:

$$SSIM(x,y) = \frac{(2\mu_x\mu_y + c_1)(2\sigma_{xy} + c_2)}{(\mu_x^2 + \mu_y^2 + c_1)(\sigma_x^2 + \sigma_y^2 + c_2)}$$
(5.2)

where:

- $\mu_x$  and  $\mu_y$  are the means of images x and y
- $\sigma_x$  and  $\sigma_y$  are the standard deviations of images x and y
- $\sigma_{xy}$  is the cross-covariance of images x and y
- $-c_1$  and  $c_2$  are small constants used to stabilize the division

SSIM index ranges from -1 to 1, with a value of 1 indicating perfect structural similarity between the two images and a value of -1 indicating no structural similarity [44]. However, it's important to keep in mind that like PSNR metric, the use of SSIM as a sole metric for evaluating image quality is limited, and it's recommended to combine it with other metrics.

#### • Fréchet Inception Distance (FID):

FID is a evaluation metric for Generative Adversarial Networks (GANs) introduced in [45]. It measures the distance between the feature representations of real and generated images in an intermediate layer of a pre-trained neural network called Inception-v3. A lower FID score indicates a higher level of similarity between the real and generated images and is therefore desirable [45, 46]. The calculation of FID involves the following steps:

- 1. Extraction of features of real and generated images from an intermediate layer of the pre-trained model Inception-v3. These features are used to obtain a multivariate Gaussian distribution for each set of images.
- 2. Calculation of means and covariance matrices for both the real and generated images.
- 3. Calculation of Fréchet distance between the two multivariate Gaussian distributions using the means and covariance matrices. The Fréchet distance is a measure of the similarity between the two distributions and can be thought of as the maximum difference between the two distributions along all possible alignments. It can be calculated as follows:

$$FID = ||\mu_r - \mu_g||^2 + Tr(C_r + C_g - 2(C_r * C_g)^{\frac{1}{2}})$$
(5.3)

where:

- $\mu_r$  and  $\mu_q$  are the feature-wise mean of the real images and generated images
- $C_r$  and  $C_g$  are the covariance matrix of the real images and generated images
- $||\mu_r \mu_g||^2$  sum squared difference between the two means
- *Tr* is the trace of a matrix.

FID has been shown to be a more robust evaluation metric compared to other metrics, as it considers both the distribution and quality of the generated images [47]. However, it's important to keep in mind that like any evaluation metric, FID has its limitations, and it's best to use it in combination with other metrics and visual inspection to make an overall evaluation of the quality of generated images.

As previously stated, relying on any single metric alone is inadequate for a complete evaluation of the GAN model's performance in image-to-image translation task. Therefore, a comprehensive evaluation should consider all previous metrics that can uncover issues like mode collapse or instability during training. Furthermore, in addition to the quantitative metrics, it's also important to consider the visual inspection of each of the images generated during the evaluation process. This inspection can provide valuable information about the performance of the models under various input conditions.

## 5.3. OCR Evaluation Metrics

Ensuring the reliability and accuracy of the resulting text is crucial for OCR systems. In our study, we aimed to assess the effectiveness of the Pix2Pix and CycleGAN models in enhancing the performance of OCR systems. To evaluate the accuracy of these systems, two commonly used metrics are character error rate (CER) and word error rate (WER).

#### • Character Error Rate (CER):

CER measures the percentage of character-level errors in the extracted text compared to the ground truth. It is calculated as the total number of substitutions, deletions, and insertions divided by the total number of characters in the ground truth [48]. The formula for CER is:

$$CER = \frac{(S+D+I)}{N}$$
(5.4)

where:

- *S* is the number of substitutions (incorrect characters substituted for correct ones)
- *D* is the number of deletions (correct characters missing from extracted text)
- *I* is the number of insertions (incorrect characters added to the extracted text)
- N is the total number of characters in the original text

#### • Word Error Rate (WER):

WER measures the percentage of word-level errors in the extracted text compared to the ground truth. It is calculated as the total number of substitutions, deletions, and insertions divided by the total number of words in the ground truth [48]. The formula for WER is:

$$WER = \frac{(S_w + D_w + I_w)}{N_w}$$
(5.5)

where:

- $S_w$  is the number of substitutions (incorrect words substituted for correct ones)
- $D_w$  is the number of deletions (correct words missing from extracted text)
- $I_w$  is the number of insertions (incorrect words added to the extracted text)
- $N_w$  is the total number of words in the original text

Both CER and WER range from 0 to 1, with a lower score indicating higher accuracy. However, it is worth noting that CER and WER may exceed 1 in cases where the extracted text contains more characters or words than the original text.

# 5.4. Models and Training details

As we delve deeper into our experimental setup, it's essential to consider the various factors that influenced the performance of our models. This includes the training datasets, architecture of the models, hyperparameters, and other relevant details that contributed to their performance. Understanding these specifics will give us a better understanding of how the models were trained and how they were able to achieve their results.

- **Training data:** All models were trained on two distinct datasets, each containing 10,000 images in each respective domain. The pix2pix model's dataset was paired, meaning each image had a corresponding target image. Conversely, the CycleGAN dataset was unpaired, with no such corresponding images. Prior to training, all images underwent pre-processing, which involved resizing them to 256 x 256 and normalizing their pixel values. The first dataset consisted of text-only images, with the aim of train the model to generate clean and binary outcomes, i.e. black and white images. The second dataset, on the other hand, contained color images, which were utilized to train the model to generate clean results while maintaining the color of the images.
- **Generator architectures:** Both of the models were trained using two different generator architectures in order to evaluate the impact of architecture on performance. The first architecture was the original U-Net generator, which consisted of 8 encoder blocks and 8 decoder blocks that were connected through skip connections (see section 4.2.1). The second architecture was inspired by the CycleGAN generator (see section 4.3.1) with 9 residual blocks.
- **Discriminator architecture:** Both of the models were trained using 70 × 70 PatchGAN (see section 4.2.2 and 4.3.2)
- Normalization layers: Both of the models were training with the use of two distinct normalization layers, namely batch normalization and instance normalization (see section 2.1.4). The purpose of this was to assess the effect of the normalization layers on the overall performance of the models.
- Loss functions: Both models were trained using three different adversarial loss functions in order to investigate the impact these functions have on the results.
- **Optimization method:** All the models were trained using the Adam optimizer (see section 2.1.3) with a learning rate of 0.0002 and beta values of (0.5, 0.999).
- **Batch size:** All the models were trained with the batch size set to 1.
- **Epochs:** The training procedure was terminated after completing 100 epochs, with periodic saving occurring every 5 epochs.

# 6. Experiments and Results

In this chapter, we will examine the experiments performed in this study and showcase the results achieved based on the evaluation metrics, testing datasets, and training details outlined in the previous chapter.

# 6.1. Experiments scenario

Our upcoming experiments will evaluate the model's performance in an essential scenario, which centers on binarization. Binarization entails the conversion of scanned, degraded images into clear black and white images that preserve all crucial details. The objective of this scenario is to assess the model's ability to clean text images effectively, which is particularly relevant for applications such as Optical Character Recognition (OCR) systems that require high-quality images to accurately extract text. The effectiveness of a binarization technique can be determined by its ability to improve text visibility in such images and eliminate any unwanted artifacts or noise.

## 6.2. Results

The results of this study demonstrate the differences between Pix2Pix and CycleGAN in improving scanned images under the aforementioned scenario. We evaluated the performance of both models using three metrics: PSNR, SSIM, and FID, which were introduced in section 5.2. The comparison of the two models was based on quantitative measures as well as visual inspection of the output images and OCR recognition. The aim of this study was to investigate which model performs better and to provide insights into their respective strengths and limitations. This section presents the results of our experiments, which include a detailed analysis of the performance of Pix2Pix and CycleGAN in improving scanned images under different hyperparameters.

In this report, we aim to present the outcomes of our experimentation conducted under the Binarization scenario. Our study was designed to examine the impact of various hyperparameters on the performance of each model. To ensure a fair comparison between the models, we used the same hyperparameters in each test case. The primary objective of our study was to determine the optimal combination of model and hyperparameters that would produce the best results for improving the quality of scanned images in the Binarization scenario. To achieve this, we performed six separate test cases, each of which aimed to investigate the effect of specific parameters on the models' performance. By analyzing the outcomes of these tests, we were able to provide valuable insights into the most effective approach to enhancing image quality in this specific scenario.

#### Test Case 1:

In this test case, we aim to compare the performance of the two models, Pix2Pix and Cycle-GAN, in enhancing the quality of scanned images, with a specific focus on their generator architectures, which are listed as follows:

- 1. A ResNet featuring 9 residual blocks, with a total of 11.378 million parameters.
- 2. A U-Net with 8 blocks in the encoder and 8 blocks in the decoder, comprising a total of 54.410 million parameters.

Our primary objective is to assess the models' capacity to eliminate noise, artifacts, and other imperfections, such as coffee stains, from the images. To ensure a fair comparison, we trained both models using the same set of hyperparameters, which are provided below:

General Hyperparameter	
Adversarial loss function	Min-Max GAN loss
Normalization type	Batch normalization
Discriminator architecture	70 × 70 PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Learning rate	0.0002
Special Hyperparameter	
Pix2Pix L1 Lambda	100
CycleGAN cycle Lambda	10

To assess the effectiveness of the models, we tested them on the testing dataset as described in Section 5.1. This dataset includes images with varying levels of noise, degradation, and imperfections as well as images of different sizes and resolutions. To evaluate the performance of the models, we used the metrics introduced in Section 5.2.

#### **Quantitative Comparison:**

Initially, we present the results of the quantitative measures applied to assess the models' performance. Our evaluation was based on various established metrics, as introduced in Section 5.2, which are widely employed in image-to-image translation. These metrics provide an objective assessment of the capacity of the model to convert images from one domain to another and enable the evaluation of its performance in comparison to a ground truth or a set of recognized labels. By analyzing these metrics, we can gain a more comprehensive understanding of how well the model is operating and identify any areas that require improvement.

To visualize the performance of the model over time, we created line charts that displayed its performance every five epochs during the 100-epoch training period for each metric. This approach enables us to efficiently compare models and identify the ones that perform better.

Figure 6.1 provides a clear representation of the progress of the model throughout the training process, as measured by the PSNR metric (higher is better).



Figure 6.1.: PSNR line chart (higher is better) for test case 1

We observed that the performance of the models varied significantly across epochs. Upon initial inspection, it is evident that the Pix2Pix model outperforms the CycleGAN model in both generator architectures.

The Pix2Pix with ResNet model stands out with the highest PSNR score across all epochs, reaching a maximum score of 21.96 dB at epoch 80. On the other hand, the CycleGAN with U-Net and ResNet models consistently show the lowest PSNR scores, with some exceptions.

The results indicate that increasing the number of epochs does not always lead to a better model performance, as demonstrated by the decreasing PSNR scores in later epochs. The choice of generator architecture also appears to play a crucial role in determining the performance, with the ResNet architecture often delivering superior performance compared to the U-Net architecture. It is worth noting that the Pix2Pix models maintained a relatively stable performance rate during training, indicating that they converged quickly. In contrast, the performance of the CycleGAN model was unstable. For instance, while CycleGAN with ResNet achieved acceptable results at epochs 5, 10, 25, 30, 40, and 50, it eventually collapsed after epoch 50. Similarly, the CycleGAN with U-Net volatile results in the earlier epochs and then experiences a decline in performance after epoch 70.

In conclusion, the PSNR results suggest that the Pix2Pix with ResNet model is better suited for the task at hand than the other models. However, to determine the models that can deliver maximum performance with the given hyperparameters, it is necessary to assess the remaining metrics. While the PSNR metric only measures the amount of image noise and may not always accurately represent image quality, the SSIM metric considers luminance information, contrast information, and structural information to assess the similarity in structure between images.

Figure 6.2 provides a clear representation of the progress of the model throughout the training process, as measured by the SSIM metric (higher is better).



Figure 6.2.: SSIM line chart (higher is better) for test case 1

It is evident that the SSIM results closely resemble PSNR results. Model performance varies based on the type of model and the number of epochs used, and increasing the number of epochs does not always improve model performance. The Pix2Pix model consistently performs the best, achieving high SSIM values across all eras, while the CycleGAN model is generally unstable and performs poorly.

In terms of the generator architecture, the Pix2Pix model with ResNet consistently outperformed the other models, achieving SSIM values close to 0.95. The Pix2Pix model with U-Net also performs well, achieving SSIM values ranging from 0.90 to 0.92 for most ages. For CycleGAN, both U-Net and ResNet architectures show similar behavior as in PSNR, with SSIM values ranging from 0.5 to 0.9 across different epochs. It is worth noting that CycleGAN with ResNet collapsed after epoch 50 and CycleGAN with U-Net after epoch 70.

In conclusion, the results obtained from the SSIM indicate that the performance of the model is comparable to that of the PSNR results, but with higher values. This suggests that although the generated image may have a greater degree of structural similarity to the ground truth image, there may still be some noise or distortion that negatively affects its PSNR score. While PSNR and SSIM are reliable metrics for evaluating image quality on an individual basis, we chose to use a third metric, namely, the Frechet Inception Distance (FID), to assess the overall performance of our models. This is because PSNR and SSIM measure image quality individually and then calculate the mean value across the entire dataset, whereas FID considers the distribution of the generated images as a whole, as compared to the distribution of the real images, providing a more comprehensive evaluation of the models' overall performance.

Figure 6.3 provides a clear representation of the progress of the model throughout the training process, as measured by the FID metric (lower is better).

As we can see, Figure 6.3 clearly indicates that the performance of the models follows a similar pattern to the previous two metrics, except for FID, where lower values indicate better results.



Figure 6.3.: FID line chart (lower is better) for test case 1

This particular metric is highly reliable in identifying the best models, as it presents a more distinct view of the differences in results, where the good models produce very low values, while the bad models produce very high values.

After analyzing the results of all three metrics, we can confidently conclude that the best model under these parameters is Pix2Pix with ResNet, closely followed by Pix2Pix with U-Net. However, it is noteworthy that the CycleGAN models did not exhibit stability in this test case.



Figure 6.4.: Test case 1

Figure 6.4 shows the mean of all epochs for each model, giving us a snapshot of the quality of the models which yielded the following results:

Model	Mean PSNR ↑	Mean SSIM ↑	Mean FID $\downarrow$
Pix2Pix with ResNet	21.4998	0.9423	4.0701
Pix2Pix with U-Net	19.1614	0.9151	6.6716
CycleGAN with ResNet	12.3379	0.6389	162.7294
CycleGAN with U-Net	12.3156	0.6444	143.5283

Table 6.1.: Test case 1 results table

#### **Visual Inspection:**

While quantitative metrics provide a useful means of evaluating the performance of models, they often fail to capture details that can only be detected through visual inspection. Therefore, after we have carried out a quantitative evaluation, we will visually inspect different samples to confirm the performance of the models. In these tests, we review only the best epochs, so the models used are as follows: Pix2Pix with ResNet at epoch 80, Pix2Pix with U-Net at epoch 90, CycleGAN with ResNet at epoch 25, and CycleGAN with U-Net at epoch 15. The following tests show the results of each model for the same inputs:



#### Input



#### Ground truth

t ut labore ( Lorem ipsum dolor sit amet, con elitr, sed diam nonumy eimod te labore et dolore magna alcuvari Voluptua. At vero ess efaccusari dolores et ea rebum. Stet citta ka sea takimata sanctus est Lorem amet. Lorem ipsum dolor sit am asadipscing elitr, séd diam nonumy eimod tempor invidunt ut labope et dolore magna aliquyam erat sed diam voluptua. At vero ess et accusam et usto du dolores de ae rebum. Stet citta kas te diam voluptua. At vero ess et accusam et usto du dolores de ae rebum. duo dolores et ea rebum. Stet clita kasd gud duo dolores ét ea rebum. Stet cilta kasd gubergren, no sea takimata sanctus est. Lorem isysti dolor sit amet. Lopem ipsum dolor sit amet, consetetur sadipsche giltr. sed ciam norumy eirmod invidunt ut labore at dolore magna aliquyam erat, sed diam voluptua. At vero dos et accusam et justo duo dolores et a jebum. Stet cilta kasd gubergren, no seal takimata singrús est Lorem ipsum dolor sit amet.

#### Pix2Pix with ResNet

Arial Lorem insum dolor sit amet, com ellit, sed diam nonumy einyöd te labore et dolore magna alguvan volupua. At vero eos et accusan dolores et ea rebum. Stet cita kat sea takimata sanctut est Lorem amet. Lorem insum dolor sit amet sadipscing elit, sed diam nonumy einyöd temoor invidunt ut laboe et dolore magna aliquyam erat. sed diam volutua. At vero eos et accusam et fusto duo dolores et ea rebum. Stet cita kas dubergren, no sea takinata sanctus est Lorem insum dolor sit amet. Lorem egna aliquyam erat. sed diam volutua. At vero eos et accusam et fusto duo dolores et ea rebum. Stet cita kas dubergren, no sea takinata sanctus est lorem insum dolor sit amet. Lorem egna aliquyam erat. sed diam volutua. At vero eos et accusam et justo duo dores et ea rebum? Stet cita kas dubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Arial t ut labore -

#### Pix2Pix with U-Net



# CycleGAN with ResNet Ariəl Arial tut labore e Lorem insum dolor sit amet. oon elitr, eed diam. nonumy eimod tem **voluntu** a labore :: dolore maana afduyar olopua, At vero eo: a/accusar doloes et aerbum. Site dirak in **ata imata sa** sad ispcina elitr sed diam nonumy eimod temcor invidunt ul labore et aciore magna aliquyam erat sed diam volucu - At vero eos et accusam et usto duo dolores et ae: ebun Site clita ka du gebergren, no sea takinata sanctus est Lorem inpun dolor sit amet. Lorem insum dolor sit amet, consetetur sed diam volucu - At vero eos et accusam et usto duo dolores et ae: ebun Site clita ka du gebergren, no sea takinata sanctus est Lorem inpun dolor sit amet. Lorem insum dolor sit amet, consetetur sed diam volucut. At vero eos et accusam et justo duo dolores et ea: ebun Site clita ka 'sd gubergren, no sea takinata saperus est Lorem ipsum dolor sit amet. t ut labore г

#### CycleGAN with U-Net





Note that the red lines are not originally present in the images and have been added to highlight specific details.

The aforementioned examples serve as a representation of the various samples that were tested. For further samples, we kindly direct you to the appendix B where additional samples are available for review.

#### OCR recognition:

Optical Character Recognition (OCR) is an important technology that enables computers to interpret and convert scanned images into machine-readable formats. Tesseract OCR, which is an open-source OCR engine<sup>1</sup> frequently used for text-recognition tasks, may encounter difficulties with low-quality scanned images. Thus, the purpose of this section is to demonstrate the performance of a models on diverse test samples, evaluating its capacity to improve Tesseract OCR's accuracy in text extraction, using two metrics: character error rate (CER) and word error rate (WER), which were introduced in Section 5.3.

Sample: sed diam voluptua. At vero eosret accusiam et justo

• Source image:

sed diam voluptua. At vero eos et accusam et justo

Extracted text: . voluptua. t – Character Error Rate: 0.74 Word Error Rate: 0.9

#### • Pix2Pix with ResNet:

#### sed diam voluptua. At vero eos et accusam et justo

Extracted text: sed diam voluptua. At vero eosfet accusiam et justo Character Error Rate: 0.04 Word Error Rate: 0.3

#### • Pix2Pix with U-Net:

### sed diam voluptua. At vero eos et accusam et justo

Extracted text: sed diam voluptua. At vero eosret accusamet justo Character Error Rate: 0.04 Word Error Rate: 0.4

#### • CycleGAN with ResNet:

### sed diam voluptua. At vero eos et accusam et justo

Extracted text: sed diam voluptua. At vero eos'et accusam et justo Character Error Rate: 0.02 Word Error Rate: 0.2

#### • CycleGAN with U-Net:

sed diam voluptua. At vero ecos en accusam et justo

Extracted text: sed diam voluptua. At vero ea.-.7 =. accusam ct iusto Character Error Rate: 0.18 Word Error Rate: 0.4

<sup>&</sup>lt;sup>1</sup>https://github.com/tesseract-ocr/tesseract

Similar to the example provided earlier, we computed the mean of Character Error Rate and Word Error Rate for 100 different sentences, and obtained the subsequent outcomes for each model:

Model	Mean Character Error Rate $\downarrow$	Mean Word Error Rate $\downarrow$
Source	0.6435	0.8125
Pix2Pix with ResNet	0.0584	0.2807
Pix2Pix with U-Net	0.0947	0.3930
CycleGAN with ResNet	0.1097	0.4045
CycleGAN with U-Net	0.2926	0.5619

Table 6.2.: Test case 1 OCR recognition results table

As shown in Table 6.2, we evaluated the performance of five different cases on 100 different sentences using mean CER and mean WER. The first row represents the performance of the source image, which is the original image without any improvement. The remaining four rows represent the performance of four different models: Pix2Pix with ResNet, Pix2Pix with U-Net, CycleGAN with ResNet, and CycleGAN with U-Net.

We can observe that the model with the highest performance is Pix2Pix with ResNet, which achieves a mean CER of 0.0584 and a WER of 0.2807. This outcome suggests that the model successfully replicated the original text with few errors. Pix2Pix with U-Net also demonstrated good performance, achieving a CER of 0.0947 and a WER of 0.3930, although it was not as effective as Pix2Pix with ResNet.

However, the CycleGAN models had lower accuracy than the Pix2Pix models. CycleGAN with ResNet had a CER of 0.1097 and WER of 0.4045, and CycleGAN with U-Net produced the highest error rates with a CER of 0.2926 and WER of 0.5619. This suggests that the CycleGAN models were less effective in enhancing the scanned images for optical character recognition (OCR), resulting in less accurate text extraction.

In conclusion, the table indicates that the Pix2Pix models with ResNet and U-Net were more effective in improving OCR accuracy than the CycleGAN models. These results suggest that the Pix2Pix models are better suited for this particular task and more appropriate for enhancing the OCR accuracy in similar situations.

#### **Best Model:**

Overall, the Pix2Pix models demonstrated remarkable proficiency in enhancing the quality of scanned images and rectifying imperfections, whereas the CycleGAN models exhibited erratic performance in numerous instances. Among all the tested models, the Pix2Pix model incorporating the ResNet architecture in the generator produced the most exceptional outcomes in all test cases (see Appendix A), with minor variances in each case. Hence, it can be deduced that the most effective model for improving scanned images and eliminating imperfections is the Pix2Pix model with the following configurations:

General Hyperparameter	
Adversarial loss function	LSGAN or Min-Max GAN loss
Normalization type	Instance normalization
Generator architecture	ResNet with 9 residual blocks
Discriminator architecture	70 × 70 PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Learning rate	0.0002
Special Hyperparameter	
Pix2Pix L1 Lambda	100

For more details, see Appendix A which contains a comprehensive set of test cases that examine the impact of different loss functions and hyperparameters on the performance of the models. These test cases were conducted to provide a thorough analysis of the models and to determine which combinations of loss functions and hyperparameters would yield the best results. By examining a range of different cases, we were able to gain a better understanding of how the models would perform under various perparameters. The results of these tests are presented in detail in the appendix and provide valuable insights into the behavior of the models. Overall, the inclusion of these additional test cases enhances the robustness and comprehensiveness of our analysis.

# 7. Conclusion and Future work

This chapter summarizes the main findings of this study and provides suggestions for future research. The conclusion section revisits the research questions and objectives and provides a brief overview of our findings. The future work section outlines the potential research directions that could build upon the current study's findings and contribute to a deeper understanding of the topic.

# 7.1. Conclusion

In conclusion, this research aimed to investigate the effectiveness of using generative adversarial networks (GANs) to enhance the quality of scanned documents that have been damaged or contain imperfections, while preserving their content.

The research addressed two central questions:

- 1. What is the potential effectiveness of using Generative Adversarial Networks (GANs) to enhance the quality of scanned documents that have been damaged or contain imperfections while preserving their content?
- 2. Can this methodology be considered a practical preprocessing technique for preparing documents for digital archiving?

To address these questions, this study sought to explore two types of GANs, Pix2Pix and CycleGAN, which are employed to transform images across two domains. The decision to employ both models was motivated by the recognition that the problem may manifest in two different ways. Pix2Pix is a supervised image-to-image translation approach that requires paired data, whereby an input image from one domain is mapped to an output image in another domain, with labeled and aligned image pairs already available. On the other hand, CycleGAN is an unsupervised image-to-image translation method that does not require paired data. In this case, the model learns to map an image from one domain to another by identifying the fundamental relationship between the two domains. Given the crucial role that data plays in the training of GANs, a data pipeline has been developed to generate data suitable for both supervised and unsupervised image-to-image translation models. A data pipeline has been designed to preprocess clean PDFs and produce data of appropriate sizes for different domains. To ensure that the generated data accurately reflects real-world scenarios, various techniques were employed to simulate different types of image damage and imperfections. These techniques emulate the damage that typically occurs in real life scanned images, including scratches, stains, creases, and other deformities. The resulting data can then be utilized to train the GAN models to effectively repair the damaged images.

The methods were evaluated for their effectiveness using multiple quantitative measures, including PSNR, SSIM, and FID. Moreover, the influence of these methods on enhancing OCR performance was assessed using Character Error Rate (CER) and Word Error Rate (WER) metrics.

The research findings indicate that GANs can significantly enhance scanned images and restore them to their original state without losing details. However, there is a noticeable difference in performance between the two models, with the Pix2Pix model surpassing the CycleGAN model in all test cases. This is due to the Pix2Pix model's use of paired data, which allows for more precise mapping between input and output images, while CycleGAN uses unpaired data that complicates the task and increases instability during the training period.

Moreover, the research demonstrates that using different generator networks can yield different results. In this study, the use of ResNet outperformed U-Net. The study also highlights the impact of hyperparameters, such as loss functions and learning rates, on the model's efficiency. The study also highlights the impact of hyperparameters, such as loss functions and learning rates, on the model's efficiency. Our findings indicate that LSGAN provides the highest level of stability overall, while WGAN has the poorest performance. The results also show that each model behaves differently concerning learning rates. For instance, decreasing the learning rate resulted in a minor decrease in Pix2Pix's performance but contributed to enhancing CycleGAN's stability to some degree. However, further research is required to explore other models and ways to enhance the ability of unsupervised image translation models.

In summary, this research confirms that GANs are a powerful method for improving scanned images while retaining their content. The Pix2Pix model is more effective than the CycleGAN model, and using ResNet as the generator network is preferable to using Unet. Increasing the number of training images can also enhance the models' capabilities. This research provides valuable insights into GANs' potential for digital archiving and restoring scanned documents.

# 7.2. Future work

The results obtained in this thesis demonstrate the potential of Generative Adversarial Networks (GANs) in improving damaged scanned images and removing imperfections. However, there are several areas for future research that could further enhance the performance of the proposed method.

One potential avenue for future work is to investigate the use of attention mechanisms in the GANs. Attention mechanisms have been shown to improve the performance of deep learning models in a variety of tasks by allowing them to focus on specific areas of the input. By incorporating attention mechanisms into the GANs used in this thesis, it may be possible to further improve the quality of the generated images and reduce artifacts.

Another potential direction for future work is to explore the use of diffusion models for image restoration. Diffusion models are a class of generative models that have recently shown promise in image generation tasks. By learning a diffusion process that gradually removes noise from the input image, diffusion models can produce high-quality images with fewer artifacts compared to traditional GAN-based methods.

Finally, it could be beneficial to investigate the performance of the methods on different types of damaged scanned images. In this thesis, the focus was on removing noise and stains. However, there are other types of noise and imperfections that can occur in scanned images, such as blur, skew, or color distortion. By testing the methods on a wider range of damaged images, it would be possible to evaluate its effectiveness in different scenarios and identify potential limitations.

In conclusion, there are several avenues for future research that could further enhance the performance of the proposed methods. By incorporating attention mechanisms, exploring diffusion models, and testing the methods on different types of damaged images, it may be possible to develop a more robust and effective image restoration technique for damaged scanned images.

# References

- Why you should digitalize your business and how to succeed. Pagero, https://www.pagero.com/blog/digitalisation-why-digitalise-your-business/, Dec. 2019. [On-line; accessed 1.11.2022].
- [2] Mohamed Ali Souibgui and Yousri Kessentini. DE-GAN: A conditional generative adversarial network for document enhancement. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 44(3):1180–1191, Mar. 2022.
- [3] Aurelien Geron. Hands-On Machine Learning with Scikit-Learn, Keras, and TensorFlow: Concepts, Tools, and Techniques to Build Intelligent Systems. O'Reilly Media, Inc., 2nd edition, 2019.
- [4] I. Vasilev, D. Slater, G. Spacagna, and P. Roelants. Python Deep Learning Second Edition: Exploring Deep Learning Techniques and Neural Network Architectures with PyTorch, Keras, and TensorFlow, 2nd Edition. Packt Publishing, 2019.
- [5] Rich Stureborg. Artificial Neural Networks for Total Beginners. Medium, https://towardsdatascience.com/artificial-neural-networks-for-totalbeginners-d8cd07abaae4, May 2020. [Online; accessed 12.11.2022].
- [6] SAGAR SHARMA. Activation Functions in Neural Networks. Medium, https: //towardsdatascience.com/activation-functions-neural-networks-1cbd9f8d91d6, July 2021. [Online; accessed 12.11.2022].
- [7] Sanket Doshi. Various Optimization Algorithms For Training Neural Network. Medium, https://towardsdatascience.com/optimizers-for-training-neuralnetwork-59450d71caf6, August 2020. [Online; accessed 12.11.2022].
- [8] Vitaly Bushaev. Stochastic Gradient Descent with momentum Towards Data Science. Medium, https://towardsdatascience.com/stochastic-gradient-descentwith-momentum-a84097641a5d, June 2018. [Online; accessed 20. Feb. 2023].
- [9] Roan Gylberth. An Introduction to AdaGrad Konvergen.AI Medium. Medium, https: //medium.com/konvergen/an-introduction-to-adagrad-f130ae871827, June 2018. [Online; accessed 20. Feb. 2023].
- [10] Diederik P. Kingma and Jimmy Ba. Adam: A Method for Stochastic Optimization. International Conference on Learning Representations, 2014.
- [11] Vitaly Bushaev. Adam latest trends in deep learning optimization. Medium, https://towardsdatascience.com/adam-latest-trends-in-deep-learningoptimization-6be9a291375c, October 2018. [Online; accessed 20. Feb. 2023].

- [12] Zhe Ming Chng. Using Normalization Layers to Improve Deep Learning Models Machine-LearningMastery.com. MachineLearningMastery, https://machinelearningmastery. com/using-normalization-layers-to-improve-deep-learning-models, June 2022. [Online; accessed 16. Feb. 2023].
- [13] Sergey Ioffe and Christian Szegedy. Batch Normalization: Accelerating Deep Network Training by Reducing Internal Covariate Shift. In Francis Bach and David Blei, editors, Proceedings of the 32nd International Conference on Machine Learning, volume 37 of Proceedings of Machine Learning Research, pages 448–456, Lille, France, 07–09 Jul 2015.
- [14] Deep Learning normalization methods. Tung M Phung's Blog, https://tungmphung.com/ deep-learning-normalization-methods, January 2021. [Online; accessed 16. Feb. 2023].
- [15] Dmitry Ulyanov, Andrea Vedaldi, and Victor S. Lempitsky. Instance Normalization: The Missing Ingredient for Fast Stylization. *CoRR*, abs/1607.08022, 2016.
- [16] Lei Jimmy Ba, Jamie Ryan Kiros, and Geoffrey E. Hinton. Layer Normalization. *CoRR*, abs/1607.06450, 2016.
- [17] Yuxin Wu and Kaiming He. Group Normalization. In Vittorio Ferrari, Martial Hebert, Cristian Sminchisescu, and Yair Weiss, editors, Computer Vision - ECCV 2018 - 15th European Conference, Munich, Germany, September 8-14, 2018, Proceedings, Part XIII, volume 11217 of Lecture Notes in Computer Science, pages 3–19. Springer, 2018.
- [18] What are Convolutional Neural Networks? IBM, https://www.ibm.com/cloud/learn/ convolutional-neural-networks, January 2021. [Online; accessed 12.11.2022].
- [19] Karen Simonyan and Andrew Zisserman. Very Deep Convolutional Networks for Large-Scale Image Recognition. In *ICLR*, 2015.
- [20] Christian Ledig, Lucas Theis, Ferenc Huszar, Jose Caballero, Andrew Cunningham, Alejandro Acosta, Andrew Aitken, Alykhan Tejani, Johannes Totz, Zehan Wang, and Wenzhe Shi. Photo-Realistic Single Image Super-Resolution Using a Generative Adversarial Network, 2016.
- [21] Kaiming He, Xiangyu Zhang, Shaoqing Ren, and Jian Sun. Deep Residual Learning for Image Recognition. In CVPR, pages 770–778. IEEE Computer Society, 2016.
- [22] PyTorch ResNet: The Basics and a Quick Tutorial. https://www.run.ai/guides/deeplearning-for-computer-vision/pytorch-resnet. [Online; accessed 12.11.2022].
- [23] D. E. Rumelhart, G. E. Hinton, and R. J. Williams. *Learning Internal Representations by Error Propagation*, page 318–362. MIT Press, Cambridge, MA, USA, 1986.
- [24] Arden Dertat. Applied Deep Learning Part 3: Autoencoders. Medium, https://towardsdatascience.com/applied-deep-learning-part-3-autoencoders-1c083af4d798, October 2017. [Online; accessed 13.11.2022].

- [25] O. Ronneberger, P.Fischer, and T. Brox. U-net: Convolutional networks for biomedical image segmentation. In *Medical Image Computing and Computer-Assisted Intervention (MIC-CAI)*, volume 9351 of *LNCS*, pages 234–241. Springer, 2015. (available on arXiv:1505.04597 [cs.CV]).
- [26] Diederik P. Kingma and Max Welling. Auto-Encoding Variational Bayes. In 2nd International Conference on Learning Representations, ICLR 2014, Banff, AB, Canada, April 14-16, 2014, Conference Track Proceedings, 2014.
- [27] Ian J. Goodfellow, Jean Pouget-Abadie, Mehdi Mirza, Bing Xu, David Warde-Farley, Sherjil Ozair, Aaron C. Courville, and Yoshua Bengio. Generative Adversarial Networks. *CoRR*, abs/1406.2661, 2014.
- [28] Jason Brownlee. A Gentle Introduction to Generative Adversarial Network Loss Functions. MachineLearningMastery, https://machinelearningmastery.com/generativeadversarial-network-loss-functions, September 2019. [Online; accessed 20. Nov. 2022].
- [29] Xudong Mao, Qing Li, Haoran Xie, Raymond Y. K. Lau, Zhen Wang, and Stephen Paul Smolley. Least Squares Generative Adversarial Networks. 2017 IEEE International Conference on Computer Vision (ICCV), pages 2813–2821, 2016.
- [30] Jason Brownlee. How to Develop a Least Squares Generative Adversarial Network (LS-GAN) in Keras MachineLearningMastery.com. MachineLearningMastery, https://machinelearningmastery.com/least-squares-generative-adversarial-network, January 2021. [Online; accessed 18. Feb. 2023].
- [31] Martin Arjovsky, Soumith Chintala, and Léon Bottou. Wasserstein Generative Adversarial Networks. In Proceedings of the 34th International Conference on Machine Learning - Volume 70, ICML'17, page 214–223. JMLR.org, 2017.
- [32] Loss Functions | Machine Learning. Google Developers, https://developers.google. com/machine-learning/gan/loss. [Online; accessed 20. Nov. 2022].
- [33] Mehdi Mirza and Simon Osindero. Conditional Generative Adversarial Nets. *CoRR*, abs/1411.1784, 2014.
- [34] Phillip Isola, Jun-Yan Zhu, Tinghui Zhou, and Alexei A. Efros. Image-to-Image Translation with Conditional Adversarial Networks. In *2017 IEEE Conference on Computer Vision and Pattern Recognition (CVPR)*, pages 5967–5976, 2017.
- [35] Ting-Chun Wang, Ming-Yu Liu, Jun-Yan Zhu, Andrew Tao, Jan Kautz, and Bryan Catanzaro. High-resolution image synthesis and semantic manipulation with conditional gans. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, 2018.
- [36] Jun-Yan Zhu, Richard Zhang, Deepak Pathak, Trevor Darrell, Alexei A. Efros, Oliver Wang, and Eli Shechtman. Toward Multimodal Image-to-Image Translation. In Proceedings of the 31st International Conference on Neural Information Processing Systems, page 465–476. Curran Associates Inc., 2017.

- [37] Jun-Yan Zhu, Taesung Park, Phillip Isola, and Alexei A. Efros. Unpaired Image-to-Image Translation Using Cycle-Consistent Adversarial Networks. In *IEEE International Conference* on Computer Vision, ICCV 2017, Venice, Italy, October 22-29, 2017, pages 2242–2251. IEEE Computer Society, 2017.
- [38] Taeksoo Kim, Moonsu Cha, Hyunsoo Kim, Jung Kwon Lee, and Jiwon Kim. Learning to Discover Cross-Domain Relations with Generative Adversarial Networks. In Doina Precup and Yee Whye Teh, editors, *Proceedings of the 34th International Conference on Machine Learning*, volume 70 of *Proceedings of Machine Learning Research*, pages 1857–1865. PMLR, 06–11 Aug 2017.
- [39] Ming-Yu Liu, Thomas Breuel, and Jan Kautz. Unsupervised Image-to-Image Translation Networks. In I. Guyon, U. Von Luxburg, S. Bengio, H. Wallach, R. Fergus, S. Vishwanathan, and R. Garnett, editors, *Advances in Neural Information Processing Systems*, volume 30. Curran Associates, Inc., 2017.
- [40] Taesung Park, Alexei A. Efros, Richard Zhang, and Jun-Yan Zhu. Contrastive Learning for Conditional Image Synthesis. In ECCV, 2020.
- [41] Greg Walters and John Hany. *Hands-On Generative Adversarial Networks with PyTorch 1.x.* Packt, 2019.
- [42] Peak Signal-to-Noise Ratio as an Image Quality Metric. https://www.ni.com/dede/innovations/white-papers/11/peak-signal-to-noise-ratio-as-an-imagequality-metric.html, August 2011. [Online; accessed 4. Feb. 2022].
- [43] Fernando A. Fardo, Victor H. Conforto, Francisco C. de Oliveira, and Paulo S. Rodrigues. A Formal Evaluation of PSNR as Quality Measurement Parameter for Image Segmentation Algorithms, 2016.
- [44] Pranjal Datta. All about Structural Similarity Index (SSIM): Theory + Code in Py-Torch. Medium, https://medium.com/srm-mic/all-about-structural-similarityindex-ssim-theory-code-in-pytorch-6551b455541e, December 2021. [Online; accessed 4. Feb. 2023].
- [45] Martin Heusel, Hubert Ramsauer, Thomas Unterthiner, Bernhard Nessler, and Sepp Hochreiter. GANs Trained by a Two Time-Scale Update Rule Converge to a Local Nash Equilibrium. In Proceedings of the 31st International Conference on Neural Information Processing Systems, page 6629–6640. Curran Associates Inc., 2017.
- [46] Ayush Thakur. How to Evaluate GANs using Frechet Inception Distance (FID). https://wandb.ai/ayush-thakur/gan-evaluation/reports/How-to-Evaluate-GANsusing-Frechet-Inception-Distance-FID---Vmlldzo0MTAx0TI, February 2023. [Online; accessed 9. Feb. 2023].
- [47] Yaniv Benny, Tomer Galanti, Sagie Benaim, and Lior Wolf. Evaluation Metrics for Conditional Image Generation. Int. J. Comput. Vision, 129(5):1712–1731, May 2021.

[48] Kenneth Leung. Evaluate OCR Output Quality with Character Error Rate (CER) and Word Error Rate (WER). Medium, https://towardsdatascience.com/evaluating-ocr-outputquality-with-character-error-rate-cer-and-word-error-rate-wer-853175297510, January 2022. [Online; accessed 5. Mar. 2023].

# A. Test Cases Appendix

This appendix contains additional test cases that were not included in the main body of the document. These test cases were conducted to further validate the effectiveness of the methodologies used in this study. The additional test cases cover a variety of hyperparameters, providing a more comprehensive view of the performance of the methods under different hyperparameters.

#### Test Case 2:

In this test case, we aimed to revisit the performance comparison between the Pix2Pix and CycleGAN models to enhance the quality of the scanned images. However, in contrast to the first test case, we used an instance normalization layer instead of batch normalization in the generator architecture of both models. The focus of this test case is to assess the models' ability to eliminate noise, artifacts, and imperfections, such as coffee stains from the images. To ensure a fair comparison, we kept the same set of hyperparameters as in the previous test case, with the exception of the normalization layer, which yielded the following set of hyperparameters:

General Hyperparameter	
Adversarial loss function	Min-Max GAN loss
Normalization type	Instance normalization
Discriminator architecture	70 × 70 PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Learning rate	0.0002
Special Hyperparameter	
Pix2Pix L1 Lambda	100
CycleGAN cycle Lambda	10

#### **Quantitative Comparison:**

Building on our previous approach, we will continue to employ quantitative comparison as a means to evaluate the performance of the models in our second test case. Similar to the first test case, we used various established metrics (see Section 5.2). By using these metrics, we can objectively assess the model's ability to convert images from one domain to another and compare its performance with that of the ground truth. Additionally, we created line charts that displayed the performance of the model every five epochs during the 100-epoch training period for each metric. This visualization technique allows us to easily compare the models and identify those that perform better over time. By utilizing these methods, we can gain a comprehensive understanding of how well the model is operating and determine areas that may require improvement.

Figure 6.1 presents a straightforward representation of how the model progressed during the training phase, as measured using the PSNR metric. A higher value on the chart indicates that the model performed better.



Figure A.1.: PSNR line chart (higher is better) for test case 2

We can observe that the PSNR performance of the models in the second test case is highly comparable to that of the first test case, particularly in terms of Pix2Pix's performance stability. The Pix2Pix model with ResNet architecture continues to outshine the other models, achieving the highest PSNR score of 22.32 dB in epoch 95 and 22.2687 dB in epoch 55. However, the CycleGAN models with U-Net and ResNet architectures exhibited only slight performance stabilization in the earlier epochs before collapsing again. In particular, CycleGAN with ResNet architecture maintains acceptable results between 15 dB to 17.5 dB until epoch 35, then experiences a sharp decline, while CycleGAN with U-Net architecture gradually loses performance after epoch 25, with its highest value being around 16 dB at epoch 10.

As in the first test case, it is worth noting that increasing the number of epochs does not always translate into better performance in most cases. The choice of generator architecture also remains an important factor in determining the model performance. The Pix2Pix models continue to converge rapidly and maintain a relatively stable performance rate during training, while the CycleGAN models still exhibit unstable performance.

Overall, based on the PSNR results in this test case, it can be concluded that the Pix2Pix model with the ResNet architecture remains the most suitable option for the given task. Moving forward, we will examine the results of these models using the SSIM scale, which considers factors such as luminance, contrast, and structural information.

The subsequent Figure A.2 displays the progress of the models throughout the training process, as assessed using the SSIM metric. A greater value depicted on the chart indicates that the model achieves better performance.



Figure A.2.: SSIM line chart (higher is better) for test case 2

As we can see, Figure A.2 shows that the behavior of the models is generally similar to the previous PSNR metric A.1, which confirms the quality of the models in terms of luminance, contrast, and structural information. The highest-performing model was Pix2Pix with ResNet after 55 epochs, which achieved an SSIM value of 0.9528. The second-best performing model is Pix2Pix with ResNet after 95 epochs, which achieved an SSIM value of 0.9504. Notably, the best model in terms of SSIM was the second-best performer in the PSNR test, whereas the second-best performer in SSIM was the top-performing model in the PSNR test.

The CycleGAN models tend to be generally unstable, as in the PSNR test. However, there is a momentary period of stability with the resnet structure, which persists until approximately the 35th epoch. Nonetheless, this stability is not sustained and the model eventually collapses, as shown in the chart. On the other hand, the Unet model displays a certain level of stability up until roughly the 25th epoch, after which it gradually begins to decline.



Figure A.3.: FID line chart (lower is better) for test case 2

#### A. Test Cases Appendix

Overall, the Pix2Pix models exhibited a more stable performance than the CycleGAN models, even in the second test case. The collapse of the CycleGAN models is clearly evident in the Figure A.3, which further supports this conclusion.

After examining the three metrics, we can state that the Pix2Pix model with ResNet performed the best under the given hyperparameters, followed by Pix2Pix with U-Net. It's noteworthy that the CycleGAN models did not demonstrate stability even in this test scenario. To compare both test cases, we prepared a bar chart that displays the mean value for each metric and facilitates comparison with the first test case.

This analysis aims to determine whether the adoption of instance normalization in place of batch normalization led to an enhancement in the models' quality.



As we can see in the Figure A.4, it can be observed that using Instance normalization instead

Figure A.4.: Test case 2

of batch normalization results in a slight performance improvement for the ResNet architecture in both models. However, when it comes to the U-Net architecture, the situation is different, and there is a decline in performance, except for the FID metric, which shows an improvement in the results.

The following table shows the results of the comparison more precisely, considering the architecture of the generator and the type of normalization:

Model	Mean PSNR ↑	Mean SSIM ↑	Mean FID $\downarrow$
Pix2Pix with ResNet and Instance norm	21.8609	0.9464	4.0700
Pix2Pix with ResNet and Batch norm	21.4998	0.9423	4.0701
Pix2Pix with U-Net and Instance norm	18.9144	0.9124	6.6881
Pix2Pix with U-Net and Batch norm	19.1614	0.9151	6.6716
CycleGAN with ResNet and Instance norm	12.7282	0.6645	143.6078
CycleGAN with ResNet and Batch norm	12.3379	0.6389	162.7294
CycleGAN with U-Net and Instance norm	12.0839	0.6454	84.9879
CycleGAN with U-Net and Batch norm	12.3156	0.6444	143.5283

Table A.1.: Test case 2 results table

#### **Visual Inspection:**

Following the analysis of the quantitative measurements, we will proceed to visually inspected various samples to validate the effectiveness of the models. Specifically, we examine only the good performing epochs, which include Pix2Pix with ResNet at epoch 95, Pix2Pix with U-Net at epoch 95, CycleGAN with ResNet at epoch 20, and CycleGAN with U-Net at epoch 10. The following tests show the outcomes of each example for a given input:



#### Input



#### Ground truth

# Setetur Iod tempor uyam erat, set diam imot tempor invidunt ut aliquyan erat, set diam usam et juss set diam valuts set door sit or sit amet, consetetur satisscing elitr, set diam noumy elimot tempor invidunt ui labore et dolore magna aliquyah erat, set diam valutus. Stet cita kasa gubergen, no sea takimata sarctus est Lorem ipsum dolor sit amet. Lorem ipsum done sit amet, consetetur satisscing elitr, sed diam noumy elimot tempor invidunt ui labore et dolore magna aliquyah erat, sed diam valutus. Stet cita kasa gubergen, no sea takimata sarctus est Lorem ipsum dolor sit amet. Lorem est avent dolor sit and bore et dolore emaga aliquyam erat, sed diam valutus est Lorem ipsum dolor sit mot and tempor te a erebur. Stet cita kasa gubergen, no sea takimata sanctus est Lorem ipsum dolor sit amet. setetur no sea takimata sanctus est Lorem ipsum dolor sit amet.

#### Pix2Pix with ResNet



Pix2Pix with U-Net

seleiur

setteiur in allquyan erat, sed diam is allquyan erat, sed diam is allquyan erat, sed diam is accusant ti usto duo is allquyan erat, sed diam is accusant ti usto duo is allquyan erat, sed diam is accusant ti usto duo is all core all of the setter is allquyan erat, sed diam nonumy elimot tempor inviduit ti altore et diolere magna allquyan erat, sed diam valuptus. At vero eos et accusam ti usto duo dolores el ar rebum Stet cita kased gubergeren no soo takimata sonctus cet Lorem ipsum dolor ti amet. Lorem ipsum dolor sit annet, correletiun sadipsoing elitr, sed diam nerumy elimot tempo inviduit ut labore et dolore magna allquer erat uto dolores et ea rebum. Site cita kased gubergeren no sea takimata sanctus est Lorem ipsum dolor sit amet.

#### CycleGAN with ResNet



#### CycleGAN with U-Net





ヌモタ客討到マ作改冠モ城社ご表17会ドか首国3択ミセ 島人ラベさぜ手故まほをル ニイ思ヤコ腰能サ状入> 那級社>メ田更ぞげうび速日全ん い伊立クドきそ寺高とほれル代庁タヒカセ能社安んへ探醒 系競工門は定義玉暖渡っ。図見コペマ在時フやざき掲紹の ド開設フジィ差69調エスメナ支年など秋6南てものぼ断用文 「予思ヤコ腰能サ状入シャ、ハーメ田更ぞげうび連日全A お伊立少杉きを寺高必ほれ。こし守をた力で能社交ん、実務 来聞なぜた広葉も聴選うの肌スは、今時ウンやさ春都留 や開表ラ・ 165歳 エズムナ支年など秋6南てものぼ断用文 3回にル末場新常クオヤク花煮さやし所新ごん相様テ煙性ナ 調出して末場有容力、など意味のしい、したやち、つき が最小な子野族んごな人ニアスタ生電みぶ実現力支払して 増充活くしてよるスラカの原道例せた。 特えているようスラカの原道例せた。 特えているような人ニアスタ生電みが長いで あったい、 コンタインによった。 コンタインによった。 コンタインによった。 コンタインによった。 ヲタイ差67源 7 ややび座月せ音後ムスリ連合ソナオ た夏の遊が相振りつ国記4社ごろ などの違かが構成りつ国記4社ごろ 地方健長ジムン第応サークや道鉄は+場析密クオ ナぼや科置 ナほや科置

67浙

Note that the red lines are not originally present in the images and have been added to highlight specific details.

64新ニ書情かうずでは分路きも

The aforementioned examples serve as a representation of the various samples that were tested. For further samples, we kindly direct you to the appendix B where additional samples are available for review.

予思ヤコ腰能サ状入シ那級社シメ田更ぞけうび速日全ん )伊立クドきそ寺高どほれル代庁タヒカセ能社安んへ探顕 、競スは社にに続ま悪激ラ。9凱ヌロヘマ毎時フやざき掲昭几 開表フメイ差67調エステナ支年など秋6南てものぼ断用文

順にル本場新窓クオヤウ花鹿さやじ所転げん相様子煙生サ 間世士ほや料置ナーるむ発得の2個の先品で、世やらずつ が最外気音解んを発入ニアスタ生電るみ家理力を起じて 着だはつ、花ネスラカ阿京遣例せた材子中の、セトジン型 最ミク入量別、びト。

ややび座月せ育なムヌリ連合ソナポ 大夏の運が視聴すう(国記社ころ) ケクリ無書いぼ可能と新久ニ間10 花宇像表史ムン帯広サークを連載計場新密クオ

- 遺信不らずでけ分降きせ

ヲメイ差67源

ナぼや科置

#### OCR recognition:

This section aims to evaluate the effectiveness of a model in improving the accuracy of Tesseract OCR. The model's performance will be assessed on different test samples using two metrics, namely character error rate (CER) and word error rate (WER). Tesseract OCR often encounters challenges when processing low-quality scanned images, thus highlighting the need to determine the models potential to improve Tesseract OCR's accuracy in text extraction. The goal is to demonstrate the models performance on various test samples to determine its ability to enhance the accuracy of Tesseract OCR using character error rate (CER) and word error rate (WER), which were introduced in Section 5.3.

Sample: invidunt ui Iabore et dolore magha aliquyam erat,

• Source image:

invidunt ut labore et dolore magna aliquyam erat,

Extracted text: invf(:;!\_;;4r»3i\_'ui Sabore ei doiore magma Character Error Rate: 0.6939 Word Error Rate: 1.0

• Pix2Pix with ResNet:

invidunt ut labore et dolore magna allquyam erat,

Extracted text: invidunt ui Iabore et dolore magné aliquyam eiat, Character Error Rate: 0.0816 Word Error Rate: 0.5

#### • Pix2Pix with U-Net:

invidunt ut labore et dolore magna eliquyam erat,

Extracted text: invrdunt ui labore el dolore magni-a aliquyam ekai, Character Error Rate: 0.1429 Word Error Rate: 0.6250

#### • CycleGAN with ResNet:

### invidunt ut labore et dolore magna, aliquyam eral,

-

Extracted text: invidunt ui labore el dolore magnépllquyam eial, Character Error Rate: 0.1633 Word Error Rate: 0.6250

#### • CycleGAN with U-Net:

invidunt ut labore et dolore magna alluuvam e

Extracted text: Invuuunt ul labore el dolore magna -muuvaw -» Character Error Rate: 0.3265 Word Error Rate: 0.6250 Similar to the example provided earlier, we computed the mean of Character Error Rate and Word Error Rate for 100 different sentences, and obtained the subsequent outcomes for each model:

Model	Mean Character Error Rate $\downarrow$	Mean Word Error Rate $\downarrow$
Source	0.6435	0.8125
Pix2Pix with ResNet	0.0517	0.2572
Pix2Pix with U-Net	0.0900	0.3883
CycleGAN with ResNet	0.2312	0.5473
CycleGAN with U-Net	0.2501	0.5906

Table A.2.: Test case 2 OCR recognition results table

Table A.2 presents the evaluation of five different cases on 100 sentences, using CER and WER. The first row of the table represents the performance of the source, which is the original image with no enhancements. The remaining four rows correspond to the performance after using the four models: Pix2Pix with ResNet, Pix2Pix with U-Net, CycleGAN with ResNet, and CycleGAN with U-Net.

Upon examination of the table, it is evident that the Pix2Pix model with ResNet achieved the highest performance in improving OCR accuracy, with a mean CER of 0.0517 and a mean WER of 0.2572. These results suggest that the model was successful in replicating the original text with very few errors. The Pix2Pix model with U-Net also demonstrated good performance, with a CER of 0.0900 and a WER of 0.3883, although it was not as effective as the Pix2Pix model with ResNet.

In contrast, the CycleGAN models had lower accuracy compared to the Pix2Pix models, with CycleGAN with ResNet having a CER of 0.2312 and a WER of 0.5473, and CycleGAN with U-Net producing the highest error rates with a CER of 0.2501 and a WER of 0.5906. These results suggest that the CycleGAN models were less effective in enhancing the scanned images for OCR, resulting in less accurate text extraction.

In conclusion, based on the findings presented in the table, it can be concluded that the Pix2Pix models with ResNet and U-Net were more effective in improving OCR accuracy than the CycleGAN models. These results suggest that the Pix2Pix models are better suited for this particular task and may be more appropriate for enhancing OCR accuracy in similar situations.
#### Test Case 3:

Our objective in this test case was to compare the performance of the Pix2Pix and CycleGAN models in enhancing the quality of the scanned images. Unlike previous test cases, we used three different adversarial loss functions for each model, instead of using only the original loss function, to gain a deeper understanding of their effects. The three loss functions included in this test case are as follows:

- Original GAN Loss (Min-Max GAN Loss)
- LSGAN Loss (Least Squares GAN Loss)
- WGAN Loss (Wasserstein GAN Loss)

For more information on loss functions, please refer to Section 2.4.2.

The main focus of this test case was to evaluate the ability of the models to remove noise, artifacts, and imperfections, such as coffee stains from the images, and how the different loss functions affected their performance. We ensured a fair comparison by maintaining the same set of hyperparameters used in the previous test cases, except for the loss functions:

General Hyperparameter	
Normalization type	Instance normalization
Discriminator architecture	70 × 70 PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Learning rate	0.0002
Special Hyperparameter	
Pix2Pix L1 Lambda	100
CycleGAN cycle Lambda	10

#### **Quantitative Comparison:**

We will begin the evaluation process for the third test case using the same quantitative comparison approach. Similar to the previous test cases, we utilized various established metrics (as outlined in Section 5.2) to objectively evaluate the models' performance in transforming images from one domain to another and compared it against the ground truth. In addition, we created line charts that showcase the model's performance at intervals of five epochs over a 100-epoch training period for each metric. This visualization technique enables us to easily compare the models and identify those that perform better over time. By utilizing these techniques, we can obtain a good understanding of the model's effectiveness.

The line charts in this test case show provide a representation of how the models progressed during the training phase for each of the three loss functions as measured by the metrics. Each chart is divided into four subcharts that describe a particular model and provide information for each of the three loss functions.



Figure A.5.: PSNR line charts (higher is better) for test case 3

We begin by presenting the results of the PSNR metric in Figure A.5. Upon initial observation, it is evident that both the Min-Max and LSGAN loss functions exhibit significantly superior performance in all models compared to the WGAN loss function. The latter demonstrated poor results and instability across boards. In contrast, the LSGAN loss function yields better overall outcomes and stability than other functions. Additionally, the Min-Max function appears to be superior to LSGAN during the early epochs of CycleGAN models; however, it fails to maintain stability, unlike LSGAN.

The Pix2Pix model with ResNet architecture maintains its superiority over the other models, even when using different loss functions. Specifically, it achieves the highest PSNR of 22.64 dB at epoch 80 using the LSGAN loss function, and 22.32 dB at epoch 95 using the Min-Max loss function. Similarly, the Pix2Pix version with U-Net architecture demonstrates the best results when using the LSGAN loss function, achieving a top PSNR of 20.12 dB at epoch 75. For the first time, the CycleGAN model with ResNet architecture exhibited performance stability when using the LSGAN loss function. In contrast, the CycleGAN model with the U-Net architecture shows less volatility when using the LSGAN loss function, but its performance gradually declines after epoch 40, with its highest PSNR value peaking at approximately 14.55 dB. Notably, CycleGAN with ResNet architecture maintains acceptable results ranging from 15 dB to 17.22 dB across all epochs.

Based on the aforementioned results, it can be deduced that the LSGAN loss function provides the most favorable and stable outcomes overall, whereas the WGAN loss function is entirely unsuitable for these models, resulting in poor outcomes for all tested models. Nevertheless, relying solely on one metric is inadequate, necessitating an assessment of the remaining two metrics. To further support this conclusion, we examine the outcomes of the SSIM metric, which provides more precise information concerning the models' performance regarding luminance, contrast, and structural information.



Figure A.6.: SSIM line charts (higher is better) for test case 3

As shown in Figure A.6, the results obtained from the SSIM metric are largely similar to those obtained from the PSNR, which supports our conclusion that the LSGAN loss function generally yields the best performance. However, that the CycleGAN model with U-Net architecture shows a more pronounced decline in SSIM values compared to PSNR when LSGAN is used.

In terms of specific results, the Pix2Pix model with ResNet architecture consistently demonstrates the best performance, with SSIM values ranging from 0.94 to 0.95 for both LSGAN and Min-Max loss functions. The Pix2Pix model with the U-Net architecture follows, with results ranging from 0.9 to 0.92. By contrast, the CycleGAN model exhibited weaker overall results. Nevertheless, it is noteworthy that the use of the LSGAN loss function and ResNet architecture provides some degree of stability to CycleGAN, with results ranging from 0.65 to 0.83.

Following the examination of the outcomes for the two previous metrics, we now turn our attention to the third measure, which yields more robust results because it considers the distribution of the generated images as a whole, as compared to the distribution of the real images, providing a more comprehensive evaluation of the models' overall performance.

As we see in Figure A.7, the FID results reveal a similar perfomance patterns across all models, supporting our previous conclusion that the LSGAN loss function exhibits the highest level of stability. Conversely, the WGAN loss function yielded poor results in all instances and was the least stable. Similar to the previous tests, the Pix2Pix model with ResNet architecture demonstrated superior performance compared to all other models, followed by Pix2Pix with U-



Figure A.7.: FID line charts (lower is better) for test case 3

Net architecture, CycleGAN with ResNet architecture, and CycleGAN with U-Net architecture. These findings are further elucidated in Figure A.8, which displays the mean of all epochs for each model in this test case.



Figure A.8.: Test case 3

As depicted in the Figure A.8, it is evident that the utilization of various loss functions has a notable impact on the performance of the models. The results of this test case indicate that the use of the LSGAN loss function leads to the most favorable outcomes for all models, with the exception of CycleGAN with U-Net architecture, which consistently performs poorly across all configurations. The Min-Max loss function, on the other hand, produces results that are generally comparable to those of LSGAN, while the WGAN loss function consistently yields suboptimal results across all models.

Model	Loss Function	Mean PSNR ↑	Mean SSIM ↑	Mean FID $\downarrow$
Pix2Pix with ResNet	Min-Max Loss	21.8609	0.9464	4.07
Pix2Pix with ResNet	LSGAN Loss	22.1263	0.9487	3.3654
Pix2Pix with ResNet	WGAN Loss	14.6482	0.7278	73.6215
Pix2Pix with U-Net	Min-Max Loss	18.9144	0.9124	6.6881
Pix2Pix with U-Net	LSGAN Loss	19.5325	0.9207	5.7343
Pix2Pix with U-Net	WGAN Loss	14.6457	0.6984	97.8312
CycleGAN with ResNet	Min-Max Loss	12.7283	0.6645	143.6078
CycleGAN with ResNet	LSGAN Loss	15.7593	0.7211	35.0972
CycleGAN with ResNet	WGAN Loss	9.2816	0.4204	231.8663
CycleGAN with U-Net	Min-Max Loss	12.0839	0.6454	84.9879
CycleGAN with U-Net	LSGAN Loss	11.7820	0.4952	106.3209
CycleGAN with U-Net	WGAN Loss	8.8682	0.3377	212.3745

The following table shows the results of the comparison more precisely, taking into account the architecture of the generator and the type of loss function:

Table A.3.: Test case 3 results table

#### **Visual Inspection:**

Following the analysis of the quantitative measurements, we will proceed to visually inspect various samples to validate the effectiveness of the models. Specifically, we will examine just the good performing epochs, which include Pix2Pix with ResNet at epoch 95, Pix2Pix with U-Net at epoch 95, CycleGAN with ResNet at epoch 20, and CycleGAN with U-Net at epoch 10. The following tests showcase the outcomes of each example for a given input:



• Pix2Pix with ResNet:

#### MinMax Loss

#### LSGAN Loss

Aria

### Arial Lorem issum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est. Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipschg elitr, sed Gläm-nonumy eirmod tempor invidunt ti tabore et dolore magna eliquyam erat, sed diam voluptua. At vero eos et accusamet justo duo dolors et ca rebum. Stet clita kasd ubergren no sea takimata sanctus est Lorem ge elitr, sed sadipscing elito ed diam nonum ti labore et invidunt ut labore et dolore magna ti labore et sed diam voluptua. At vero eos si voluptua. V duo dolores et ea rebum. Stet clita kasd et can to no sea takimata sanctus est. Iore es et can to no sea takimata sanctus est. Iore es et can to sed diam voluptua. At vero eos si voluptua. V no sea takimata sanctus est Lorres et ca re met.

## Arial Lorem insum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero es et accusam et justo duo dolores et ea rebum. Stet cital kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipschg elitr, sed diam enumy eimod tempor invidunt ut labore el dolore magna eliuperan no sea takimata sanctus est Lorem ge elitr, sed sadipscing elitr, sed diam nonum invidunt ut labore et dolore magna eliuperator no sea takimata sanctus est Lorem ge elitr, sed sadipscing elitr, sed diam nonum invidunt ut labore et dolore et o sed diam voluptua. A vero es es sed diam voluptua. Stet dia Nasch ut labore et uo dolores et ea rebum. Stet dia sed diam voluptua. Stet dia no sea takimata sanctus est Lorres et ca re amet.

#### WGAN Loss

orem ipsum dolor sit amet consetetur sadipscing				
labore et dolore magna alíguvam erat sed diam				
voluptua. At vero eos et accusam et justo duo				
dolores et ea rebum. Ctet clita kasd gubergren, no				
sea takimata sanctus est Lorem ipsum dolor sit				
amet. Lorem ipsum dolor sit amet, consetetur				
sadipscing elitrased diam nonumy eirmod tempor				
invidunt ut labore et dolore magna aliquyam erat,				
sed diam voluptua. At vero eos et accusam et justo				
duo dolores el ca rebum Stet clita kasa auberaren				
no sea takimata sanctus est Lorn delitra sed				
amet. Lorem psum dolor sit ame				
sadipscing eliti, sed diam nonum 11 Jabore et.				
Invidunt ut labore et dolore magr				
due delense et es rebure. Stat al				
duo dolores el ea reburn. Stel ci				
amel				

#### • Pix2Pix with U-Net:

# <section-header>Number DossNumber DosNumber Dos



#### • CycleGAN with U-Net:



Note that the red lines are not originally present in the images and have been added to highlight specific details.

The aforementioned examples serve as a representation of the various samples that were tested. For further samples, we kindly direct you to the appendix B where additional samples are available for review.

#### OCR recognition:

This section aims to evaluate the effectiveness of a model in improving the accuracy of Tesseract OCR, an open-source OCR engine, which is widely used for text recognition. The model's performance will be assessed on different test samples, as Tesseract OCR can encounter difficulties with low-quality scanned images. Therefore, the goal is to showcase the performance of the models on various test samples to determine its ability to enhance the accuracy of Tesseract OCR in text extraction using character error rate (CER) and word error rate (WER), which were introduced in Section 5.3.

Sample: voluptua. At vero eos et accusam et justo duo

• Source image:

voluptua. At vero eos et acousam et justo avo

Extracted text: Strange and incorrect symbols were extracted Character Error Rate: 1.0667 Word Error Rate: 1.0

#### • Pix2Pix with ResNet and Min-Max Loss:

voluptua. At vero eos et acousam et justo duo

Extracted text: voluptua. At vero eds et acduisam et justo duo Character Error Rate: 0.0667 Word Error Rate: 0.2222

#### • Pix2Pix with ResNet and LSGAN Loss:

#### voluptua. At vero eos et acousam et justo duo

Extracted text: voluptua. At vero eds et acduisam et justo duo Character Error Rate: 0.0667 Word Error Rate: 0.2222

#### • Pix2Pix with ResNet and WGAN Loss:

voluptua. At vero eco et accusam et jutto a...o

Extracted text: Strange and incorrect symbols were extracted Character Error Rate: 1.8889 Word Error Rate: 1.8889

#### • Pix2Pix with U-Net and Min-Max Loss:

voluptua. At vero ecs et accusam et justo duo

Extracted text: voluptua. At' vero ecs et accuaam et justo auo Character Error Rate: 0.0889 Word Error Rate: 0.4444

#### • Pix2Pix with U-Net and LSGAN Loss:

#### voluptua. At vero eos et accusam et justo duo

Extracted text: voluptua. At vero eds et accusam et justo duo Character Error Rate: 0.0222 Word Error Rate: 0.1111

#### • Pix2Pix with U-Net and WGAN Loss:

voluptua: At vero eos et accusam et	jut d'io
-------------------------------------	----------

Extracted text: voluptua'. At vero ass ét acbuisam étjuf '"0 Character Error Rate: 0.3111 Word Error Rate: 0.7778

#### • CycleGAN with ResNet and Min-Max Loss:

voluptua; At vero eus et accusam et ju	dNo	1
--	-----	---

Extracted text: voluptua./>41' vero eos et accusém éf jué' duo Character Error Rate: 0.2444 Word Error Rate: 0.5556

#### • CycleGAN with ResNet and LSGAN Loss:

voluptua: At vero ecs et acousam et justo no

Extracted text: voiuptua'. At' var ecé .:t'accui5 Character Error Rate: 0.5778 Word Error Rate: 1.0

#### • CycleGAN with ResNet and WGAN Loss:

Extracted text: Character Error Rate: 1.0 Word Error Rate: 1.0

#### • CycleGAN with U-Net and Min-Max Loss:

Extracted text: voluptua: Ar var' ,5 erabéuaam em :0 Character Error Rate: 0.4889 Word Error Rate: 1.0

#### • CycleGAN with U-Net and LSGAN Loss:

voluptua. At verdieus et accusam et juite no

Extracted text: Strange and incorrect symbols were extracted Character Error Rate: 0.9333 Word Error Rate: 1.0

#### • CycleGAN with U-Net and WGAN Loss:

Extracted text: Character Error Rate: 1.0 Word Error Rate: 1.0

Similar to the example provided earlier, we computed the mean of Character Error Rate and Word Error Rate for 100 different sentences, and obtained the subsequent outcomes for each model:

Model	Loss Function	Mean CER↓	Mean WER $\downarrow$
Source	-	0.6435	0.8125
Pix2Pix with ResNet	Min-Max Loss	0.0476	0.2374
Pix2Pix with ResNet	LSGAN Loss	0.0526	0.2389
Pix2Pix with ResNet	WGAN Loss	0.1969	0.5118
Pix2Pix with U-Net	Min-Max Loss	0.0900	0.3883
Pix2Pix with U-Net	LSGAN Loss	0.0622	0.3023
Pix2Pix with U-Net	WGAN Loss	0.1297	0.4508
CycleGAN with ResNet	Min-Max Loss	0.1198	0.4296
CycleGAN with ResNet	LSGAN Loss	0.2379	0.5773
CycleGAN with ResNet	WGAN Loss	0.9988	1.005
CycleGAN with U-Net	Min-Max Loss	0.2312	0.5473
CycleGAN with U-Net	LSGAN Loss	0.4173	0.6587
CycleGAN with U-Net	WGAN Loss	0.9943	1.0

Table A.4.: Test case 3 OCR recognition results table

Table A.4 presents the evaluation of 13 different cases on 100 sentences, using two metrics, namely mean CER and mean WER. The models compared include the source image with no enhancements, as well as several variations of the Pix2Pix and CycleGAN models with different loss functions and network architectures.

The results show that the source image has the highest mean CER and WER, indicating that the models are improving the accuracy of the generated text. Among the Pix2Pix models, the best performance was achieved by the model using ResNet and Min-Max Loss, with a mean CER of 0.0476 and a mean WER of 0.2374. The worst performance among the Pix2Pix models

was observed for the model using ResNet and WGAN Loss, with a mean CER of 0.1969 and a mean WER of 0.5118.

For the CycleGAN models, the best performance was achieved by the model using ResNet and Min-Max Loss, with a mean CER of 0.1198 and a mean WER of 0.4296. The worst performance among the CycleGAN models was observed for the model using ResNet and WGAN Loss, with a mean CER of 0.9988 and a mean WER of 1.005.

Overall, the results suggest that the choice of loss function and network architecture can have a significant impact on the performance of image-to-image translation models. The Pix2Pix model with ResNet and Min-Max Loss appears to be the most effective, followed closely by the Pix2Pix model with ResNet and LSGAN Loss, and the Pix2Pix model with U-Net and LSGAN Loss. However, further experiments are needed to confirm these results and to explore the performance of these models on different datasets.

#### Test Case 4:

The primary objective of this test case was to conduct a comparative analysis of the efficacy of the Pix2Pix and CycleGAN models in enhancing the quality of scanned images by integrating a VGG loss function into each model's loss function. Specifically, the aim was to investigate whether the inclusion of a VGG loss function had a significant impact on model performance. For more information about the VGG loss function, please refer to section 2.2.1.

The main focus of this test case was to evaluate the models' ability to eliminate noise and imperfections such as coffee stains from images, and the impact of incorporating perceptual loss on their performance. To ensure a fair and unbiased comparison, we kept a consistent set of hyperparameters throughout the experiment which are as follows:

General Hyperparameter	
Adversarial loss function	LSGAN loss
Normalization type	Instance normalization
Discriminator architecture	$70 \times 70$ PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Learning rate	0.0002
Special Hyperparameter	
Pix2Pix L1 Lambda	100
CycleGAN cycle Lambda	10

In this particular test case, the LSGAN adversarial loss function was employed due to its superior performance observed in the previous test case. Additionally, it also confers some degree of robustness to the CycleGAN model.

#### **Quantitative Comparison:**

Similar to previous test cases, the evaluation process for the fourth test case commenced with the adoption of a quantitative comparison approach. To objectively evaluate the models' performance, we utilized the three metrics described in the 5.2 section. Furthermore, we will generate line charts illustrating the model's performance at 5-epochs intervals over a 100-epoch training period for each metric, similar to the previous test cases.

The line charts presented in this test case represent the models' progression throughout the training phase, measuring their performance using the metrics, regardless of whether the VGG loss function was used. Each chart is split into four subcharts, each describing a specific model.

Figure A.9 shows that the incorporation of the VGG loss function resulted in a slight improvement in the performance of the Pix2Pix models with both generator architectures, as determined by the PSNR metric, while maintaining a relatively consistent performance throughout all epochs. Conversely, the utilization of the VGG loss function in CycleGAN models with both generator structures resulted in significantly inferior outcomes.



Figure A.9.: PSNR line charts (higher is better) for test case 4

The Pix2Pix model with ResNet architecture maintains its superiority over other models. As we have seen in all previous tests, even when using the VGG loss function, the model's performance continued to improve, while remaining superior to other models. Notably, the model achieved the highest PSNR of 23.08 dB in epoch 90 when using the VGG loss function, which is the best result obtained so far. Similarly, the Pix2Pix with the U-Net architecture exhibits better results with the VGG loss function, achieving a peak PSNR of 20.59 dB in epoch 85. However, the CycleGAN models with both architectures experienced a decline in performance when using the VGG loss function, likely because of the construction of the CycleGAN Loss function.



Figure A.10.: SSIM line charts (higher is better) for test case 4

A comparable pattern to the PSNR can also be observed in the SSIM metric for all the models in Figure A.10. This indicates that the Pix2Pix models for both generator architectures exhibit an enhanced ability to preserve structural information when the VGG loss function is incorporated, albeit only slightly. This is exemplified by the superior performance of the Pix2Pix model with the ResNet architecture, which achieves the highest score so far of 0.959 in epoch 90. Similarly, the Pix2Pix model with the U-Net architecture achieved the best result of 0.935 at epoch 85. However, the addition of the VGG loss function resulted in a significant decline in the performance of the CycleGAN models.



Figure A.11.: FID line charts (higher is better) for test case 4

As we can see in Figure A.11, the FID metric also shows similar behavior to PSNR and SSIM, which reinforces the observations made by these metrics. The Pix2Pix models with ResNet and U-Net architectures consistently outperform the CycleGAN models in terms of FID scores as the other metrics, which indicates that they are better at generating images that are closer to the ground truth.

Moreover, the addition of the VGG loss function has a positive impact on the performance of Pix2Pix models, as they achieve even lower FID scores in some epochs with VGG loss than without it. However, the same cannot be said for CycleGAN models, as their FID scores increase significantly with the inclusion of the VGG loss function.

To gain a better understanding of the overall performance of each model, we examined the means of their scores across all epochs for each metric. This is presented in Figure A.12.

As observed in Figure A.12, incorporating the VGG loss function results in a marginal enhancement in the performance of the Pix2Pix models across all metrics. By contrast, the addition of

#### A. Test Cases Appendix



Figure A.12.: Test case 4

the VGG loss function diminished the quality of the CycleGAN models. We can see the results more accurately in table A.5

Model	With VGG	Mean PSNR ↑	Mean SSIM ↑	Mean FID $\downarrow$
Pix2Pix with ResNet	True	22.4333	0.9517	3.4039
Pix2Pix with ResNet	False	22.1263	0.9487	3.3654
Pix2Pix with U-Net	True	19.7779	0.9250	5.5988
Pix2Pix with U-Net	False	19.5325	0.9207	5.7343
CycleGAN with ResNet	True	11.8914	0.3541	113.5098
CycleGAN with ResNet	False	15.7593	0.7211	35.0972
CycleGAN with U-Net	True	7.6837	0.2071	194.2524
CycleGAN with U-Net	False	11.7820	0.4952	106.3209

Table A.5.: Test case 4 results table

#### Visual Inspection:

After analyzing the quantitative measurements, we will visually inspect sample to verify the effectiveness of the models. We will focus on the good-performing epochs, namely Pix2Pix with ResNet at epoch 90, Pix2Pix with U-Net at epoch 85, CycleGAN with ResNet at epoch 15, and CycleGAN with U-Net at epoch 30. These tests will demonstrate the outcomes of each example for a given input:



• Pix2Pix with ResNet:



With VGG Loss

Arial Lorem ipsum dolor sit amet, constitut sadipkcing elit, sed diam nonumy eirmor timbor invidum ut labore et dolore magna all dyam fait. Sed diam voluptua, At vero eos et accusam et lusto dud dolores et ae rebur. Stet clia kasd buergen, no sea takimata sanctus est Lorem ipsum dolor sit amet. Loren posum dolor sit amet, consetur sadipscips elit, sed diam nonumy eirmost tempor inviduur ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et lusto tetur sadippus est Lorem ipsum dolor sit por inviduur un site clia kasd gubergren, labore et dolore magna aliquyam erat, set diam voluptua. At vero eos et accusam et lusto tetur sadippus est Lorem ipsum dolor sit por inviduur un nonumy eirmos tempor liore mogna aliquyam erat, set diak sed gubergren, m. Stet clia kasd gubergren, set Lorem ipsum dolor sit

• Pix2Pix with U-Net:



• CycleGAN with ResNet:



• CycleGAN with U-Net:







Note that the red lines are not originally present in the images and have been added to highlight specific details.

The aforementioned example serve as a representation of one sample that was tested. For further samples, we kindly direct you to the appendix B where additional samples are available for review.

#### OCR recognition:

This section aims to evaluate the effectiveness of a model in improving the accuracy of Tesseract OCR, an open-source OCR engine, which is widely used for text recognition. The model's performance will be assessed on different test samples, as Tesseract OCR can encounter difficulties with low-quality scanned images. Therefore, the goal is to showcase the performance of the models on various test samples to determine its ability to enhance the accuracy of Tesseract OCR in text extraction using character error rate (CER) and word error rate (WER), which were introduced in Section 5.3.

Sample: amet. Lorem ipsum dolor sit amet, consetetur

#### Source image:

amet. Lorem ipsum dolor sit amet, consetetur

Extracted text: orern ipsum dblfiametfl consetetur Character Error Rate: 0.4091 Word Error Rate: 0.7143

#### • Pix2Pix with ResNet (without VGG Loss):

#### met. Lorem ipsum dolor sit amet, consetetur

Extracted text: hmet. Lorem ipsum dolor sit amef, consetetur Character Error Rate: 0.0455 Word Error Rate: 0.2857

#### • Pix2Pix with ResNet (with VGG Loss):

#### amet. Lorem ipsum dolor sit amet, consetetur

Extracted text: amet. Lorem ipsum dolor sit amef, consetetur Character Error Rate: 0.0227 Word Error Rate: 0.1429

#### • Pix2Pix with U-Net (without VGG Loss):

#### amet. Lorem ipsum dolor sit amet, consetetur

Extracted text: gimet. Lorem ipsum dolor sit amef, consetetur Character Error Rate: 0.0682 Word Error Rate: 0.2857

#### • Pix2Pix with U-Net (with VGG Loss):

#### lamet. Lorem ipsum dolor sit amet, consetetur

Extracted text: ;amet. Lorem ipsum dolor sit amef, consetetur Character Error Rate: 0.0455 Word Error Rate: 0.2857

#### • CycleGAN with ResNet (without VGG Loss):

#### met. Lorem ipsum dolor-sit amet, consetetur

Extracted text: Lorem ipsum dolorsit amef, consetetur Character Error Rate: 0.1818 Word Error Rate: 0.5714

#### • CycleGAN with ResNet (with VGG Loss):

amet. Lorem ipsum dolor sit amet, consetetur

Extracted text: m§tV§i'VLo rem ipsurn amef, consetetur Character Error Rate: 0.4318 Word Error Rate: 0.8571

#### • CycleGAN with U-Net (without VGG Loss):

amat. Lorem ipsum dolor-sit amet, consetetur

Extracted text: amet. Lorem ipsum dolorrsit amet. consetetur Character Error Rate: 0.0455 Word Error Rate: 0.4286

#### • CycleGAN with U-Net (with VGG Loss):

amet. Lorem ipsum dolor sit amet, consetetur

Extracted text: Character Error Rate: 1.0 Word Error Rate: 1.0

Similar to the example provided earlier, we computed the mean of Character Error Rate and Word Error Rate for 100 different sentences, and obtained the subsequent outcomes for each model:

Model	With VGG	Mean CER↓	Mean WER $\downarrow$
Source	-	0.6435	0.8125
Pix2Pix with ResNet	False	0.0494	0.2376
Pix2Pix with ResNet	True	0.0522	0.2521
Pix2Pix with U-Net	False	0.0625	0.3041
Pix2Pix with U-Net	True	0.0640	0.2933
CycleGAN with ResNet	False	0.2379	0.5773
CycleGAN with ResNet	True	0.6497	0.7817
CycleGAN with U-Net	False	0.3101	0.5791
CycleGAN with U-Net	True	0.8340	0.9243

Table A.6.: Test case 4 OCR recognition results table

Table A.6 presents the evaluation of 9 different cases on 100 sentences, using two evaluation metrics, mean CER and mean WER, for OCR recognition. The models being compared include the source image with no enhancements, as well as several variations of the Pix2Pix and CycleGAN models with different network architectures, and with or without VGG Loss.

Firstly, we can see that the source image without any enhancements has the higher mean CER and mean WER compared to all the models except for CycleGAN with U-Net and VGG, which indicates that the models are improving the accuracy of the OCR recognition.

Among the Pix2Pix models, the best performance was observed for the model using ResNet without VGG Loss, with a mean CER of 0.0494 and a mean WER of 0.2376. This model outperformed the Pix2Pix models using U-Net architecture, which had higher mean CER and mean WER values. Adding VGG Loss did not significantly improve the performance of the Pix2Pix models.

For the CycleGAN models, the best performance was observed for the model using ResNet without VGG features, with a mean CER of 0.2379 and a mean WER of 0.5773. However, this model's performance was still not as good as the best-performing Pix2Pix model. Adding VGG Loss to the CycleGAN models had a negative effect on their performance, as all the models had higher mean CER and mean WER values when VGG Loss were added.

Overall, the results suggest that the Pix2Pix model using ResNet architecture without VGG features is the most effective model for OCR recognition, as it outperforms all the other models in both mean CER and mean WER values. It is interesting to note that adding VGG Loss did not improve the performance of the OCR, and in some cases, had a negative effect. However, further experiments may be needed to confirm these results and to explore the performance of these models on different OCR recognition datasets.

#### Test Case 5:

In this test case, the main goal was to conduct a comparative analysis of the effectiveness of the Pix2Pix and CycleGAN models in enhancing the quality of scanned images by varying the learning rate for each model. Specifically, we aimed to determine whether decreasing the learning rate has a significant impact on the stability and performance of the models. The focus of the analysis is to evaluate the models' capability to eliminate noise and imperfections, such as coffee stains, from the images and to examine the impact of the learning rate on their performance. To ensure a fair comparison, we maintained a constant set of hyperparameters throughout the experiment, as listed below:

General Hyperparameter	
Adversarial loss function	LSGAN loss
Normalization type	Instance normalization
Discriminator architecture	70 × 70 PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Special Hyperparameter	
Pix2Pix L1 Lambda	100
CycleGAN cycle Lambda	10

For this test case, the learning rates utilized were 0.0001 and 0.0002.

#### Quantitative Comparison:

Similar to the preceding evaluations, the fifth test case begins with a quantitative comparison approach to assess model performance. An objective evaluation of the models will be performed using three metrics: PSNR, SSIM, and FID (see Section 5.2). The performance is presented through line charts that display the progress at 5-epochs over the course of the 100-epoch training period, analogous to the prior tests. Performance metrics for each learning rate will be represented through subchart, demonstrating each model's progress throughout the training process.

The results depicted in Figure A.13 reveal that a learning rate of 0.0001 marginally degraded the PSNR performance of the Pix2Pix models when compared to a learning rate of 0.0002, while maintaining consistent performance across all epochs. Conversely, learning rate 0.0001 exhibited superior PSNR performance for CycleGAN models, particularly in the early epochs with the ResNet architecture and generally with the U-Net architecture.

Consistent with previous test cases, the Pix2Pix model with ResNet architecture retains its superiority over the other models, even when the learning rate is altered, as evidenced by the findings. It is worth highlighting that the model yielded the highest PSNR of 22.64 dB in the 80th epoch with a learning rate of 0.0002 and the highest PSNR of 22.37 dB in the 80th epoch with a 0.0001 learning rate, which are close results with a slight drop when using a smaller learning rate. Similarly, the Pix2Pix model with the U-Net architecture exhibited a similar pattern.



Figure A.13.: PSNR line charts (higher is better) for test case 5

However, both architectures of the CycleGAN models demonstrate improved performance at a learning rate of 0.0001, with the ResNet architecture of CycleGAN still outperforming U-Net. Remarkably, the CycleGAN model with ResNet architecture achieved the highest PSNR of 18.13 dB at epoch 30 with a learning rate of 0.0001 and the highest PSNR of 17.22 dB at epoch 15 with a learning rate of 0.0002.



Figure A.14.: SSIM line charts (higher is better) for test case 5

The SSIM metric displayed in Figure A.14 exhibits a trend parallel to that of PSNR across all models. This indicates that the Pix2Pix models for both generator architectures show a good ability to preserve structural information for both learning rates with a very small superiority in the learning rate of 0.0002. On the other hand, reducing the learning rate shows a good improvement in the performance of the CycleGAN models in preserving structural information, but Pix2Pix is still highly concordant. Noting that the performance of CycleGAN, with a learning rate of 0.0001, declined after epoch 65 under a learning rate of 0.0002.



Figure A.15.: FID line charts (higher is better) for test case 5

As we can see in Figure A.15, the FID metric also shows similar behavior to PSNR and SSIM, which reinforces the observations made by these metrics. The Pix2Pix models with ResNet and U-Net architectures consistently outperform the CycleGAN models in terms of FID scores as the other metrics, which indicates that they are better at generating images that are closer to the ground truth.

Moreover, using a learning rate of 0.0001 had a slightly negative effect on the performance of the Pix2Pix models, as they achieved higher FID scores in some epochs compared to a learning rate of 0.0002. However, the same cannot be said for the CycleGAN models, as FID scores with a learning rate of 0.0001 are shown to be lower than a learning rate of 0.0002, interestingly, the quality of the CycleGAN models worsens as the training progresses for both learning rates.

To gain a more understanding of the overall performance of each model, we can examine the mean of their scores across all epochs for each metric. This is presented in Figure A.16.

As shown in the bar chart A.16, reducing the learning rate leads to a slight decrease in the performance of the Pix2Pix models for all metrics. However, decreasing the learning rate results

#### A. Test Cases Appendix



Figure A.16.: Test case 5

in an overall improvement in the quality of the CycleGAN models. We can see the results more accurately in table A.7

Model	Learning rate	Mean PSNR ↑	Mean SSIM ↑	Mean FID ↓
Pix2Pix with ResNet	0.0001	21.9005	0.9446	3.6975
Pix2Pix with ResNet	0.0002	22.1263	0.9487	3.3654
Pix2Pix with U-Net	0.0001	19.5116	0.9207	5.9305
Pix2Pix with U-Net	0.0002	19.5325	0.9207	5.7343
CycleGAN with ResNet	0.0001	15.9100	0.7409	30.3451
CycleGAN with ResNet	0.0002	15.7593	0.7211	35.0972
CycleGAN with U-Net	0.0001	13.2257	0.5892	74.9787
CycleGAN with U-Net	0.0002	11.7820	0.4952	106.3209

Table A.7.: Test case 5 results table

#### Visual Inspection:

Following the analysis of the quantitative measurements, we will proceed to visually inspect various samples to validate the effectiveness of the models. Specifically, we will examine just the good performing epochs, which include Pix2Pix with ResNet at epoch 90, Pix2Pix with U-Net at epoch 85, CycleGAN with ResNet at epoch 15, and CycleGAN with U-Net at epoch 30. The following tests showcase the outcomes of each example for a given input:



• Pix2Pix with ResNet:



• Pix2Pix with U-Net:



• CycleGAN with ResNet:







• CycleGAN with U-Net:







Note that the red lines are not originally present in the images and have been added to highlight specific details.

The aforementioned example serve as a representation of one sample that was tested. For further samples, we kindly direct you to the appendix B where additional samples are available for review.

#### **OCR recognition:**

This section aims to evaluate the effectiveness of the models in improving the accuracy of Tesseract OCR. The model's performance will be assessed on different test samples, as Tesseract OCR can encounter difficulties with low-quality scanned images. Therefore, the goal is to showcase the performance of the models on various test samples to determine its ability to enhance the accuracy of Tesseract OCR in text extraction using character error rate (CER) and word error rate (WER), which were introduced in Section 5.3.

Sample: sea takimata sanctus est Lorem ipsum dolor sit

• Source image:

sea takimata sanctus est Lorem ipsum dolonsit

Extracted text: Strange and incorrect symbols were extracted Character Error Rate: 1.0217 Word Error Rate: 0.875

#### • Pix2Pix with ResNet (With Learning rate 0.0001):

#### sea takimata sanctus est Lorem ipsum dolor sit

Extracted text: sea takimata sanctus est Lorem ipsuim dolbrsit Character Error Rate: 0.0652 Word Error Rate: 0.375

#### • Pix2Pix with ResNet (With Learning rate 0.0002):

#### sea takimata sanctus est Lorem ipsum dolor sit

Extracted text: sea takimata sanctus est Lorem ipsuim dolhr sit Character Error Rate: 0.0435 Word Error Rate: 0.25

#### • Pix2Pix with U-Net (With Learning rate 0.0001):

#### sea takimata sanctus est Lorem ipsum dolorisit

Extracted text: sea takimata sanctus est Lorem ipsuim dolbnsit Character Error Rate: 0.087 Word Error Rate: 0.375

#### • Pix2Pix with U-Net (With Learning rate 0.0002):

#### sea takimata sanctus est Lorem ipsum dolorisit

Extracted text: sea taklmata sanctus est Lorem ipsuvm dolbrxsit Character Error Rate: 0.087 Word Error Rate: 0.5

#### • CycleGAN with ResNet (With Learning rate 0.0001):

sea takimata sanctus est Lorem ipsum dolorisit

Extracted text: sea lakimata sanctus est Lorem ipsuim dolbrxslt Character Error Rate: 0.1087 Word Error Rate: 0.5

#### • CycleGAN with ResNet (With Learning rate 0.0002):

sea takimata sanctus est Lorem ipsum dolonsit

Extracted text: sea takimala sanctus est Lorem Ip§u'm dolE>nsi1 Character Error Rate: 0.1739 Word Error Rate: 0.5

#### • CycleGAN with U-Net (With Learning rate 0.0001):

sea takimata sanctus est Lorem ipsum dolorisit

Extracted text: sea takimata sanctus est Lorem iosu\_rn dolbrxsit Character Error Rate: 0.1304 Word Error Rate: 0.375

#### • CycleGAN with U-Net (With Learning rate 0.0002):

sea takimata sanctus est Lorem ipsum dolorasit

Extracted text: sea takimata sanctus est Lorem ipsum dolomslt Character Error Rate: 0.0652 Word Error Rate: 0.25

Similar to the example provided earlier, we computed the mean of Character Error Rate and Word Error Rate for 100 different sentences, and obtained the subsequent outcomes for each model:

Model	Learning rate	Mean CER↓	Mean WER↓
Source	-	0.6435	0.8125
Pix2Pix with ResNet	0.0001	0.0523	0.2520
Pix2Pix with ResNet	0.0002	0.0494	0.2376
Pix2Pix with U-Net	0.0001	0.0598	0.2888
Pix2Pix with U-Net	0.0002	0.0625	0.3041
CycleGAN with ResNet	0.0001	0.1765	0.4957
CycleGAN with ResNet	0.0002	0.2401	0.5750
CycleGAN with U-Net	0.0001	0.3198	0.6123
CycleGAN with U-Net	0.0002	0.2935	0.5868

Table A.8.: Test case 4 OCR recognition results table

Table A.8 shows the performance of the different models and learning rates for OCR recognition, as measured by the mean CER and WER. The first row in the table labeled "Source" represents the baseline performance without any image-to-image translation. The other rows represent the performance of different models using different learning rates.

From the table, we can see that the best-performing model is Pix2Pix with ResNet using a learning rate of 0.0002, CER of 0.0494, and WER of 0.2376. This model outperforms the baseline by a large margin, indicating that image-to-image translation can significantly improve OCR recognition performance.

We can also observe that using a higher learning rate generally leads to worse performance in some models, as seen in the higher CER and WER values for Pix2Pix with U-Net and CycleGAN with ResNet when trained with a learning rate of 0.0002, compared to 0.0001. This is likely due to overfitting caused by the higher learning rate, which can cause the model to converge too quickly and not generalize well to new data.

Overall, this table provides insight into the impact of different image-to-image translation models and learning rates on OCR recognition performance. However, to ensure the reliability of these findings and to explore the efficacy of these models on diverse OCR datasets, it may be necessary to conduct further experiments, as the results are somewhat divergent from the quantitative performance evaluation. The underlying reason for this discrepancy lies in the fact that our study only employed a sample of 100 sentences, which is comparatively small, and the law of small numbers posits that such small samples can potentially lead to inconclusive or equivocal outcomes.

#### Test Case 6:

The main goal of this test case was to perform a comparative analysis of Pix2Pix models to evaluate their effectiveness in improving the quality of scanned images. The analysis compared the performance of the models with and without the L1 loss function, which is a component of the Pix2Pix loss function (see Section 4.2.3). The primary focus was to determine whether the addition of the L1 loss function to the cGAN loss function had a significant impact on the model results. The evaluation primarily concentrates on assessing the capacity of the models to eliminate noise and defects, such as coffee stains, from the images and the influence of the L1 loss function on the performance of the Pix2Pix models. To ensure a consistent and fair comparison, we kept the hyperparameters constant throughout the experiment, and they are listed below:

General Hyperparameter	
Adversarial loss function	Min-Max loss
Normalization type	Instance normalization
Discriminator architecture	70 × 70 PatchGAN
Optimization method	Adam optimizer with beta values of (0.5, 0.999)
Batch size	1
Learning rate	0.0002

#### **Quantitative Comparison:**

The sixth test case employed a quantitative comparison approach, similar to the previous cases, to evaluate the performance of the models. The evaluation is conducted objectively using three metrics: PSNR, SSIM, and FID, as explained in Section 5.2. Performance will be presented through line charts, which will display the progress of the models in 5 epochs over a 100 epoch training period with and without L1 Loss.

The results of the test case are presented in Figure A.17, which demonstrates that using the L1 loss function leads to a significant improvement in the PSNR performance of the Pix2Pix models at all epochs when compared to not using it. This improvement is due to the ability of the L1 loss function to preserve small details during image conversion from one domain to another.

The Pix2Pix model with ResNet architecture was the best performer, achieving the highest score of 22.32 dB in epoch 95 when the L1 loss function was used. On the other hand, the highest score without L1 is 21.48 dB in epoch 95. Similarly, the U-Net architecture Pix2Pix model, achieving a score of 19.34 dB in epoch 45 when the L1 loss function is utilized, whereas the best score without L1 is 18.76 dB in epoch 70.

Figure A.18 shows a similar pattern in the SSIM metric to that of the PSNR for all models. This observation implies that the Pix2Pix models with both generator architectures exhibit an enhanced ability to preserve structural information when the L1 loss function is used in almost all epochs. Among the two models, the Pix2Pix model with ResNet architecture remained the best performer, achieving the highest score of 0.9528 in the 55th epoch when the L1 loss



Figure A.17.: PSNR line charts (higher is better) for test case 6

function was used, while the highest score without L1 was 0.9412 in the 95th epoch. In addition, the Pix2Pix model with the U-Net architecture also behaved similarly, achieving a score of 0.9184 in epoch 45 when using the L1 loss function, whereas the best score without L1 was 0.9130 in epoch 70. Interestingly, the model without L1 outperformed at epoch 100 for this metric.



Figure A.18.: SSIM line charts (higher is better) for test case 6

The results of the FID scores reinforce our observations from the previous metrics, as shown in Figure A.19. The models that incorporate the L1 loss function consistently outperform the others across all the cases. In addition, the use of L1 provides greater stability during training, resulting in less fluctuation in the results. Conversely, the absence of L1 often leads to fluctuations in results.



Figure A.19.: FID line charts (higher is better) for test case 6

To gain better insights into the performance of each model, you can look to the Figure A.20, which presents the mean score for each metric across all epochs.



Figure A.20.: Test case 6

As we can see in the bar Figure A.20, reducing the learning rate leads to a slight decrease in the performance of the Pix2Pix models for all metrics. On the other hand, decreasing the learning rate results in an overall improvement in the quality of the CycleGAN models. We can see the results more accurately in table A.9

Model	With L1 Loss	Mean PSNR ↑	Mean SSIM ↑	Mean FID $\downarrow$
Pix2Pix with ResNet	False	20.5234	0.9267	10.7174
Pix2Pix with ResNet	True	21.8609	0.9464	4.0700
Pix2Pix with U-Net	False	17.8971	0.8935	13.4092
Pix2Pix with U-Net	True	18.9144	0.9124	6.6881

Table A.9.: Test case 6 results table

#### Visual Inspection:

Once the quantitative measurements have been analyzed, we will proceed to visually inspect various samples to validate the models' effectiveness. Specifically, we will examine only the epochs that performed well, such as Pix2Pix with ResNet at epoch 90, Pix2Pix with U-Net at epoch 85, CycleGAN with ResNet at epoch 15, and CycleGAN with U-Net at epoch 30. The subsequent tests will demonstrate the outcomes of each example for a given input:



• Pix2Pix with ResNet:



• Pix2Pix with U-Net:

	Without LI L035
and," said 1, and the dense is for it seems to u for $y_{2}$ and I don't much care; for it seems to u for $y_{2}$ a little damaged in the head. But if $y$ has been been been been been been been bee	$U_{j}^{*}$ said $U_{j}^{*}$ with the damaged in the head. But if $y_{jk}$ and I don't much care; for it seems to II $y_{jk}$ and Itale damaged in the head. But if $y_{jk}$ and Itale damaged in the head. But if $y_{jk}$ and $U_{jk}$ and
bout in en sure you do all i sure." figer pointed and eye levelled at the Pequ figer pointed and eye levelled at the Bequ figer pointed and eye levelled at the Pequ figer pointed at the Pequ	that it en-sure, you do all i
auries down on the papers Captain	thes down on the papers
d; and what's to be, will a solution of the so	and what's to be, with any how you, that
; and some sailors or as well these as any out it, eh-	and some sailors of as well these as any ofut it ch-

Note that the red lines are not originally present in the images and have been added to highlight specific details.

The aforementioned example serve as a representation of one sample that was tested. For further samples, we kindly direct you to the appendix B where additional samples are available for review.

#### OCR recognition:

The purpose of this section is to assess how well the models improve the accuracy of Tesseract OCR. The objective is to demonstrate how the models perform on different test samples and determine their ability to enhance Tesseract OCR's accuracy in extracting text using character error rate (CER) and word error rate (WER), which were introduced in Section 5.3.

Sample: elitr, sed diam nonumy eirmod tempor invidunt ut

• Source image:

elitr, sed diam nonumy eirmod tempor invidunt ut

Extracted text: nonum eirmod tem or invidunt ut Character Error Rate: 0.375 Word Error Rate: 0.75

#### • Pix2Pix with ResNet (With L1 Loss):

#### elitr, sed diam nonumy eirmod tempor invidunt ut

Extracted text: elitr, sed diam nonumy eirmod tempor invidunt ut Character Error Rate: 0.0 Word Error Rate: 0.0

#### • Pix2Pix with ResNet (Without L1 Loss):

#### elitr, sed diam nonumy eirmod tempor invidunt ut

Extracted text: elitr, sed diam nonumv eirmod tempor invidunt ut Character Error Rate: 0.0208 Word Error Rate: 0.125

#### • Pix2Pix with U-Net (With L1 Loss):

elitr, sed diam nonumy eirmod tempor invidunt ut

Extracted text: elitr, wed gigm nonumy emnod tempor invidunt ut 7 Character Error Rate: 0.1667 Word Error Rate: 0.5

#### • Pix2Pix with U-Net (Without L1 Loss):

etitr, sed diam nonumy eirmod tempor invidunt ut

Extracted text: e titr, =ed giam nonumy e|rmqcl\_tgmgor inv:du\_r1t ut 1 Character Error Rate: 0.3542 Word Error Rate: 1.0

Similar to the example provided earlier, we computed the mean of Character Error Rate and Word Error Rate for 100 different sentences, and obtained the subsequent outcomes for each model:

Model	With L1 Loss	Mean CER↓	Mean WER $\downarrow$
Source	-	0.6435	0.8125
Pix2Pix with ResNet	True	0.0476	0.2374
Pix2Pix with ResNet	False	0.0652	0.3014
Pix2Pix with U-Net	True	0.1124	0.4402
Pix2Pix with U-Net	False	0.1442	0.5070

Table A.10.: Test case 4 OCR recognition results table

Table A.10 presents the results of different Pix2Pix models with/without L1 Loss for improving OCR recognition. The performance of the models is measured by mean CER and mean WER, where lower values are better.

The first row of the table labeled "Source" represents the baseline performance without any improving. The subsequent rows represent the performance of different Pix2Pix models with ResNet and U-Net architectures, both with and without L1 Loss.

The results indicate that all the Pix2Pix models outperform the baseline performance, as evidenced by the lower CER and WER values. Among the models, the Pix2Pix with ResNet architecture and L1 Loss achieved the best performance with a CER of 0.0476 and a WER of 0.2374. On the other hand, the Pix2Pix with U-Net architecture and no L1 Loss achieved the worst performance with a CER of 0.1442 and a WER of 0.5070.

In general, adding L1 Loss to the Pix2Pix models appears to have a positive impact on OCR recognition, as the models with L1 Loss achieved better performance than those without it. The ResNet architecture also appears to be better suited for OCR recognition, as it outperforms the U-Net architecture in all cases.

#### **B. Results Appendix**

This appendix contains additional image inputs that were used for visual inspection as part of our research analysis. These images were collected to provide a more comprehensive understanding of the data and to ensure the accuracy of our results. Visual inspection is an important step in many research projects as it allows for the identification of potential errors or anomalies that may not be evident through statistical analysis alone. By including these additional images in our results appendix, we aim to provide a more transparent and thorough presentation of our research findings.

#### Test Case 1 and 2 Results

Input

#### • Pix2Pix with ResNet:



#### ch Norm

With Batch Norm	
J nUO ITZ NQ HOUSE, b.b. McDonald * co., NEW YORK CTTY. mile INCREASING BUSINESS OF OC A ESTABLISHIENT. together with c vants of the trade, have made it necessary estalified a house in the Oily of NEW YOR Our R, IL McDonadi is box in that city, a from the facilitae' for ourchasine, shiroin	
etc, which we now have, we feel confide that we can supply the trade at much low rates than heretofore.	
We invest Assession to see in tota- carefully selected stock of superior Dra and Medicines, Chemicals of all kinds, So and Fluid Extracts, Shaker's Herbs, Eclee Preparations, Patent Medicines, Fancy goo Trusses and Shoulder braces. Abdominal Su Trusses and Shoulder braces. Abdominal Su chies Druggists Glassware. Breverss Not Soda Water Muieral, Panits und Oils, Ket sine and Coal Oils. Lard And Sperm Oi Catanbene and Burning Huid. Actobia in th	
Together with a complete aaaortmr of all articles in the Ding Due. We do not Intend to b- undersold by an bouse in California. Orders respectful	

個ラしきン選諸須ば。円オメスキ題子 レヌ電来へム手論提席マ探美に航謝 最6覧マテツヌ高機ふ県連足えぶる。 るゆ人親廃り育73説ど暮女フ量投ヒ初 リすリきせ高申よせとら況演悪ーごまめ 6表がきも報豊灯キ覚引らねぼ付覧ラ スナ備企リか権航ヘケ況徳チヒヤ自信 「ば。円オメスキ題子ケエウ如堀ーシレ 提席マ探美に航謝定ヱ総午紙ヨソ最 機ふ県連兄えぶる。全ばず性世をぼ 説ど暮女フ量投と初会ざどン企新利 とら況滴悪ーごまめ。生ば社携よ生6 とゆこうち もの 一部 アンド モデドリ

CDSDRTED-CDSDL CHARGE-31 - NACE MALLES, COMMUNICATION OF A STATE AND A STAT

mp foot aransa the  $k^2$  price the boost of the boost of the inter-restructables areas for her also of any boost were in a most-meant. Ever and mone a bright, hot, also, deceptive idea would dart you through,—It's the Mack Sea in a mindight gala.—It's he unshard and an a bright, hot, also, deceptive idea would dart you through,—It's he Mack Sea in a mindight gala.—It's blated heath,—It's a Hyperborean winter scene.—It's the blated heath,—It's a Hyperborean winter scene.—It's the preaking-up of the loo-bound sterm of Time. But at last all dues fincies yielded to that one portontous something in tho-gigantic field reven the great lowing and bring the rest wave plain. But stop; does it not bear a faint resemblance to a generative investigation on the aggregated opinious of many aged persons with whom I conversed upon the subject. The picture proposels a Gape-Horner in a great hurricase; the hald-foun-dered miny wettering there with its three dismanticed masts one within an an exaptenet whale, purposing in anging class over the earth, this scenarios of impaling himself . The opposite wall of this earth years high and upon. Some were were tufied with knots of human hair; and one was iddex-shaped, with a vast handle sweeping r-und like the sequent made in

#### With Instance Norm

the farm of J. D. Ste-	J nUOIfZNQ HOUSE,
cordance with a plan on file in said office,	b.b. McDonald * co.,
nd, the Board of Su-	m II E INCREASING BUSINESS OF OCI
uld they think them	A ESTABLISHMENT, together with the wants of the trade, have made it necessary to estali'i.h a house in the Oity of NEW YORK
d of Supervisors. 3.	Our R. II. McDonald is bow in that city, and from the fncilfue? for purchasing, shipping
ES E. 8 ILL, Pres't.	ctc, which we now have, we feel confiden that we can supply the trade at much lower
DNTRACTORS.	rates than heretofore.
LS will be received by visors of Yolo County and, until Monday, the	Wk invite Attention to our implant carefully selected stock of superior Drag and Medicines Chemicals of all kinds. Soli

We list Artestine to our instances of superior State carefully selected stock of superior State and Fuid Entracts. Shaker's Henris Henrice Toron and Fuid Entracts. Shaker's Henris Edecide Proparations. Patent Medicines. Fancy goods, Tursses and Shoulder haves. Abdominal Sup-ribles Druggists Glassware. Brevers Stock Soda Water Maieril, Paints und Olis, Kero sine and Coal Olis, Lard Auf Sperm Olis, Camphene and Duming Maid. Acadomin time TogeThet with a complete and Distance We do not intend to b- undersol by any bouse in California. Ordern respectfully visors of Yolo County and, until Monday, the 63, for the erection of bw Slough, near the to be done in accord-pecifications now on ork to be paid for hy e County Road Fund, s re'erving the right of said bids, If they 1 of Supervisors. 5. .ES E. SILL. Pres t

Douse in California. Orders respectfull ロークリーンション 個ラしきン選諸須ば。円オメスキ題子 ルマ電来へム手論提席マ探美に航謝 最6覧マテンジス高機ふ県連足えぶる。 るゆ人親廃り育73説ど暮女フ量投ヒ初 リすリきせ高申よせとら況演悪ーごまめ 6表がきも報豊灯キ党引らねぼ村覧ラ マナ備企りか権航へケ況徳チヒヤ自信 ぼ。円オスキ題子ケエウ如堀ーシレ 提席マ探美に航謝定工総午紙ヨソ最 機ふ県連兄えぶる。全はず性世をぼ 説ど暮女フ量投ヒ初会ざどい企新利 とら況演悪ーごまめ。生ば社獲よ生6 と学びいち しんかい しょうちょう

**EGUMMES CANCE CONTACT CONTACT** 

my floct arrains that  $\frac{M_{H}}{M_{H}} \sim ter;^{-1} ters beyows from narrange the series as a non-$ rananstate series are series as a more series of a series of the series the series of the series of the series of the series series of the series of the series series of the series series of the series series of the series of

#### • Pix2Pix with U-Net:



#### With Batch Norm

onwood Stougn J nUOIIZNQ HOUSE, in suid office, b. b. MrD. b.b. McDonald \* co., on file in said office. t by warrants di wir ind, the Board of Su-right to reject any one uld they think them an file in said office. b. b. retters at a 'Co., by warman & WW TORK CUTY. and the Board of So-and the Board of So-and the Works of the INCREASING OF OCR Market Solution (Statistical), and the day which have an the Dity of NEW YORK cut all back which have an the Dity of NEW YORK Cut all the Works of the Solution (Statistical), and the Solution (Statistical), and the NER at the Solution (Statistical) and PATERCLORE
THE UNIT DEFINITION OF THE D to Supervisors. to Supervisors. to Supervisors. LES E. SILL. Prest We do not Intend to b- undersold by any bouse in California. Orders respectfully UTTLETTION 個ラしきン選諸須ば。円オメスキ題子

レヌ電来へム手論提席マ探美に航謝 最6覧マテツヌ高機を県連足えぶる。 るゆ人親廃り育73説ど暮女フ量投ヒ初 リオリきせ高申よせとら況演悪ーごまめ 6表がきも報豊灯キ党引らねぼ何覧ラ スナ備企りか権航へケ況徳チヒヤ自信 しば。円オメスキ題子ケエウ如堀ーシレ 最席マ探美に航謝定エ総午紙ヨソ最 機ふ県連兄えぶる。全はず性世をぼ B説ど暮女フ量投ヒ初会ざどン企新利-どら況演悪ーごまめ。生ば社携上生6 と登引たりには展らみ いごに思わたっ

 $2-4\pi$  **ITT.**  $4-12\pi$  **ITT.**  $4-12\pi$  **ITT.**  $1-12\pi$  **ITT.**  $1-12\pi$ 

the glown towards the north with the dafkness towards hot contu-wherever in your window you may coexide to lodge for the night, any daw kilomed, its sures to inquire the price, and dark its to aparticular. Write induity apps I pased the strengt, and pased the sign of "The Corosed Harnon"—but its loaded to arguestic and july these. Further can from the height red windows of the "Swood-Fah lan," there came such ferent rays, that it assends to have nabled the packed around and form. Is don't be house, for entrywhere she the campaind from layer to the house, for entrywhere she the campaind from layer to the child its and, aphalitic promotion-tasker nearly for any, where I starts with search to task. Increase from the house, four heat of the where the high task. Increase from the house.

The sequence the finity projection, hearing from hand, single many the finity projection, hearing from hand, single many to trick a harrown from the bowquit. How having angles, a c' - and cit is early tright. However, server and angles is a single of the second second second second provide the second second second second second second and the second second second second second second second and the second second second second second second second with the second second second second second second second and the second to be second to second second second second second to second second second second second second to second second second second second second second to second second second second second second second to second second second second second second second second to second second second second second second second second to second second second second second second second second to second to second s

Associated by a set of the set o

breaking-up of the loe-bound stream of Time. But a this till -line-shares arisely side loc table. -one-portrons--onsething-in-the-picture's midit. Thet one-found-own-and silt-the-rest-ave-phin. But stop: (does in not bers a faint resemblance to a genoric fab I even the great lovalithan himself I In fact, the striked-kedings assemd-the is-striked-theory of my ora, partly based upon the aggregated opinion of many aged pences with whom I owversed upon the subject. The pictures represents a Cope-Homere in a great hurriconce, the half-foun-ded ship velocing these with its three dismutaled masts about the strength of the strength of the strength of the strength about the strength of the strength of

clean over the send, is in the numranov upon the three matcheads. The opposite wall of this entry was hung all over with a healtanish array of monotous olitos and spars. Some were thickly set with glittering to their penething 'rory may; cahere-ver tunke with show of human hir; u do new as side/shaped, with a wat handle-wreeping r-rand-like-the-sequent make in

#### With Instance Norm

he farm of J. D. Ste-	J nUOlfZNQ HOUSE,
cordance with a plan m file in said office. by warrants di wi id, the Board of Su- ight to reject any one ild they think them.	b.b. McDonald * co., NEW YORK CITY. Sn HE INCREASING BUSINESS OF OCR A ESTABLISHMENT. together with the wans of the trade, nave made it necessary to etdFilth house in the Other of NEW YORK
l of Supervisors.	Oun R. II. McDonald is bow in that city, and from the facilities for purchasing, shipping,
ES E. 8 ILL, Pres't.	etc., which we now have, we feel confident that we can supply the trade at much lower
INTRACTORS.	rates than heretofore.
S will be received by visors of Yolo County rd, until Monday, the 53. for the erection of w Slough, near the o be done in accord- occifications now on ork to be paid for hy e County Road Fund. i re'erving the right of said bids. If they	We level, Altreature 16 error 16 error carefully selected steck of superior Drag <sup>4</sup> and Medicines, Chemicals of all kinds, Solid Proparations, Pattern Medicines, Energy goods, Programmer Asserver, Material Medicines, Energy goods, portex Asserver Materials (Hotentral and Crin- cibles Drungsits (Hassware, Brewer, Stock Sola Water Material, Patints und Olis, Kero sine and Coal Olis, Lard And Seerro Olis, Cataphene and Zuming Huid, Acoholin Int.
of Supervisors.	Together with a complete aaaortmrat of all articles in the Ding Dne.
ES E. SILL. Pres t	We do not Intend to b- undersold by any bouse in California. Orders respectfully
個ラしきン 遅	<b>麗諸須ば。円オメスキ題子</b>
1 ヌ電来へ.	ム手論提席マ探美に航護
最6覧マテツ	ヌ高機ふ県庫兄えぶる。
るゆ人親廃	の育73説ど暮女フ量投ビネ
すりきせ高い	申よせとら況演悪ーごまめ
6表がきも報	豊灯キ賞引らわぼ付覧
スナ備金川ナ	いた航へケ況徳チャャ自
	本語子、ケマウ加爾一、ハ
提供った主	に転動会マ総ケが出
機や県理工	になる。主はり住世をは
説と春女ノ	重投ビ初会さど、企新利
とら況演悪ー	こまめ。生は社携よ生6
七世日にわ	年は脂ラサルに圧損った
to throw at the whi nigh enough to risk	alos, in order to discover when they were a harpoon from the bowsprit?

sign energin to risk a harpoon from the brought! Now having a night, a day, and will another uight following before me in New Stefford, end I could embork for my defined perturbations and the stefford of the stefford of the stefford and shop meanwhile. It was a very dubition-bolding, may, a very davh and disma leight. Hittiggs outd and cheviess. I Knew no use in the pizze. With anxions grapped I had somethic my problem, and only brought, up a dw pizzer of silver\_os, wherever you go, lahmad, and it to mynelf, at I tool in the mblied of a davey starts shoulding my logg at comparing the gloom tormain the moth with the ulainess towards bey solut\_where the start without on any constroled to bolies.

the gloon towards the north with the dailness towards the south—wherever in your wisdow you may conclude to lodge for the sight, my deve klammel, to save to inquire the price, and don't to to oparticular. With hading steps I pased the strends, and passed the steps of "Ha Consed Harroons—but it looked to expressive and july these. Further ca, from the hright red windows of the "Bwool-Visah linn," there came such forcent rays, that is assend to have milded the packed arour and is from before the house, for everywhene she the campealed from lay to the blows. for everywhene she the campealed from lay the inter the line and, spatiality grounds\_arouty for any form any first mail assents the firsty projecticas, because from hardy with someth to take, horecose from the horemuti

back, aphabilic parenteuty—rather eventy for naw, when I started may foce against the finity projections, because form hard, might enough to risk a harpoon from the bowsprit. Now having a night a day, and still nearbar night following poorts in an RWP instrum, and y and still nearbar night following poorts in a RWP instrum, and y and still nearbar night following poorts in a still dismain night, hingly cold and checkers. This are been still a start of consermment where I was to et-er of a start and the start of a start of a start of a to be the start of the start of a start of a start of a nearbard start of the start of the start of a start of a start of the start of the start of the start of the start and the start of the start and the start of the start of the start of the start of the start and the start of the start of the start of the start of the start start of the start part of the start the start part of the start of the start of the start of the start the start of the start the start part of the start of the start start of the start of the start the start part of the start of the start start of the start start of the start start of the start of th

maked mean-rate a supprovem array control of the source with an array of the second steam of the source of the so

#### • CycleGAN with ResNet:



#### With Batch Norm

n U O IfZ N Q HOUSE, b.b. McDonald \* co., b. 11 FORMATING BUSINESS OF OCR ALE TABLESING BUSINESS OF OCR ALE STABLESING BUSINESS OF OCR Ventperific that the constraints of the constraints establish a house in the Oay of NEW YORK OMR. INAME Alexand up above in that city, and from the Incliftee for purchasing, shipping ter, which we now have, we to cle confident that we can hupply the trade at much lower rates than hereofolge. t to reject any c they think the 1 of Supervisors. 5 E. 8 ILL, Pres't. ONTRACTORS LS will be received by visors of Yolo County and, until Monday, the 63, for the erection of pw Slough, near the to be done in accord-pecifications now on york to be paid for hy the County Road Fund. W1 Bis2 ( UTS10) . File and and the control of superior carefully valected stock of superior and Medicines. Chemicals of all kinds, and Fluid Extracts. Shaker's Herbs- be Preparations. Patent Medicines, Fancy Trusses and Shoulder braces. Abdomuse pattern & asswer Materials (Hort And cost Winterstein). The stock of the cost of the stock of the the stock of stock re erving the right of said bids. If they of Supervisors. of Supervisors. St E. SLL. Prest Structure for the sumplement of the supervisors. St E. SLL. Prest Supervisors. St E. SLL. Prest Supervisors. Structure for the supervisor of th of Supervisors ES E. SILL, Pres t 個ラしきン選諸須ば。円オメスキ題子 レヌ電来へム手論提席マ探美に航護 最6覧マデッマ高機ふ県連足えぶる。 るゆ人親廃り育73説ど暮女フ量投ヒ社 リすりきせ高申よせとら況演悪ーごまめ 6表がきち報費灯や第一らねぼ付覧う スナ備企リか権航ヘケ況徳チヒヤ自 ほば。円オメスキ題子ケエウ如堀ーシー 最席マ探美に航謝定エ総午紙ヨソ量 機ふ県連兄えぶる。全ばず性世をぼ 説ど暮女フ量投ヒ初会ざどン企新利 とら況演悪ーごまめ。生ば社携よ生6 と呼こうちちなかしま こうぶ 玉さち is the state is a second state of the state of the second state of the state is a second state of the stat

#### 117:+l, T

Cottonwood Slough	
ine farm of J. D. Ste-	J nUO IfZNQ HOUSE,
on file in said office.	b.b. McDonald * co.,
id, the Board or ou-	IN HE INCREASING BUSINESS OF OCI
ild they think them,	a bouse in the Oily of NEW YORK
of Supervisors.	Ourl & INMs Bonald is bow in that city, and from the facilities for burchasing, shipping
S E. 8 ILL, Pres't.	tter, which we now have, we feel confider that we can supply the trade at much lower
NTRACTORS	rates than heretofore
S will be received by isors of Yolo County	Wh invite Artention : out on 45 carefully selected stock of superior Drag
3. fot the erection of	and Medicines, Chemicals of all kinds, Soli- and Fluid Extracts, Shaker's Herbs, Eclecti
be done in accord-	Preparations, Patent Medicines, Fancy goods Trusses and Shoulderbraces, Abdominal Sup
ork to be paid for hy	cibles Druggists Glassware Brewers Stock
re erving the right of said bids. If they	sine and Coal Oils. Lard And Sperm Oils Catmbere and Burning fluid A'cohol in the
of Supervisors.	Together with a complete aaaortmra
ES E. SILL. Pres t	We do not Intend to b- undersold by any
<u>, / y / / y   =</u>	Bouse in Canfornia. Orders respectfully
個ラしきン	解諸須ば。円オメスキ題-
レヌ雷来へ	ム手論提席マ探美に航話
最6階ーテッ	マ高機な県連兄こらこ
るゆん親庭	り音73説ど暮女フ最投とな
しかりたれた	カトキレに沿海亜ージセン
19 20周	
のなかさり新	国人生地力のななり見
スナ備企り	い権机へケ況徳チヒヤ目
は。円本立	、年期子生于如掘一ジ
提席マ探美	<b>に航謝定</b> ヱ総午紙ヨソ
機ふ県連咒	えぶっ。全ばず性世をは
説ど暮女つ	量投と初会ざど: 企新利
とら況油悪。	
と登出にち	ほたちをうせ いいい 出 トエ
to throw at the wh	alco, in order to discover when they were
nign enough to risk	a harpoon from the bowsprit?
Now having a nig	rbt, a day, and still another night following
port, it because a n	naiter of concennant where I was to eat
and sleep meanwhi very dark and disma	le. It was a very dubious-looking, nay, a l night, hitingly cold and cheerless I knew
no one in the place	. With auxious grapnels I had sounded
my pocket, and only wherever you go. L	y brought up a new pieces of silver,
middle of a dreary	street shouldering my bag, and comparing
south-wherever in	your wisdom you may conclude to lodge
for the night, my d and don't be too par	lear Ishmsel, be sure to inquire the price ticular.
With halting step	s I paced the streets, and passed the argn of
jolly there Furthe	r on, from the bright red windows of the
"Sword-Fish Inn," i to have melted the	here came such fervent rays, that it seemed
for everywhere else	AND TO THE THE TO NOTE OCIOID COS DOLLAR.
	the congealed front lay ten inches thick in a
hard, asphaltic paves	the congested frost lay ten inches thick in a ment, -rather wears for me, when I struck the finity projections, because from bood
hard, asphaltic paves my roos against the nigh enough to	the congealed frost lay ten inches thick in a ment, -isther weary for me, when I struck is finity projections, because from hard, risk / na. poon trom the bowspri
hard, asphaltic paves my 1006 against th nigh enough to Now having a	las congested frost lay ten inches thick in a ment, -rather weary for me, when I struck as mary projections, because from hard, risk : na noon trom the bowson a night a :a 'a other nigh low.
hard, asphaltic pave my root against th nigh enough to Now having a genore me in Ng port, it tecama	he congenied front by tree inches thield in a ment, -rather weary for me, when I struck as many projections, because from hard, risk r na noon from the bowsport i anght ara 'a.c.ther right. low: w Benjordt recomment where I was no a matter of concernment where I was no a
hard, asphaltic pave my 1006 against ti nich enough to Now having a perton it tecams en 1	Ine congredied from lay ten inches their in a methy -rable wears for me, when I struck is many projections, because from hand, risk's na noon tron the bowren in gight a i a the methy is a wear of the structure in the structure in the wear of the structure in the structure in the methy of the structure in the structure in the methy of constraints when it was no st in the structure
hard, asphaltie pave my 1006 against ti Now having a petore me in Ng port, it tectma and analy a and to one-sa tion	he consended from lay ten inches their in a methy -stable wear for me, when I struck is many projections, because from hard, risk r an zoon tron the bowson a gight a
hard, aphaltic pave my 1006 against 11 nigh enough to Now having a port, it tecams and the ny dat and di to one in Ag up port, it tecams and the ny dat and di to one in Ag	The comparison from the two index the kin is not account of the two index that the two is as many properties of two weaks from hard, which as a cone to main the however the is a cone to the two is the two is the matter of concernment where I was a co- ing the two is the two is the two is a matter of concernment where I was a co- ing the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two is the two only browshift the a set - two is of silter. Is the there is the two is the two is the two is the two is the two is two is the two is the two is the two is the two is two is the two is the two is the two is the two is two is the two is the two is the two is the two is two is two is the two is the two is two
bard, aphaltic pave my 1006 against U nigh enough to Now having a port, it tecams and itecams and itecams and itecams and itecams and itecams	The comparison from by the inclus theking in matrix -taskier waves from as, when I struck the maxy projections, because from hard, inglets
bard, asphaltic paves my 1006 against U nigh encough to Nor hains : centre in the second of the second second rational second second rational second second rational second second rational second second models of a second the second second second rational second	The compared from by the incluse theking and the structure that the st
hard, asphalie pare my 1006 aganst U nigh ecourt to Nor barine; before me m Ng port, it Jecama and it. and di Jo Obs-15 Lec me port. And terrat vte middle or a the artntot. art. or Us night	The correspond from by the inclusion thick in a first projection, because from new ball struck in many projections, because from hard, which a conce more has been in this random term that a structure of the st
hard, asphaltis pave my rook against ut nish account io Now having , george is a second to Now having , good is a second of the second second middle or a second of the second second second of the second second second of the second second second of the second second second second of the second second second second second of the second second second second second second of the second second second second second second second of the second se	The comparated from by the inclusion thek in a maxy tropectors, because from new hard struck is many proportions, because from hard, the nanows more hard a supplet $S_{\rm eff} = 1.2$ and $S_{\rm eff} $
hard, apphaltis pare my roos against u nish ecourts to Nor haring: cooror me in Ng port, it decans wy ds. and di at or the second the second second second the second second second the second second second the second second second second the second second second second the second second second second second the second second second second second second the second second second second second second the second second second second second second second the second second second second second second second second second second second second second second second second second second second seco	The comparison from the two inclusions that in a mark variable wave from new ball struck from hard, which wave more than a struck from hard, which was not a subscription of the structure of th
hard, arphaltis pare my roos against u nich ecourt to Nor haring: cooro me m Mg port, it dectors wy da and di or <del>Jones 1.2</del> wy port and decourt v 6 middle or a s the m_n local decourt to to fit halter "Thecool to U there to "Thecool to U there to "Bootto to	The correspond from by the inclusion their in a mark -table wave from new ball struck in a mark projection, because from hard, in this na cone form the become inght, $c_{-12} = 1 + 2 + 2 + 2 + 2 + 2 + 2 + 2 + 2 + 2 +$
hard, arghaltis pare my nos agams u lish ecouth to Nor having: - course me Mg port is testens any ds. and di to Observations and the second second and the second second me second second the second second second second second second second the second second second second second second second second the second second second second second second second second second the second second second second second second second second the second	The corrected from by the inclus thick in a function of the second structure
hard, apphaltis pare my roos against u wip nose against u Nor havies : eeorome me hig port is testina wy da. and di so Orran Lee wy poi . so at terrar tterrar whe er, n loss with - whe ar have a to have terrar tterrar the er, n loss terrar tth thethe the halos - arthe set of the night terrar tothis halos - "Thecould in halos - o have methed orrai to be a	The composite from by the inclusion their in a mark -stable wave from ne, when I structure is mark proportions, because from hard, the saccose from the based of the saccose from the saccose is a substantial signal of the saccose from the saccose is a substantial signal sign
hard, suphatic pare my nos agams u Nor having: coors man Mg pot, i beens m Mg pot, i beens of da - add of <del>Simple Lieb</del> we por i ara tratthe middle or a the strat  	the original from by the inclusion their in a mark value wave for me, when a transfer one, when a transfer that a concern them a mark of the saccore thread a gamma for a mark of the saccore thread a gamma for the saccore thread a gamma for the saccore thread a gamma for the saccore thread a saccore the saccore thread a saccore thread saccore thread a saccore t
hard, suphaltic pare my nos agams. U Nov havie: costor me n Ng port. It decime and the non- my read of the non- my read of the rest of the non- my read of the rest of the non- my read of the rest of the non- net the non- net the non- the nort test. . Tith halfor "The	The original foot by the inclusion thek in a mark -table wave from ne, when I structure in a mark projection, because from hard, in this rances can be achieved a mark of the rance of the structure in the struc
hard, arghnitis pare my nos argunas u Nov havies : cooron me n Mg port, is centra arg ds. and db to Orong Let my foor and argent to the middle of a my foor and argent to the with halor with halor of the standard to the sta	The original foot by the inclus their in a first matrix - table ways for me, when I struct the III to projection, because from hard, which is a consort more the beword in this rank one structure is a structure that is a structure of the structure that is a structure of the str
hard, apphaltis pare my 1004 against u nish eccouch to Nor having: cooror me m Mg port is testing and the second to the second testing of the second and the second testing of the second the second testing of the second testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of testing of t	The original foot by the inclusion their in a function of the second se
hard, suphaltic pare my roos against u Nov having: ecours me in Mg port, i testions wy root against the second second second or the supha or the supha or the supha or the supha or the supha course the supha supha second second table these the back means of the back monoger	the comparison from by the inclusion that in a mark -stable wave from ne, when I structure is a mark proportion, because from hard, in this - a cone from the basened to the stable of the stable of the structure is a structure in the structure is a mark of the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in the structure is a structure in the structure in t
hard, apphalite pare my roos agams: u Nov having: : cooror nen Ng port i testens ary 4. and 3 to 0 m here and ary 4. and 3 to 0 m here are the sight of a middle of a m the str. of Lar with hele c the sight of a sight of a m the sight of a middle of a m the sight of a middle of a mi	He original foot by the inclus their in a first marked from the large that the started th
hard, arphaltis pare my 1005 argumst u Nor hariss: : control nor m Mg port is centre and the main set of the set of the set of the my do and di to Object to the my do and di to be an expected to the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the set of the	The original from by the inclusion thek in a first structure of the second st
hard, suphaltic pare my roos against u Nor having: coorse non NA port 1 beens of the second of the port 1 beens of the second of the normalized of the norma	The outgoind food by the inclus thek in a runk that have the inclusion of the second
hard, anghaltis pare my nos agams. U Nor havie: econo me n Ng port, i tectors with a sector of the sector with a sector of the sector with a sector with a sector with a sector with a sector with a sector at the sector with a s	the original foot by the inclus thek in a large hard start of the second start of the
hard, apphaltis pare my nose argums: u Nor having: costor me n Ng port, it detains and the second second and the second second and the second second the second second second second second the second second second second second the second second second second second second the second second se	The original foot by the inclus thick in a true in the structure is a mary projection, because from hand, in this random the basened is an interpreter in the basened is a structure of a structure of the struct
hard, apphaltis pare my nos agamas u visite accounts to Nor havies: control mess m Mg port, is centra wy do. and di to Oro-Labor wy do. and di to Oro-Labor wy do. and di to Oro-Labor wy do. and di to Oro-Labor with halor with halor with halor with halor with halor or a second of the oro-Labor with halor with halor or a second of the oro-Labor based hash- based hash- branking-app of hast stop gigments hi I ow In fact, the out over party based over the holes of the oro-labor based hash- branking-app of the theory fact of the oro-labor In fact, the out over party based over the holes of the over party based	be original foot by the inclus thick in a function that in a most transform a when a transform that a structure of the struc
hard, explaitie pare my roos agemas u nich ecouch io Nor hanice: coorse non Ng port i beens of the second of the s	the ourgend from by the inclus thick in a function of the second

The opposite wall of this dary was hung all over with a heathenish array of monstrous allos and spears. Some were thickly set with glittaring teeth resembling ivory saws; others were tufked with Knote of human hair; and one was cickle-shaped, with a vast handle sweeping r-und-like the segment-made in
Input With Batch Norm n U O IfZ N Q HOUSE n U O IfZ N Q HOUSE of J L ste e with a plan b.b. McDonald \* +> h.b. McDonald - co. NEASING BUSINESS OF O JoinNESS OF OCI avieNT together with the le, nave made it necessary it in the Oity of NLW YORK ald iss bow in that city. for purchasing astives of oct astives and a necessary to use in the Oity of NEW YORK ponald is box in this cus-fic for purchase ASING BUSINESS OF tali'isle a hou: uttR: LIAMel& of Supervisors. of Supervisors. E. 8 ILL. Pres E. S ILL, Pres TRACTORS INTRACTORS INTRACTORS Swill be received by isors of Yolo County al, until Monday, the 3, for the erection of w Slough, near the be done in accord-celifications now on ork to be paid for hy county Road Fund ₩1 - 6d]s Wt carefull- elected stock of superior I and Medicines. Chemicals of all kinds, and Fluid Estrates. Shakers Herbs. Lu Preparations: Patient Medicines Functe Universe and Shoulderbraces. Advontin is rubies to more state and the state of the chiles Dimensional State and Other Sold Water Mairent. Panits und Oth. J sine and Coal Othe. Lard And Soem Catanohere and Burning Tuluk A coholi II carefull<sup>10</sup> selected stock of super and Medicines, Chemicalos of all kin and Fluid Extracts, Shaker s Herb Preparations, Patent Medicalense, Far Trusses and Shoulderbraces Abdio parter a-kszwarer Matériale (lettort-chles Drugslist Glassware Brewe Soda Water Maleral, Palitas und C sino and Coal Olis, Lard Auf Cataphene and Burning fluid, A'co erving the right id bids. If they re erving the right said bids. If they Logether with a complete of all articles in the Di We do not Intend to b- und bouwa un Collegeta Together with a complete aaa of all articles in the Ding I we do not intend to b- undersol Supervisor ES E. SILL. Pres t ES E. SILL, Pres t " " " 近平 " 7772Lª 以早ててへノ加正リが推測 V用エピンム 加上ソル 個ラしきン選諸須ば。円オメスキ題子 個ラしきン選諸須ば。円オメスキ題子 個ラしきン選諸須は。円オメスキ題子 レヌ電来へム手論提席マ探美に航藤 最6覧マテツヌ高機を県連兄えぶる。 レヌ電来へム手論提席マ探美に航藤 最6覧マデジス高機な県連見えぶる。 ルヌ電来へム手論提席マ探美に航静 最6覧つデジス高機を県連足えであ。 るゆ人親廃り育73説ど暮女フ量投ヒ初 るゆ人親廃り育73説ど暮女フ量投ビ初 るゆ人親廃り育73説ど暮女フ量投ビ初 すりきせ高申よせとら況演悪ーこ すりきせ高申よせとら況演悪ーごまめ \*=\* すりきせ高申よせとら況演悪ーごまめ 6表がきも報豊灯キ党引らねぼ付置 6表がきち報豊灯キ党引らねぼ付置う 6表がきち報望灯キ党引らねぼ何覧う スナ備企りか権航へケ況徳チヒヤ自住 スナ備企りか権航へケ況徳チヒヤ自信 十備企りか権航へケ況徳チヒヤ自住 は。円<u>オススキ組子ケエウ如場</u>シル 提席マ探美に航謝定ユ総千派ヨンル 機ふ県連見えぶる。全任学性世をほ 「ば。円オズを題子ケエウ如堀ー、ル 目ば。円オスキ題子ケエウ如堀一シル 静席マ探美に航期定 ヱ総十紙ニツ最 諸提席マ探美に航謝定ヱ総午紙ヨソ最 機ふ県連先えぶる。全ば工性世をぼ 機か県連先えぶる。全は丁性世をぼ 説ど暮女フ量投と初会ざどン企新利 とら況演悪ーごまめ。生ば社携よ年6 説ど暮女フ量投ヒ初会ざどい企新利 説ど暮女フ量投ビ初会ざどい企新利 とら況演悪ーごまめ。生ば社携上生6 とら況浦悪ーごまめ、生ば社機上生6 ト 母 己 い ち 気 子 育 り キ マン 市 玉 す ナ を登立にそこそれです。 2-cMF AILC 4-DEF/ABC -2-LE -2-NEF/MA 2-ext for these at the owner of the second owner owner the inge energy to raik a harpoon fause the lowestark! New haring a might a day, and still moders injut following before min is New Belfeld, see Tould analysis of the second parts is because a matter of consenting and any own dusting parts is because a matter of consenting and any own dusting and shown management. It is thinking and the second and the mean management of the second and second any because on own of the Defined, see a Toulous 4-booking may, a wave dark and design and the second and because the owner on the parts. With managements 1 and control mp possets and can be moder up as a passes of alwer-ross, wherever vue on confined and in the dainased to owners is south—wherever in your wadom you may constain the posses and dust be too particultar. to the nicht, my date klammel. In same to laquine the prior and dark bit to constraints. wran naming stars I pased the streets, and passed the sign of "The Crossed Harroon"—but it locked to a caracteristic and starts. The constraints of the street is the streets to have anised the packed near and iso from before the busi-band, aphabilit, parametic, "starks" even any the integer hard, aphabilit, parametic, "starks" even y for may be determined who explained the flux type of the two is types of block in g hard, aphabilit, parametic, "starks" even y for may be determined who explained and the flux type of the two is types of block in g hard, aphabilit, parametic, "starks" even y for may be determined. Now having angular, a "starks" even starks (starks), and and distingt high starks, and the starks is starks, and distingt high starks, and the starks is a starks and distingt high starks, and the starks and the starks and distingt high starks, and the starks and the starks and distingt high starks, and the starks and the starks and the stark is a stark of decommendant when it is a stark of the starks and durit to tao particular. "With making steps I passd the stream, and passed the sign of "The Ground Harmoons"—but it looked too excensive and join "how Further can be took hight rad windows of the "downd-bia law." Here cause such formations to seema to have multied the passiest arow and loo from biofore the Jonan to have multied the namesiad formation to the strength with a first excerning and the compared of the two to be the biological hard, explainting particular, the shafter every fan may sheen I struges we took examine the first processing because from hard, we took examine the first processing because from hard, we took examine the first processing because from hard, ther weary for me, when I stru-projections, because from ha st the fin at the finty ne my where where port. and and ente hno velut -iai of of silv 12A tl-s night, \_\_\_\_\_ an ... Prove of erve ... . andel . ticul vith haltnesse seen to street "The present through ""-but it h folly there purther on no through as the horizontal second through a strength of the streng streat o uco en ren vandovis ... - 11 JOW Nword is i have seen "es from bete" the "ost lay teu inches the ini Can inch % . for me, when a strug \_sphs1t eads: of this entry was hung all ov nonstrous onbs and spears. So ring teeth resembling ivory saw

#### With Instance Norm

7作用间山

1215.14 Log-

pde

810\*\*

# **Test Case 3 Results**

#### • Pix2Pix with ResNet:

Input

Lorem ipsum dolor sit amet, consetetur sadipscing eilir, sed som nonumy eimod tempor invidunt ut labore et olore magna alguyam erat, sed diam voluptua. At vero eos et/accusam et justo duo dolores et ea rebum. Stet clira kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit andib.com ipsum dolor sit amet, consetetur sadipscing eilir, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et pusto duo dolores et ea rebum. Stet clira kasa gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit finet, conselbrut sadipscing elir, sed diamrunumy ormod tempor invidunt ut labore et dolore nagna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stut clira kasa gubergren, no sea takimata sanctus eu corem insum dolor sit amet. um dolor sit amet, consetetur sadipscing F Г Note magna anguyam erat, seo diai vere eos et accusam et justo duo a rebum. Stet clina kasal gubergran in ipsum dolor sil amei, consettut ellir, sed diam nonumy elimod term abore et dolore magna higuvern or buptur. At vero eos et accusam et et ac arabum. Stet, clina kasal gubers et ac arabum. Stet, clina kasal gubers sei cai reoum Siet cuite kaso guoen mata sanctus sei Loren ingura dolo mipsum dolor sit amet, conseteur elitz, sei diem nonumy elimad teng labore et dolore magna aliquyam er olumum. At vero eos et acctissim et seites rebum. Stet diri keva guben matg sanctus est Lorem ipsum doin دَّ من أن تُتلقى بَعَض الأَلفَاظ من اللُّغاتِ المُجاورَة النّ أتي بِأَمْثَلَة كَثِيرة على مَزايا اللُّغة العَربيَّةِ في لاشْتِقَاق وَطُواعِبَةِ النِّعبير وَدَقْةُ الدَّلالَة وَمُرونَةِ التَّرْكِي وَكَتَابُنَا هٰذا لا نُرِيدُه كَتَابَ لُغَة، وَنَخْشى أَن يُؤَدِّي وليس الأمر كذَّلك، فإنَّ في تَعلُّم اللُّغات كُلُّ جبُ وتكاد تَخْتَفي إذا كانَتِ اللُّغَة حَيَّةً يتَحدَّثُ ولا يُعلِّمونَها تَعليماً فَحسب. وَالمَصاعبُ التي مردَّها كلها إلى المرحلة التَّاريخية التي يتجتازها مُهرة النَّاس يَكْفيهم مُجرَّدُ البِّيان للإعرابِ عن حا マテツヌ高機ふ県連兄えぶる。全ばず性世 ヨソ最6號マ

リン最高度マテリア高速及展連邦之志る。とす了学性世 を任る少人環境的育力就ど奪去フ量投む初会ざとご金 等利利力等をは伸発せらら認識及して、なよれ北 ま主を決だきも報題灯や意引し、「貸付電ラサノソ近単 マモスナ備全」が冷観インプの増テンと可能プレンジ単 マモスナ備全」が冷観インプの増テンと可能プレンジ単 アモスナ備全」が冷観インプロ 思え、これ、なく超子とマロを引つとなる 事態優勝や気濃に気急なくして、すべきなな のないため、 事態度が高いたい、 をしたの発展にないため、 などの表現し、 に、 をしたの発展したと大日留できござかを新行りできて高中 をじたの発展したと大日留できござかを新行したでありま をしたいため、 をしたの子供生したものなど、 なられたいため、 なられたいため、 をしたのないため、 なられたいため、 なられため、 なられたいため、 なられため、 なられたいため、 なられたり、 なられたり、 なられたいたいため、 なられため、 なられため、 なられたり、 なられたり、 なられたいたちか。 なられたり、 なられたり なられたり、 なり、 なられたり、 なりたり、 なり、 なり なりたり、 なりたり、 なり、 なり、 なり、 なりたり、



#### MinMax Loss

Aria

Lorem ipsum dolor sit amet, consetetur sadipscing Lorem pisum dolor sit amet, consertetur sacipscing, ellir, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilia kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sacipscing elitr, sed diam nonumy eimod tempor invidunt ut bieven et delore erange olineures per sadipscring elitr, sed diam nonumy eirmot tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmot tempor invidunt ut labore et dolore magna aliquyam erat, end diam vulnutin. At vero nes et accusam et justo sed diam voluptua. At vero eos et accusam et justo duo dolores et ea reburn. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet.

Arial ×C Lorem ipsum dolor sit amet, consetetur sadipscing r eilit; sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam volupua. At vere ose et accusam et justo duo dolores et aa rebum. Stet ciita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitt, sed diam nonumy eirmod temoor invidunt ut labore et dolore magna aliquyam erat, sed diam volupua. At vero ese et accusam et justo duo dolores et ea rebum. Stet ciita kasd gubergren, no ese takimata sanctus est. Lorem ipsum dolor sit auo doiores et ea rebum. Stet citta kaso gubergren, no sea takimata sanctus es Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elmod tempor invidunt ut labore et dolore magna allquyam erat, sed diam voluotuga. At vero eos et accusam et justo duo dolares et ea rebum. Stet dita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit avort.

لدَّ من أن تُتلقى بَعض الأَلفُاظ من اللُّغات المُجاورَة الت

نَاتِي بِأَمْثلة كَثيرة على مَزايا اللُّغة العَربيَّةِ في الاشتقاق وَطَواعيَةِ التَّعبيرِ وَدِقَة الدَّلالَة وَمُرونَةِ النَّركيب وَكتانُنا لهٰذا لا نُرِيدُه كتَابَ لُغة، وَنَخْشى أَنْ يُؤَدِّي العَربيَّة، وليس الأمر كذلك، فإنَّ في تَعلُّم اللُّغاتُ كُلُّم تحتَجبُ وتكاد تَخْتفي إذا كانَتِ اللُّغة حَيَّةُ يتَحدَّثُ ناً، ولا يُعلَّمونَها تَعلَّماً فَحسبَ. وَالمَصاعبُ التي تَع مردُّها كلها إلى المَرحلة التَّاريخية التي يَجْتازها ا ا تَمام الأتقان.

جَمْهرة النَّاس يَكْفيهم مُجرَّدُ البَيان للإعرابِ عن حاح

Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eignod tempor invidunt ut labore et dolore magna eliquyam erat, sed diam voluptua. At vero ecs et accusam et justo duo dolores et ea rebibm. Stei dila kasi gubelgren, no sea takimata sanctus est Lorem ipsum dolor sit amet Lorem.njesim dolor sit amet, conseletur sadipscing clitr, sod diam nontimy olimică tempor invidunt ut labore et dolore, magna aliquyam erat, ac ed diam voluptua. At vero ese et accusam et lusto duo dolores et ea rebum. Stei clita kasi gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ecs et accusam et porte en elitro en est et corem setti no sea takimata sanctus est Lorem ipsum dolor sit amet. orem ipsum dolor sit amet, conseletur sadipscine

#### LSGAN Loss

Aria

Lorem ipsum dolor sit amet, conseletur sadipscing elitt, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliguyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet dita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos est accusam et justo Lorem ipsum dolor sit amet, consetetur sadipscing sed diam voluptua. At vero eos et accu am et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit

Arial Lorem ipsum dolor sit amet, conseteur sadipscing , elltr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese at accusam et justo duo dolores et ea rebum. Stet cilita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetu sadipscing elltr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese at accusam et justo duo dolores et ea rebum. Stet cilita kasd gubergren, no sea takimata sanctus est. Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elltr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluotu. At vero ese at accusam et justo duo dolares et ea rebum. Stet cilita kasd gubergren, no sea takimata sanctus est. Lorem ipsum dolor sit no sea takimata sanctus est Lorem ipsum dolor sit

لدَّ من أن تَتلقى بَعض الأَلفُاظ من اللُّغات المُجاورَة الت نَاتِي بِأَمْثِلَة كَثِيرة على مَزَايَا اللُّغَة العَرِبِيَّةِ في سَلا

الاشْتِقَاق وَطُواعيَةِ التَّعبيرِ وَدِقَة الدَّلالَة وَمُرُوَنَةِ النَّركب وَكَتَابُنَا لِهٰذَا لا نُرِيدُه كَتَابَ لُغَة، وَنَخْشى أَن يُؤدِّي لعَرِبِيَّة، وليس الأمر كذلك، فإنَّ في تَعلُّم اللُّغات كُلَّ حتَجبُ وتكاد تَخْتفي إذا كانَتِ اللُّغة حَيَّةٌ يتَحدَّثُ ناً، ولا يُعلَّمونَها تَعليماً فَحسبَ. وَالمَصاعبُ التي تَع مردُّها كلِها إلى المَرحلة التَّاريخية التي يَجْتازها ا تَمام الاتْقان.

جَمْهرة النَّاس يَكْفيهم مُجرَّدُ البَيان للإعرابِ عن حاج

Lorem ipsum doior sit amet, conseleitur sadipscing elitr, sed diam nonumy eimod tempor invidunt ut labore el 100 re magna aliguyam erat, sed diam voluptua. At vero eos et accusam et justo duo doiores et ea rebum. Stei dila kasa gubetgren, no sea takimata sanctus est Lorem ipsum dolor sit amet 1 oram Inpum dolor sit amet, conseletur sadipscing olitr, sod diam nonumy oimod tempor invidunt ut labore et dolore magna aliquyam erat, « sed diam voluptua. At vero eos et accusam et justo duo doloras et ea rebum. Stei dita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stei ciita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. **1** orem ipsum dojor sit amet, conseletur sadio

#### WGAN Loss

Anal

Lorem ipsum dolor sit amet, consetetur sadipscing elitr, ad diam nonumy ermot tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese staccusam et justo duo dolores et ea rebum. Clet clita kasd gubergren, no sea takimata sanctur est Lorem insum dolor sit amet. Lorem ipsum dolor sit amet, consetetur adipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum. Stat clita kasd gubergren, no sea takimata sanctus est Lorem insum dolor sit amet. Lorem ipsum dolor sit amet sonsetetur sadipscing elitr, sed diam nonumy eirmod tempor unvidunt ut labore et dolore magna aliquyam erat, Lorem ipsum dolor sit amet, consetetur sadipscing invidunt ut labore et dolore magna aliguvam erat sed diam voluptua. At vero eos et accu m et justo duo dolores et ea rebum. Stet clita kasd gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet

Arial AND THE <u>۳</u> a. Loram tosum dolor sit amet, consister in sa osus g Litin, <u>"art diam</u> nonumy em<u>od tempor</u> \_\_\_\_\_\_int ut labore, dolore manne aliquyam erat, sad diam oluguna At verse at 1.2 oggam <u>\*\_\_</u>sto duo tolores at earcutus ast Lorom insum dolor : area taxinata canctus ast Lorom insum dolor : area taxinata canctus ast Lorom insum dolor : adipsuno ettir, "ad diam nonumy et in od temt a unvidunt ut labore at soorer, magna aliqu'm erat, ast diam vit on. At are use at acrusse et lorom 

لدَّ من أن تَتلقى بَعض الأَلفَاظ من اللُّغات المُجاورَة الت

أتى بأَمْثلة كَثيرة على مَزايا اللُّغة العَربيَّة في سَلا الاشتقاق وطَّواعيّة التّعبير وَدِقَّة الدَّلالَة وَمُرونَةِ التَّركيب وَكتَابُنَا هٰذا لا نُرِيدُه كتَابَ لُغْهَ، وَنَخْشى أَن يُؤَدِّي لعَربِيَّة، وليس الأَمر كَذَلك، فإنَّ في تَعَلُّم اللُّغات كُلَّ حتجبُ وتكاد تَخْتفي إذا كانَتِ اللُّغة حَيَّةُ يتَحدَّثُ أَنه ولا يُعَلِّمونَها تَعليماً فَنحسب. وَالمَصاعبُ التي نَ مودُّها كلما إلى المرحلة التَّاريخة التي يَحتازها ا تمام الأثقاق.

ممهوة النَّاس يَكْفيهم مُجرَّدُ البَّبان للإعراب عن حا-

مهود الناس بكموم مُجرد البان للإمراب عن حال معلم المحلم المحلم

Lorem ipsum dolor sit amet, conseteutr sadipsong elitr, sed diam nonumy eiging d tengor invidunt ut lahore et i dolore regins elitiguan erat sed dolores et es rebuin. Site ditta kasd gubérgen, no sea takimata sanctus est Lorem ipsum dolor sit anet 1 nem ipsum dolor iti amet conseteur cadipsoling ditts, cad duram nontimy elithod tempor-imvidunt ut labore et dolore magna aliquava rest, es so daam rotucus - vero, ece et accusam et une to dolores et es e a recum "itel dig sad quenerren no set, talamate, send ter dolore sit amet Lorem ipsum dolor sit anet, Lorem ipsum dolor sit anet, Lorem ipsum dolor sit anet labore et dolore magna aliquava rest duo dolores et e a recum "itel dig sad quenerren no set, talamate, conclute cel "oren, ipsum dolor sit amet, Lorem ipsum dolor sit anet, lorem et psum dolor sit anet anata sanctus est Lorem et psum dolor sit amet anata sanctus est Lorem ipsum dolor sit amet anata sanctus est Lorem ipsum dolor sit amet anata sanctus est Lorem psum dolor sit amet anata sanctus est Lorem psum dolor sit amet anata sanctus est Lorem psum dolor sit amet anata sanctus site come teres site anata sanctus est Lorem psum dolor sit amet anata sanctus site come teres site anata sanctus site site anata sanctus site site anata sanctus site come teres site anata sanctus site anata sanctus site anata sanctus site site anata sanctus sit orem ibsum golor sit amet, sunseletur sad

#### Input

Anal Lorem issum dolor sit amet, consetetur sadipscing eilir, sed com nonumy eimod tempor invidunt ut labore et colore magna aliguyam erat, sed diam voluptua. At vero eos etaccusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem jesum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing eilir, sed diam nonumy eimod tempor invidunt ut labore et dolore megna aliquyam erat, sed diam voluptua. At vero ese et accusan et pusto duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem jesum dolor sit amet. Lorem ipsum dolor sit finet, consetetur sadipscing elir, sed diam nonumy alimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusante invidunt ut labore et dolores magna siguyam erat, sed diam voluptua. At vero ese et accusante diam colores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est, corem ipsum dolor sit amet.

psum do

negna allquyam erat, sed dia eos at accusan el jasto duo um. Stat diakas jubergan india si al subergan motor si ame i constatur esta di orise majeri constatur est di orise majeri constatur est di orise majeri allquyam el anbur. Stat chi ta read guber sanctus esi Lorem ipatri, dona est di ann nonumy elimo d'aug est di orise magna allquyam er est di orise magna allquyam era est di orise magna di orise di orise di orise est di orise magna di orise di orise di orise est di orise magna di orise di orise di orise di orise est di orise di orise di orise di orise di orise di orise est di orise di orise di orise di orise di orise di orise est di orise di orise di orise di orise di orise di orise est di orise di orise di orise di orise di orise di orise est di orise di orise di orise di orise di orise di orise est di orise est di orise est di orise d

nes al ea rebum. Stel cilita kasă gubergen Nimata sanctus est Lorem ipsum dolor sit

لدَّ من أن تُتلقى بَعض الأَلفُاظ من اللُّغاتِ المُجاورَة الت

نَاتِي بِأَمْثِلَة كَثِيرَة على مَزَايَا اللُّغَة العَرِبِيَّةِ في سَلا

الانتيفاق وتلواعية النّبيم ودينة الذلالة وترونة التركيمية وتتنائط هذا لا ترويد، كتاب لغة، وتنخش أن يُوَدِّق المُريبية، ولين الأمر كذلك، فإن في تملم اللغات كُلْ تحتجي وتكاد تنخشي إذا كاني اللغة حَمّ يتحدُّثُ

ولا يُعلِّمونَها تَعليماً فَحسب. وَالمَصاعبُ التي

ردُها كَلْهَا إلى المرحلة التَّاريخية التي يَجْتَازها

بهرة النَّاس يَكْفيهم مُجرَّدُ البيان للإعراب عن حا

又高機な県連兄え

ヨン最6覧マ

. ... ころ、全ばず性世

#### MinMax Loss

Arial

Aria

Lorem ipsum dolor sit amet, consetetur sadipscing elitt, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et.accusam et justo duo dolores et ea rebum. Stef citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor sadipscing eilir, sed datin nonumy eimod tempor invidunt ul talore et dolore magna aliquyam erat, sed diam voluptua. At vero ecs et accusam et justo duo dolores et a rebum. Stet cita kasag dubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseteur sadipscing elitr, sed diam nonumy eirmod tempor sed diam voluptua. At vero eos et accusam et justo duo dolores et ea reburn. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. invidunt ut labore et dolore magna aliguyam erat

E E Lorem losum dolor sit amet, consetetur sad psur g citit, sad diam por<u>umy</u> ermod tempor invidunt ut labor v; dolores magna aliguyam erat; sad diam volugiua. At vero eoc et accus<u>am erijusto duo</u> doloe se et a rebum. Stet dina kasa gubergrei, . sen takimata sanctus est Lorem ipsum dolor sit andt. Lorem lipsum dolor sit amei; consetetui sadipscing elitr, sed diam nonumy elimod tem.cor uividunt ut labore et dolore magna aliguyam erat, sad diam volupiu. At vero eos et accusam et justo duo dolores et eo. 3bun "site citita trasd gubergrein, no sea takimata sanctus es. Lore m ipsum dolor sit amet. Lorem losum dolor sit amet, consetetur sadipscing elit, sed dam nonumy elimod ten.por invifunt ut labore et dolore magna aliguyam erat, sed diam volunia. At vero eos et accusam et justo duo dolores et eu ebum. Stet dita krsd gubergrein, no sea tati mata sanctus es. Lorem ipum dolor sit duo dolorare al eu rebum. Stet dita krsd gubergrein, no sea tati mata sanctus es. Lorem pum dolor sit Lorem ipsum dolor sit amet, consetetur sadiosci g no sea taଧmata sanctus est Lorem ipsum dolor sit amet.

#### لدَّ من أن تَتلقى بَعض الأَلفَاظ من اللُّغات المُجاورَة الت

نَاتِي بِأَمْثِلة كَثِيرة على مَزايا اللُّغة العَربيَّةِ في سَلا الاشتيقاق وَطُواعيَةِ التَّعبيرِ وَدِقْنَه الدَّلالَة وَمُرونَةِ ٱلنَّرِك وَكتابُنا لهذا لا نُريدُه كتَابَ لُغة، وَنَخْسى أَن يُؤدِّي المَربِيَّة، وليس الأَبَر كذلك، فإنَّ في تَعَلَّم اللُّغات كُلُّ وحتج وتكاد تَنْخفي إذا كانَت اللُغة حَيَّة يَتَحَدَّثُ إ ناً؛ ولا يُعَلِّمونَها تَعَلَّيماً فَحَسَبٍ. وَالمَصاعَبُ التي تَع مَردُّها كلها إلى المَرحلة التَّاريخية التي يَابُجَازها ا ا تمام الأثقان.

جَمْهِرَةِ النَّاس يَحْفِيهِم مُجَرَّدُ البَيَانِ للإعرابِ عن حا

コノ最低な<sup>24-10</sup>、高裕県東辺兄ぶら。全代す「仕世 をはるや人場題等有助認と違い」「最大いのなどとな 新村りりをせ活申ませとらば河路×こまめ、生ば社が よきながたもい最近年登りに「ほ付覧ラナリン近単 ヤモスナ幅立」が活動、ペッパ焼手をいき値得可、きン強 新気は、同サ「マス理子」とつが知一シレス現象、ム 手論提振っ訳美に航前定す您午知コリス気質、ペッ っ高機ら、現立したぶる。(ボブド社をにざるゆ、マッ っ高機ら、現立したぶる。(ボブド社をにざるゆ、マッ っ高機ら、現立したぶる。(ボブド社をにざるゆ、マッ っ高機ら、現立したぶる。(ボブド社をにざるゆ、マッ っ高機ら、現立したぶる。(ボブド社をにざるゆ、マッ っ高機ら、現立したぶる。(ボブド社をにざるゆ、マッ っ高機ら、現立したぶる。(オビ社集)によるかも利 豊かて近れたいこと、などの表情に、 すった。 なら、シニゴザー性をにざるり、現め方「コル」をあった。 なら、シニゴザー性をにざるい、現むの ため、シニゴザーという違い。 ため、こというなが したいが、また。 「見つ」したいでは、「オンスンを 「見つ」したいでは、「オンスンを 「見つ」したいでは、「オンスン 「見つ」したいでは、「オンスンを 」」 ヨソ最6覧マデッツー高機を県連兄えぶる。全ばす性世

#### Aria

Lorem Ipsum doior sit amei. consetettir sadipschig elitt, sed diam nonumv ermod temporiavidunt ut labne et Alotter munu aley vane rett sed diam voluptua. At vero cos et accusam et ju im mino dolores et ear erbum. Stet elita kasa guueyten, no sea takimata sanctus est Lorem ipsum dolor sit amet I orem ljeuth adlor sit amet consetettur adipschig elitt, sed diam nonumv utimod tempor invidunt ut labore et dolore magna aliquyam enut, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum Stet elita kasa gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitt, sed diam nonumv ermod tempor invidunt ut labore et dolore ces est accusam et justo duo dolores et ea rebum. Stel citta kasa gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem situ sont colores est accusam et justo duo dolores et ea rebum. Stel citta kasa gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. orem insum dolor sit amet conselettir sadioscido

#### LSGAN Loss

Lorem ipsum dolor sit amet, consetetur sadipscing elitt, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna eliguyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor sadipscing eitir, sed diatir nonumy eimod tempor invidunt ut läbore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et a rebum. Stet dita kasd gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eimod tempor sed diam voluptua. At vero eos et accusam et justo duo dolores et ea reburn. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. invidunt ut labore et dolore magna aliguyam erat,

Arial

Lorem insum dolor sit amet, consetetur sadieseir g elitir, sed diam norumy elimod tempor invidunt ut labore el dolors magna alquyam erat, sed diam volupiua. At vero eos el accusam er justo duo dolores et ea rebum. Stet clina kasa gubergran, re-sea takimata senctus est Loram ipsum dolor s't amet. Lorem ipsum dolor s'i amet consetetur sadipecing elitir, sed diam nonumy elimod temoor invidunt ul labore et dolore magna aliquyam erat, sed diam volupiua. At vero eos et accusam el justo duo dolores et ea rebum. Stet clina kasa gubergran, no sea takimata sanctus est Lorem insum dolor sit anet. Lorem ipsum dolor sit amet, consetetur sadipscing elitir, sed diam nonumy elimod tempor invidunt ul labore et dolore magna aliquyam erat, sed diam volupiua. At vero eos et accusam el justo duo dolores et ea rebum. Stet clina kasa gubergran, ned am volupiua. At vero eos et accusam et lusto Lorem ipsum dolor sit amet, consetetur sadipscing duo dolores et ea reburn. Stet clita kasd gub no sea tattimata sanctus est Lorem ipsum dolor sit amet.

#### لَّ من أن تَتلقى بَعض الأَلفَاظ من اللَّغاتِ المُجاورَة الت

نَاتِي بِأَمْثِلة كَثِيرة على مَزايا اللُّغة العَربيَّةِ في سَلا الاشْتِقَاق وَطُواعيَةِ التَّعبيرِ وَدِقَّة الدَّلالَة وَمُرونَةِ ٱلنَّركيْ وَكتابُنا لهٰذا لا نُريدُه كتَابَ لُغة، وَنَخْشى أَن يُؤدِّي لعَربيَّة، وليس الآمر كذلك، فإنَّ في تَعلُّمُ اللُّغات كُلُّ حتَجِبُ وتَكاد تَخْتفي إذا كانَتِ اللُّغة حَيَّةً يتَحدَّثُ ناً، ولا يُعلِّمونَها تَعليماً فَتَحْسَب. وَالمَصاعَبُ التي تَع مَرِكُها كَلِهَا إلى المَرْحَلَة التَّارِيافية التي يَهْجَنَازِهَا ا تمام الاتقان.

#### جَمْهِرة النَّاس يَكْفيهِم مُجرَّدُ البِّيان للإعراب عن حاج

ヨソ最6覧マテツヌ高機ふ県連兄えぶる。全はず性世 をほるゅ人親廃り育73説ど暮女フ量投ビ初会ざどン企 付覧ラサノソ近単ヤモスナ備企りか権航 ケ況徳チヒ ヤ自個『しきン選諸須ば。円オメスキ握子ケエウ如果

10 4

orem ipsum dolor sit amet conseletur sadipscing Lorent ipsum oblor sit anter. Conseteutor sacipsong aftr, sad diam nonumv eirmod termor/invidunt ut labore et dolore m-upna aliguvam etrat sed diam voluptua. At vero cos et accusam et iguito euro diolores et ea rebum. Stet ellla kaad guoergren, no sea takimata sanctus est lorem ipsum dolor sit amet I orem ipsum dolor sit amet, conseteur sadipscing diftr, sed diam nontimy olimod tempor invident ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum. Stet ellit kaad gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing diftr, sed diam nonumy eimod tempor invident ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese est accusam et justo duo dolores et ea rebum. Stet elit kaad gubergren used diam voluptua. At vero ese est accusam et justo duo dolores et ea rebum. Stet elit kaad gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. elitr, sed diam nonumv eirmod tempor invidunt ut

#### WGAN Loss

Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos etaccusam et justo duo dolores et e rebum. Stat citta kasd gubergren, n sea takimata sanctus est Lorem ipsum dolor sit ren no amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor sadipscing eitir, sed dath nonuny eimod tempor invidunt ul talore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo dou odores et ea rebum. Stet cilta kasag gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliguyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea reburn. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet.

#### E E ursadiosurg orem insum dolor sit amet consete:

Lorem ibsum dolor sit amet, conseteur sad osu gʻ-ciut; sad diam norumy ermot tempor texi unit ur labor: ...dolore manna aliquiyam eret, sad diam volupua, At vero eor et accusam e-justo duo dolo es et a returm. Stet dina Kasu, gubergran, ser takimata senctu ast Loram (rsum dolor s' amu, Lorem losum dolor ai eme), consetutu sadipscina ellir, sed diam nonumy eringat temi or nuvaturu tu labore at dolore magna aliquiyam erat, sad diam volupius. At vero eos et accusam et justo duo dolores et ea: abum. 'Stet citta resid quibergran, no sea taliamata sanctus esi. Lorem (rsum dolor sit amet, Lorem losum dolor ai amet, connetteur sadipscina ellis, sea Casm noumy erima ti supor invifuntu ti labore et dolore magna aliquiyam erat, sed diam volumi<u>s. At vero eos et accusam</u> et justo duo dolors esi esi abum. Stet ciha e...si gubergran, no sea ta inata sanctus esi. Lorem (sum dolor sit amat. Lorem losum dolor sit amat. Lorem losum dolor sit amata tas anctus esi curen in pur dolor sit at ata sanctus esi curen in pur dolor sit at diam volumi<u>s. At vero eos esi at accusam</u> et justo no sea tai imata sanctus est Lorem ipsum dolor sit

#### لدَّ من أن تَتلقى بَعض الأَلفُاظ من اللُّغاتِ المُجاورَة الت

نَاتِي بِأَمْثِلَة كَثِيرة على مَزايا اللُّغة العَربيَّةِ في سَلا الاشتقاق وَطَواعيَةِ التَّعبيرِ وَدِقْهُ الدَّلالَة وَمُرونَةِ النَّركِيْ وَكتابُنا هٰذا لا نُرِيدُه كتَابَ لُغة، وَنَخْشى أَن يُؤَدِّي العَربيَّة، وليس الأمر كذلك، فإنَّ في تَعلُّم اللُّغات كُلُّ حتَجبُ وتَكاد تَخْتفي إذا كانَتِ اللُّغة حَيَّةً يتَحدَّثُ ناً، ولا يُعَلِّمونَها تَعليماً فَحسب. وَالمَصاعبُ التي تَع مَردُها كلُّها إلى المَرحلة التَّاريخية التي يَجْتَازها ا با تَمام الأثقان.

مْهِرَةِ النَّاسِ يَكْفيهم مُجرَّدُ البَيَانِ للإعرابِ عن حا-

نغهرة النَّاب يَكْنَهُم مُجَرُدُ النَّان للإمرابِ عن حا عنهرة النَّاب يَكْنَهُم مُجَرُدُ النَّان للإمرابِ عن حا عارة وقتي المحالية الم

# 



# • CycleGAN with ResNet:

Input	MinMax Loss	LSGAN Loss	WGAN Loss
Anna Lorem ipseum dolor sit amet, consetetur sadipacing elitr, sed "an morumy eimod tempor invidunt ut latore et olore magna alguyam erat, sed diam voluptua. At vero ese se accusam el justo duo dolore set ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem lingum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipacing elitr, sed diam, nonumy elimod tempor invidunt ut labore et dolore magna alguyam erat, sed diam voluptua. At vero ese taccusam el justo duo dolores et ea rebum. Stet cita kas gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit sinel: conseleur sadipacing elitr, ad diam fonum volmod tempor invidunt ut labore et dolore magna si aquyam erat, sed diam voluptua. At vero ese a haccusam et upso duo dolores et ea rebum. Site una kasd gubergren, no sea takimata sanctus es corem insum dolor sit amet.	Anai Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed u um nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero cos et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod dempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eo es et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet.	Lorem ipsum dolor sit amet, consetetur sadipscing elifr, sed m nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. A tvere ocs etaccusam et ljusto duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elift, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eo set accusam et ljusto duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elift, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptu. At vero eo set accusam et ljusto duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet.	
Anel Loram ipsum doior sit amet, conseteur sadipsur a linr, sad siam norum, eimod tempor invidini ti adore et doior megna aliquyam tent, sad dia tabore et doior megna aliquyam tent, sad dia sadipscipa eitr sad diam nonumy elmod tempor mydani ut labore et doior magna bliquyam tent, sad diam voluptus. At vero ess et accusam et justo dio doiors et acharum ster tilta hasd ubergran, na sea takimata sanctus est. Loram ipsum doior sil anel. Loram ipsum doior sil amet, conneletur aadipsing eitr, sad diam nonumy elmod tempor mydan tilta sad tabore et doior magna aliquyam end, sed diam voluptus. At vero ess et accusam et justo dio doiors et al doior magna aliquyam end, sed diam voluptus. At vero ess et accusam et lusto dio doiors et al doior anga aliquyam end, sed diam voluptus. At vero ess et accusam et lusto dio doiors et al doior anga aliquyam ender sed diam voluptus. At vero ess et accusam et lusto dio doiors et al doior anga aliquyam et al.	Arial E E E E E E E E E E E E E E E E E E E	Arial Loram losum dolor sit amet, conseteur sadipsol 9 rillr, sad diam nonum einmo tempor invi tunt ut tabore . dolore manna alugyam erat, sed diam volupua, At vero eo. et accusam ei justo duo dolore set a metum. Site tilln kas ujubergrein, sen takimata senctus est torem ipsum dolor et amet. Loren ipsum doio al ame. consetetu sadipsona elitr. sed ajam -onumy ein.od temu ar iuviouru til tabore en doner magna faliguyam erat, sed tam volupua. At vero eos et accusam et justo duo dolores et a. zhum. Tate citta eras dubergrein, no sea takimata sanctus est Lore nipsum dolor sit amet. Loren ibsum dolor sit amet, consetetur sadipsona elitr, asa, dam nonumy eim: d tempor inviduru til babore sit doner nipsum dolor sit amet. Loren ibsum dolor sit amet, consetetur sadipsona elitr. asa, dam nonumy eim: d tempor inviduru til <u>abore sit doner magna faliguyam</u> erat. sed diam voluma At vero eos et accus m et justo duo dolares et au rebum. Site citia kr.sd gubergrein, no sea takimata sencius est Loren ipsum dolor sit amet.	
لا من أن تتلقى بُعض الألفاظ من اللغات المجاورة ال تاتي بأنطة تحيرة على توايا اللغة القريقة في تلا الاغتيفاق وطراعة القيبر وقلًا الذلائة وهرونة التركيم وتعاقلا غذا لا تريد تعال لغة، وتخش أن يُوتي القريقة، وليس الأمر كذلك، فإن في تمام اللغات كل متحمة رتكاة مُختفي إذا كانت اللغة عنة يتمدّنك مرغما كلها إلى المرحلة القاريخة التي تجاوها 1 تمام الأتان. منهوة النبي تكتيم مُجردُ البان للإعراب عن حام منهوة النبي تكتيم مُجردُ البان للإعراب عن حام	لاً من أن تتلقى بعض الألفاظ من اللغات المُجاورة الت تَاتِي مَا نَظْلَة عثيرة على مَزايا اللَّغة المَربَّة في سَلَا الاَضْقاق وَطَرامة التَّبِير ودقَّة اللَّلالَة وَمُروبَة التَّركِيرِ وتتائما لمذا لا تُريدُه تتات لَقَة، وَتَخْش أَنْ يُوَدُّي العَربَة، وتكاة تُتَخْشي إذا كانت اللَّفة حَمَّة يتحدَّث بِ عا، ولا يُملُونها تعلماً فحس. وَالمَصاعت التي تع ل مرهُما كلها إلى الترحلة التَّاريخة التي يَابَتاما التي الم ل تمام الإثنان. حضورة اللَّاس يتَغْبِهم مُجَرَّة المَان للإعراب عن حاح	لاً من أن تتلقى بَعض الألفاظ من اللّذاتِ السُجاورَة التَّ تَآتِي بِأَنشلة تَغَيرة على مَرَايا اللَّذَة المَربَّةِ في سَلا الاضِقاق رَطُواعياً التَّبير رَوفَة الدَّلاَة رَمُروَة الدَّلاَن وَتَعَايُها هٰذا لا بُريدُه تنابَ لَذَه ، وَنَخْض أَن يُودُي العَربَيَّة، ولكاذ تَتَّفَضي إذا تانَ اللَّذَة حَبَّ يَتَحَدُّ بِ عَا، ولا يُعْلمونَها تَعَليماً مَحسب. وَالمَصاعبُ التي تَعَ بَنُهمواة النَّاس يَتَفْيهم مُحَرَّة البَان للإعرابِ عن حاج جَمهواة النَّاس يَتَفْيهم مُحَرَّة البَان للإعرابِ عن حاج	
ックのシーンスになった。 などのシーンスになった。 などの人気があった。 なた。 などの人気があった。 なりた。 な	コン成62マ <sup></sup>	コンルの空い、「おおおよいななない」とない コンルの空い、高橋な県並見えるため、会社す性世 をは合わくんの限分育の設と募って豊快ない会さといな 新利すりをは高中ませこちは注意インにおいた。生ぼ社会 よこのために高級長が不営力」ら、ほぼ見等ケッジ注 ヤモスト悩む」が「絵柄・への超手といき相野」をご達 野領法に用け、マモ運行入口。「おいて男やーンジョ 特領法に見出、たいの短手という相談・ない 当時復法に用け、マモ運行人にない、「おいた」の人の男・一い 「高橋ふ乃達見」が記録して、悠年れロンパの豊い一い 「おいとなな」」という、し、「お」に登した」の人の思 「おい」とない、「お」」とない、「おい」」とない 電気からしていき」という、「お」、「おい」」とない 電気からしていき」という、「お」、「お」、「お」、「お」、「 「お」、「」」、「」」、「」」、「」」、「」」、「」」、「」」、「」」、「」」 「」、」、」、「」、「」、「」、「」、「」、」、」、」、	
Arrai Locative and a second	Ariai Lorent ipsum doior sir amer, conseiletür sadijäschtg elitir, sedrälam, nonumy eigmod tergorolinvidumi ut- labore el droforsmogrue obreharen nätt sed diator voluptua. At vero c.c. sit accuaam ét justo o doiores et ea retutin. Siet elita kasd glubergien, no sea takimata sanctus est Lorent ipsum doior sit amet I oren ipsuh aloir sil amet, consetetur sodipschog tuft, Sod attranonfing dirhoft dithops ad glubergiet, Sod attranonfing dirhoft dithops du olores et ea rebum. Stel rälik kasd glubergren no sea takimata sanctus est Lorent ipsum doior sit amet Lorent ipsuh olor sil amet, consetetur sadipschog elit, sed diam nonumy dirmo dior sit adu olores et ea rebum. Stel rälik kasd glubergren no sea takimata sanctus est Lorent aliguyam erati sed diam volptus Al vero eso est accusam ef justo du olores et ea rebum. Stel cilita kasd glubergren, no sea takimata sanctus est Lorent psum doior sit amet	Angl Lorem ipsum doior sirthine, conserveur sadipschig elitr, sadidam conum, enmod lengocihviduni ut- lahore al doine visito elitra kaso glastem, no sea takimas sancius set lorem ipsum dolor si amel Lorem ipsim dolor sil amet conserveur sadipschig citr, dod diaranofim dimbig sadipschig citr, sad diara nonumy emod dempor invidunt ut labore et dolore magna aliquyam eral, sad dian voluptus. At veroese set cocusan et justo duo dolores et ea rebum. Stat citla hasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet	

Input	MinMax Loss	LSGAN Loss	WGAN Loss
Anal Lorem issum dolor sit amet, consetetur sadipscing elit, sed camonouny eimod tempor invidunt ut labore et colore magna aliquyam erat, sed diam voluptua. At vero ese seacousam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elit, sed diam, nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accugan et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit anet, consetetur sadipscing elit, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese it ancugan et justo duo dolores et ea rebum. Stat cita kasd gubergren, no sea takimata sanctus est. Jorem ipsum dolor sit amet.	Anal Lorem ipsum dolor sit amet, consetetur sadipscing eilitr, sed	Anal Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed u um nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ces etaccusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasa gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod met post invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet litta kasa gubergren, no sea takimata sanctus e ji kuorem ipsum dolor sit amet.	n n n n n n n
Anal <b>E E</b> Lorem ipsum dolor sit amet, consetetur sadipsurg nitr, sad diam nonumy sirmod tempor invidunt ut labore et dolore magna aliquyam erat, sad dolores et ae rebum. Stet dira kasa gubergen, no sab takimata sanctus est Lorem ipsum dolor sit and Lorem ipsum dolor sit amet, consetutu sadpscing elitr, sad diam nonumy elimod tempor invidunt ut labore et dolore magna niliqui en erat, sed diam voluptus. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasa gubergen, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadpscing elitr, sed diam nonumy elimod tempor nividunt ut labore et dolore magna alleuy am erat, sed diam voluptus. At vero eos et accusam et justo duo dolores de arebum. Stet citta kosa gubergen, no sea takimati, sed diam noumy elimod tempor nividunt ut labore et dolore magna alleuy am erat, sed diam voluptus. At vero eos et accusationes tusto duo dolores de site set torem ipsum dolor sit amet.	I in ribuling elimination in the intervention of the interventi	Lorem insum dolor sit amet i consettet in saciosul ( and tigan nonumy etimo tit tempor instrum tu oor i dulor manna situyuam erat, sad diam bubuga, At vere eo et abusam erat, sad diam bubuga, At vere eo et abusam erat, sado duo nolo es et ar erbum. Site turk kas unbergran, sen takinata senctus est toram issum dolor e amo, colem tusam dolar si amet, consettetu saduscano ettit, seg aram ronumy en uod temi au abunt lu tabo, era dout, magno niticier mi erat, sadi sam vinuon. At veru eos et accusam ef lusto duo aclores et ea' abun piec citat ream dolar si amet, Lorem iosim noolar sit amet, connettur sadi sam vinuo. At veru eos et accusam ef lusto duo aclores et ea' abun piec citat ream dolar si ameti, Lorem iosim noolar sit ameti, connettur sadi sita est, ae cam nonum y emo felupor mindium tul jabore et olore magna allouyam erat sed diam vinuo. Tetum cos sit accus, at pisto duo dol i rebum. Site citier in a uberni no sea tar, inata sanctus est corem torum doim sit aniet.	a affair i a annuar a a annuar an anna an anna an anna an anna an anna an an
لا من أن تتلقى بَعض الألفاظ من الألعات المُجاورة الذ تاتي بأنثلة كثيرة على مترابا اللغة التربية في سلا الاشتقاق وطواعية الشير ووقة الدلالة وشوئة التركيم وتعادي هذا لا تربية يتاب لغة، وتخشى أن يُوقتي العربية، ولين الأمر كذلك، فإن في تملم اللغات كُل حجب وتكاد تخلفي إذا كانت اللغة حجة يتحدث ا المرابية عليه المانية فحسب والتصاعب التي ته با شمام الإتفاق. حضورة الناس يكفيهم مُجرد الليان للإعراب عن حام	لاً من أن تتلقى بعض الألفاظ من اللغات المُجاورة الت تَاتِي بِأَمْطَة تَعْيَرة على مَرَايا اللَّنَّة المَربَيَّةِ في سَلا الاشتقاق وَخُواعِة التَّعبير وَدَقَّة اللَّالَة وَمُروبَة الطَّرِكِيج وتتاكا الا لا لا رُينُه كتات لغة، وَتَخْشى أن يُوهُي المَربَيَّة، ويس الأمر كذلك، فإن في نَعْلم اللغات كُلُّ م حجب وتكاة تُقْتَضي إذا كانَتِ اللَّنَة حَيَّة يَتَحَدَّ عنه الأمريقي، تمليماً نَحسب وَالمَعاعبَ التي يَتَحادُها ال عنه الاتقان. ي دماء كلها والى المَرحلة التَّاريخية التي يَتَجادُها ال عَنها الاتقان.	لا من أن تتلقى بَعض الألفاط من اللّذاتِ المُجاورَة الت تَاتِي بِأَنْدَلَة تَعْيَرَة على مَرَايا اللَّهُ المَربَيِّة في سَلا الاشْتِقاق رَعْلَ إِنَّهُ التَّبِيرِ وَيَّة الدَّلاَة رَمُرونَة التَّركي وتتابا المالا لا بُريدُه كتاب لَعْه وَتَحْمَى أَن يَوَدَيَّه المَربَيةِ وَقِبْسَ الأَمِر حَلْكَ مَانَ في تَشْم اللَّفات كُلُّ عُصَبُ وَكَاه تَتَعْمَى إذا كانَ اللَّه حَبَّة يَتحَدُّ المَربَع لَكُون المَنْعَان الحَمَد اللَّه عَبَّة يَتحَدُّ المَربَع اللَّه عَلَيْنَ أَحْمَى اللَّه عَلَيْه عَلَيْنَ عَلَيْه مَنْ المَربَع اللَّه عَلَيْنَ عَمَانِها لَحَدًا اللَّه عَبَّة يَتَحَدُّ المَان عَلَيْنَ عَلَيْنَ المَانِي اللَّه عَلَيْنَ عَمَانِها عَلَيْنَ عَلَيْ عَلَيْه اللَّه عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْ عَلَيْنَ عَلَيْنَ عَلَيْنَ اللَّه عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ اللَّه عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ المَانِ اللَّهِ اللَّهِ اللَّهِ عَلَيْنَ عَلَيْنَ عَمَانِي اللَّهِ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْ عَلَيْنَ عَلَيْنَ عَلَيْنَ اللَّه عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْ عَلَيْنَ عَرَبُونَ عَلَيْنَ عَتَعَانَ عَامَانِي اللَّه عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ اللَّانِ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ الْعَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ الْعَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَ عَلَيْنَا عَلَيْنَا عَلَيْنَ اللَّهِ عَلَيْنَ عَلَيْ عَلَيْنَ عَلَيْنَ وَعَلَيْنَا عَلَيْنَ ع	
コン振め置マテツス高橋為県進党えなのの人体性性 を注る少人気限的育っ親と奮えて着发される。会はず性世 を注る少人気限的育っ親と奮えて着发される。されず性世 たこの大きないたち、したいため、たいため、 生きな大きないため、したいため、たいため、 非常になったから、「「「「「「「」」」」」」 やマスナ幅全いたき線、シスト等した」「「「」」」」」 やマスナ幅全いたき線、シスト等した」「「」」」」」 やマスナ幅全いたき線、シスト等した」「「」」」」」 やマスナ幅全いため、「」」」」」 第3(は、「」」」」」」」」」」」」 第3(は、「」」」」」」」」」」」」」」」」」」」 第3(は、「」」」」」」」」」」」」」」」」」」」」」 第3(は、「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」 第3(は、「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」 第3(は、「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」 第3(は、「」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」」	コン長6.2 ペーン、高にない、いたなしたい、します。 コン長6.2 ペーン、高になら振空元となっ、計量で低生 と良くやくへいの約 育めになっ、計量では公式となっ、計量で低生 よこ6.2 からもいんがごデーニから生活とし よこ6.2 からもいんがごデーニから生活とい たこ6.2 からいしんでいた。 いたないたい、このになったが、このにないたい。 当今先進たべいたい、このに、このに、この人の見かい マ高騰なび進行にかけ、このに、この人の見かい マ高騰な不見ない。 ご知っ、このことにないたい、この人の見かい マ高騰な不見ない。 このしたない、 このため、この、このにはたいたい。 ないたい、 ない、 ないたい、 ない、 ない、 ない、 ない、 ないたい、 ない、 ない、 ない、 ないたい、 ない、 ない、 ない、 ない、 ない、 ないたい、 ない、 ない、 ない、 ない、 ない、 ない、 ない、 ない、 ない、 な	1000004130、10000000000000000000000000000	
And the second development of the second dev	Amai to standolousi ameté visaditskung ellit is elekenkondum sutai ellitak is standolousi ameté visaditskung ellit is elekenkondum sutai set del talvis elekenkondum sutai set del dolou realisti set ditakas i is standolousi ameté ameté lo site oblatakas i is set del ameté lo site oblatakas i is set del metansam sutai set autoria del metansam sutai del ameté conseletar metansam sutai set autoria del metansam sutai del ameté conseletar metansam sutai del ameté conseletar metansa del se rebum Stat obla kaset duberrare metansam sutai del set del ameté conseletar ameté visad dam nonum imod tempor lovidint tal labole - chore marana allonivam erat set dam voluto del set or sutai dous est ameté toria del set or sutai dous sit ameté conseletar sutai del set or sutai dous sit ameté del set or sutai dous sit ameté del set set or sutai dous sit set del set	Anal Lorentification doin "Information and an additional Lorentification doin" Information and a second control of the second and the second and the control of the second and the second and the second and the second and the second and the due dolores are the second and the second and the due dolores are the second and the second and the due dolores are the second and the second and the due dolores are the second and the second and the due dolores are the second and the due dolores are the second and the second and the due dolores are the second and the second and the due the second and the second and the second and the due the second and the second and the second and the due the second and the second and the second and the second and the due the second and the second and the second and the second and the due the second and the second and the second and the second and the due the second and the sec	

# **Test Case 4 Results**

• Pix2Pix with ResNet:

# Input 「城社」 本ヤ貫8事 Internet amel, conseteur sadipscing nonum veineet empor invidunt ut er nagna aliguyan erat, sed dian ero eos et accuisam et justo duo, rebum. Stet cital kasd gubergren ne sanctus eet Lorem ipeum dolor sit nsum dojo sit amet conseteur ta dian nonumy eimod tempor inter magna aliguyam erat inter magna aliguyam erat et n ipsum dolor elitr, sed diam labore et dolor oluptua. At ver oore et dolore magna aliquyam e iptua. At vero eos et accusam et it ea rebum. Stet clita kasd gube ata sanctus est∄ orem ipsun dolo 皆ざい チシサヒ基芸づるぱで午負35春げ注目

#### Without VGG Loss

だみじ会24知フばクと高中モサシヤ誇憲ホオマ博快オネユ庙 モテ東阪よ世俗ラムキリ築禁ムネ変滅ばび入権・1則薬はそし3 汉スキジ客引知っ作佐辺石と城にご表れるドか首国3択ミセモ浮 2週人ラベさぜ手依まほをル。

はご予思ヤコ腰能サ状入シ那級社シメ田更ぞげうび速日全ん より中立りや影にすめんション酸化シールを行うしません。 それり申立のドをそ寺高必ほれルイバ庁タヒカセ化社会んへ採開 も系鏡スげはて続主悪激ラ。9週ヌロヘマ毎時フやざき掲昭広 カキ開表ヲメイ差67源エスムナ支年など秋6南てものぼ断用文

島圃にル井場新密クオヤウ花重さゃじ所新げん相線ア選性サー 新聞しすぼや料置シーるむ発検92種多死品イ、世やちずづ融 だか振外夏オ評検んさ然人ユアズタ生雪みぶ理力支属とく型 皆指末にズ、道本スラカ阿京遺例せな情五切へマカ夫行ス使生 着最ミク入最明ふびト。

やへやび座月せ育愚ムヌリ連合ソナホヤ貫8事にた緊痛きおさ ホ夏87道が2掲携ト万汪国記4社ごク造二ち寄日うあや生理さ ヤクノ戦暑りす可急と、第12~間10部69地ム朝文ラ三座中ト 節松ヲ儀表ツムソ帯応サハチ熱選紙6ま千思ルヘクベ指食イ

# 館4紙ニ遺信がうずでけ分降きせこう道測しつう故境回街:

Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese at accusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam volputua. At vero ese at accusam et justo duo doloros t ca rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit aub doiords at ca rebuilt. Stet dilta kasa guborgren, no sea takimata sanctus est Lorem ijsum dolor sit amet. Lorem ijsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilta kasa gubergren, no sea takimata sanctus est Lorem ipsum dolor sit avot amet.

#### Aria

Anal Lorem ipsum dolor sit amet, consetetur sadipscing elifit, sed diam nonumy eirmogi tempor invidunt ut labore et dolore magina aliguyam erat. sed diam voloptua, At vero ces et accusam et justo duo dolores et ag rebum. Stet dila kasd gubergren. no sea takimate sanctur set torem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseteur sadipscing elifit; sed diam nonumy eirmogi empor invidunt ut labore et dolore magna aliquyam erat; sed diam voluptua. At vero es et accusam et justo duo dolores et ae rebum. Stet dila kasd gubergren, no sea takimate sanctur set torem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseteur adibacing elifit; sed diam nonumy eirmog empor invidunt ut labore et dolore magna aliquyam orat; sed diam voluptua. At vero ese et accusam et lusto duo dolores et ae rebum. Stet dila kasd gubergren, invidunt ut labore et dolore magna aliquyam orat; sed diam voluptua. At vero ese et accusam et lusto duo dolores et ae rebum. Stet dila kasd gubergren, no sea takimate sanclus est Lorem ipsum dolor sit amet.

祥台シセイリ民地雲るき提卓ユ昨提へ間切97革夜 くび最納み求回い東京ざさてお完逢ろ銅後モオサオ 延ほせ。

芸びゅ祭引り論欲くお光指失令80会オノレ 策無児 朝づクぎッ図制んりしッ注刊うおこぜ奈34佈胜勢防 鼻都つ経愛ね報場コケフヘハチみとがほ康抜爆籍書 り現ほぜでぎ所葉れ民楽ホ賀東トモヨ申村戻さ国

掲高目り望断タユロ義倒あづレ伸车6ーゼ海連コへ さ下転現な郵店ロ床軒9万ホセ査御裕相ふ。模スヤ 政治でぜたか炭掲いてぜド会60値ドいし論引個キ3 **毒皆ぎい。**球員勤百ぶぐ。代はてまづ税加ミチワサ を監ぎ地通ワ保報イぎつづ連事チャク団塩ソス人所

#### ピチシサヒ基芸づるぱで午負35春げ注図 ~市熱なリ点事ナミカ知査ハヤテチ血山ずに問平び

Arial Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut latore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr. sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren. no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren. no sea takimata sanctus est Lorem ipsum dolor sit amet.

#### With VGG Loss

だみじ会24知フば少と高中モサシヤ誇憲ホオマ博快オネユ モテ東取よ世毎ラムキソ業発ムネ変滅ばび入権す則要ほそし 又モタ客労利知べ作改活を就しご表17会ドか首国3択ミセモ将 に掲入ラべきぜ手故まほをル。

よご予思ヤコ瞭能や状入シ那級社シメ田更ぞげうび連日全ん に、シリーロッドきそ寺高ぶはれい代行タヒカセ能社交人へ探醒 は系鏡みげはで統主思感う。90,3ロハマや間野マやさ基礎の かり開表ヲタイ差67源エスムナ支年など秋6南てものぼ断用文

4価にル井場新密クオヤウ花重さやじ所新げん相機で煙性サー 紙間せすほや料置シーるむ発検92種の死品イ、世やちずづ融 が振発更才評検んさ発人ニアスタ生雪みぶ理力支属とご型 皆指表定ろ、道本スラカ阿京遺例せな情五切へマカ夫行ス使生 皆数ミク入量明ふびト。

やへやび座月せ育愚ムスリ連合ソナホヤ貫8事にた緊満きお ホ夏87週が7場携や5年国記4社ごク造二ち寄日ちかを生理さ ヤノの概念では可説と。新ノスー間10部69地ム朝文ラっ座中ト 畜松ヲ儀表ツムソ帯応サハナ熱選紙63年紀んヘクペ指食イ

6倍4紙ニ遺情否うずでけ分降きせこう道測しつう故境回術

I orem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus set Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr. sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, ed diam volutua. At veno ese et accusam et justo invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ces et accusam et justo duo dolores et ca rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. amet

Artai Lorem ipsum dolor sit amet, consetetur sadipscing elit, sed diam nonumy eirmog iempor invigunt ut labore et dolorer magna aliguyam erat. sed diam voluptua, At yero ese et accusam et juisto duo dolores et ea rebum. Stet dila kasd gubergren. no sea takimate sanctus est Lorem ipsum dolor sit armet. Lorem ipsum dolor sit amet, consetgur sed diam voluptua. At vero ese et accusam et juisto duo dolores et ea rebum. Stet dila kasd gubergren, no sea takimate sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetgur no sea takimate sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetgur sadiescing elit, sed diam noumy eirmod etmpor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et fusto duo dolores et ea rebum. Stet dila kasd gubergren, invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et fusto duo dolores et ea rebum. Stet dila kasd gubergren invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et fusto duo dolores et ea rebum. Stet dila kasd gubergren invidunt sanctus est Lorem ipsum doior sit amet.

Aria

洋台シセイリ民地雲るき提卓ユ昨提へ間切97革夜 くび最納み求図い東豆ざさスお究逢ろ銅後モオサ; 延ぼせ。

芸びゅ祭引り論欲くお光指失令80会オハレ策無男 朝づクぎッ図制んりしッド刊うおこゼ奈34億萬勢時 身都つ経受ね報場コケマヘハイみとがほ康抜歩籍第 月要ほぜでぎ所葉れ民業示賀東トモヨ申村戻さ国

粘高目り望断々ユロ義倒かづレ伸年5,一世語述ロッ さ下転現な難店口康軒57ポセ査御浴相ふ。模スヤ 政治で世行か視私いてせた会60値いい造計[個書] 整書27.0。現員の百ぷぐ。代はて主う役加ぶチワサ) 88監ぎ地通り保報イぎつづ連事チヤク団塩ソス人肉

#### アチシサヒ基芸づろぱで午自35春げ注目

~市熱なリ点事ナミカ知査ハヤテチ血山ずに問平ひ

Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing



#### Without VGG Loss

たみじよ24知フばクと高中モサシヤ誇憲ホオマ博快オネユ由 モテ東取よ世毎ラムキリ築禁ムネ変減ばび入権す則薬ほそし マモタ客討到マ作改冠モ域社ご表17会ドか首国3択ミセモ活 湯人ラベさぜF故まほをル。 はご予思ヤコ腰能サ状入シ那級社シメ田更ぞげうび速日全ん

そネリ伊立クドきそ寺高とほれルパ行序をしたを能社安んへ探覧 も系競スげはて続主悪激ラ。9凱ヌロヘマ毎時フやざき掲昭 がキ期表ヲメイ差67源エスムナ支年など秋6南でものぼ断用文

画ヒル井場新密クオヤウ花重さゃじ所新げん相裸テ煙性サ 4価ビル井場新密クオヤウ花重さゃじ所新けん相称アスロヒン 報用世すほや科置シーるむ発検92階の完品イ、世やちずづ醸 ひか振外東非務長んさ発んエアスタ生当るみ変加力支展と行 情先ぼう。道ネスラカ阿京遺例せな情五切へマカ夫行ス使生 客級ミク入量明ふびト。

やへやび座月せ育愚ムスリ連合ソナホヤ貫8事にた緊濃さおう ホ夏89遭が9掲携ト月3目間に4社ごク造ニち寄日うあや日理さ イクノ戦暑りは可能と、第12~間10部69地ム朝文ラコ座中ト 第松ヲ儀表ツムン帯応サハチ熱選紙らま千思ルヘクへ指食イ

#### 低4紙二許法不うずつけ分除きせこり道測1 つう故境同街 Arial

Anai Loram ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum. Stet cilta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr. sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, ed diam volubula. At vero ese et accusam et justo duo doloros et ca rebum. Stet cilta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum. Stet cilta kasd gubergren, no sea takimata sanctus est. Lorem ipsum dolor sit amet. Lorem labore et dolore magna aliquyam erat, sed diam voluptua. At vero ese et accusam et justo duo dolores et ea rebum. Stet cilta kasd gubergren, no sea takimata sanctus est. Lorem ipsum dolor sit no sea takimata sanctus est Lorem ipsum dolor sit armet.

#### Arial

Arial Lorem losum dolor sit amet, consetetur sadjoscing elitir, sed diam nonurny eimod tempor invidunt ut labore el dolore malan aliquyam erat. Sed diam voluptus, At vero eos el accusam el lusto duo dolores el se apolum. Stel dila kada gibergren nuo sea taltimata sanctus est Lorem ipsum dolor sit amet. Lorem lupsum dolor ali tamet, conseteur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptus. At vero ese et accusam el lupero duo dolores et ae rebum. Stel dila kada gubergren, no sea takimata sanctus est Lorem ipsum dolor sit sadiscing elitr. sed diam nonumy elimot tempor invidunt ut labore et dolore mogna aliquyam orat, sed diam voluptus. At vero ese et accusam et lusto duo dolores et a rebum. Stel dila kada gubergren, no sea takimata sanctus est Lorem gran dolor sit amet. Lorem voluptus. At vero kos et accusam et usto duo dolores et ai rebum. Stel dila kada gubergren, no sea takimata sanctus est Lorem psum dolor sit amet.

洋台シセイリ民地震スき提卓ユ昨提へ間切97革夜 人び最初な求国い東管させスお完達55個後モナサ 紙ほせ

まびゆ祭引り高統、お尤指失令80会すハレュ 亜甲 朝・5クぎッ図制んりレット刊うおこせ奈34他僅到可 身都つ経受れ報場コケフへパ子みとがほ球抜爆籍。 月契ほぜでぎ断葉れ民紫ホ賀東トモヨ申村戻さ国

掲品目り望断タユロ表倒かった。伸手5---ぜ面連コー さト転現な鉱店口接軒'ファ'ホセ査御宿相ふ。模スヤ 政治でごぜたか実指いてごせえる60航いいし面引開考当 **着書でい**。現じ勤行ふ。「ごにま 5.7歳のデチワサ」 88驚ぎ地通7保報 (ぎっつ速事チヤマ用塩ソス人声

モチシサヒ基芸づるぱで午負35春げ注図 市熱なリ点事ナミカ知査ハヤテチ血山ギに間平ひ Aria

orem insum dolor sit amet, consetetur sadinscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam labore et doiore magna anquyam erat, sed diam voluptia. At vero ece et accusam et juisto duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimate sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elltr. sed diam nonumy eimod tempor invidunt ut labore et dolore.magna aliquyam erat, Invidunt ut labore et dolore, magna aliquyam erat, sed diam voluptua. At vero eso et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren. no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet. consetetur sadipscing elitr, sed diam nonumy eimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet clita kasd gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet.

#### With VGG Loss

だみじょ24知フばりと高中モサシヤ誇憲ホオマ博快オネユ由 モテ東取ょ世毎ラムキリ築禁ムネ変滅ばび入権す則薬ほそし1 ヌモタ客討到マ作改冠モ域社ご表17会ドか首国3択ミセ 引人ラベさぜ手故まほをル。

ご予思ヤコ腰能サ状入シ那級社シメ田更ぞげうび速日全ん ネリ伊立クドきそ寺高必ほれル代庁タヒカセ能社安んへ探究 系鏡スげはで続主悪激ラ。9凱ヌロヘマ毎時フやざき掲昭が キ闘表ヲタイ差67源エスムナ支年など秋6南てものぼ断用文

iiヒル井場新密クオヤウ花重さゃじ所新げん相裸テ煙性サー 紙間せすほや料置シーるむ発検92個の死品イ。世やちずつ湯 か振外更オ評検んさ発人ユアヌタ生雪るみ家理カ支属どぐ? 情充ぼで、這ネスラカ阿京遺例せな情五切へマカ夫行ス使 :最ミク入量明ふびト。

やへやび座月せ育愚ムヌリ連合ソナホヤ貫8事にた緊張きおさ ホ夏87遭が2場推り汚国記4社ご2造こち客日ラあや生理さ ヤクノ暖書ッビロ第起と「新ノスニ間にの前68回ん胡文2つ座中ト 節松7歳表ツムソ帯応サハナ熱選紙らま千思んヘクベ指食 行

#### 協調に設持あう手では分除きせこり道測しつる故境回復 Aria

Anai L orem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr. sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, do diam voluptua. At vero eso et accusam et justo duo doloros ct ca robum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eso et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit ad diam voluptua. At vero eso et accusam et justo duo dolores et ea rebum. Stet citta kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit no sea takimata sanctus est Lorem ipsum dolor sit amet.

#### Arial

Artal Lorem lipsum dolor sit amet, consetel/ur siglpscing etiir, sed diam nonurny eimpot invidunt ur, labore ad ea engen. Stel diak sad gubergren. no sea takimate sanctus est Lorem lipsum dolor sit amet. Lorem lipsum dolor sit amet, consetsur sadipscing ellir, ed diam nonumy eimot tempor invidunt ut labore et dolore magna aliquyam erati, sed diam voluptus. At vero ese et accusam el lusto duo dolores et ae rebum. Stel diak sad gubergren, no sea takimate sanctus est Lorem lipsum dolor sit amet. Lorem, lipsum dolor sit amet, consetsur sed diam voluptus. At vero ese et accusam el lusto duo dolores et ae rebum. Stel diak sad gubergren, no sea takimate sanctus est Lorem ipsum dolor sit amet. Lorem jugun dolor sit amet, conseteur sadioscing elitr, sed diam nonumy dirmot tempor duo dolores et a rebum. Stel diak sad gubergren, no sea takimate sanctus est Lorem ipsum dolor sit amet.

洋台シャセイリ尿地震るき提卓ユ昨提へ間切ぬ7革夜 、ひ最納み水内い東貸ささスお完達ろ別後モオサ、 モモセ

芸びゆ祭引り贏欲くお光指失令80 点オハレ 東亜甲 朝-5クぎッ図制んりしッ注刊うおこせ奈34俺岱到厉 身都つ絶愛れ報場コケフへ八子みとがほ康抜嫌辞 月契ほぜでぎ断葉れ民紫ホ賀東トモヨ申村戻さ国

掲高目り望断シュロ義倒あづと伸守6一ぜ面近ー さト転現な郵店ロ球車やかホセ査御紹相ふ 模スヤ 政治でぜたか炭掲いてせド会60値ドいし論引個**キ 毒皆ぎい。**球員勤百ぶ、(代はてまづ税加ミチワサ ま8監ぎ地通ワ保報イきつづ連事チヤク団塩ソス人

ピチシッサヒ基芸づるぱで午負35春げ注図 市熱なリ点事ナミカ知査ハヤテチ血山ずに間平び

Lorem ipstm dolor sit amet, consetetur sadipsoing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum: Stet cilta kasa gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilta kasa gubergren. no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilta kasa gubergren. orem insum dolor sit amet, consetetur sadipscing duo dolores et ea rebum. Stet clita kasd gube raren no sea takimata sanctus est Lorem ipsum dolor sit amet.

# • CycleGAN with ResNet:



#### Without VGG Loss

これしよ24知77ほクとあり単モサンヤ務連ホオマ博快オネユ曲 テ東取に世俗ラムキリ発見ムキ変減はひ人権小国業ほそし」 ヌモダ客討知ッド後辺と転社ご美17会ドか首国3状ミセモ浮 易人ラべさぜ手故まほをル

まご予思ヤコ腰能サ状入シ 郎観社シメ田更ぞげうび速日全ん 

やへやひ座月せ右塁ムスリ連合ソナホヤ貫8年にた緊適さお ホ夏87遺がり指携トラぼ国乱4日ごク造ニち寄日ラルッキ理さ ヤクノ限帯シぼ可就と 新ノスニ問わ前80姓ム朝文ラコ№中ト 第松ヲ儀表ツムソ帯にサハチ熱道秋らま千思ルヘッハ指食イ

#### (仏)紙ニ消損ボラポでけ分除キサータ消測1 つめ故壇回街・ま Arial

Arial Lorem lipsum dolor sit amel, conseletur sadipscing elitr, sad diam nonumy-eimed tempor invidunt ut labors et dolore magna aliquiyam erat. sad diam voluptua At vero eos et accusam et lusto duo dolores et ea rebum. Stet cilta kad gubergren. no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr sed diam nonumy etimod tempor invidunt ut labor- ei dolore magna aliquiyam erat, tod diam voluptu. At vero eos et accusam et justo duo doloras et oa rebum. Stet cilta kad gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing eitur sed diam.nonumy etimod tempor invidunt ut labore et dolore magna aliquiyam erat, sed diam voluptua. At vero eos et accusam et-justo duo dolores et ea rebum. Stet cilta kaad gubergren, no sea takimata sanctus est-Lorem ipsum dolor sit umet.

#### rial

Anal Loren Ibsum dolor sit amet, consete/ur asdipscing elliv sed diam nonumv eimpa tenpot/Tuwdum ur lahone diolere angen allinvam nevi sen triami volupuia, At vero eos et docusarg el justo duo rotornes et a estabum. Sitel dia kasd gilbergen im e saa taktimata sanctus ast Loren ißsum digt, sit anet. Loren jusum dolor sit amet, conseteur sadipscing elliv, sed diam nonumy elmos tento novose taktimata sanctus est Loren igsum delor sit anditu volupuia. At vero eus et accusar el tusto novose taktimata sanctus est Loren igsum dolor sit anditu tabore et dolore ungan alleuyar efatt uo dolores et el enterum. Site dilta kasd gubergien, novose taktimata sanctus est Loren igsum dolor sit anditu tabore et dolore nogan alleuyan efatt sed diam volupuia. At vero eus et accusar et lusto sadinkolog bitr, ad diam nonumy elmog tampor midunt ut labore et dolore nogan alleuyam efatt sed diam volupuia. At vero eus et accusar et lusto duo dolores et dolore nogan alleuyam efatt and dia no volupuia. At vero eus et accusar et lusto duo dolores et dolore nogana gilbeurgiren, no soa takmata sanctus est Loren ibsum dolor sit anet

#### 洋台にすて伊屋地震るき提昇の時間へ間切り遊夜 くい最初へ来れい地行さて「おん」ド「加快モール」 絶任せ S

立、546祭研り成化、お光悟火音80六日、山東豊田 朝しひきへ図期、5日、小門うおこせっ346(低気度) 事都つ経転れ報場コケン、代子ユとがよば夜晩時 対衆はまて芝街葉れ民奈尔得取してヨリトロ、546

協高日町羽豚やこれを何あった 個事に一世命団に 時応目10名間がないなた時かう1、10%に一て初辺によ と下かれたたがたいな様年レアホーた後期常用した。換くイ 次行ででたっが見招いてした。260位にし、面引団本名 舞者ぎい。球員勁口、55、【たはてよっ税加えチャット 28監さ地通り保湿イミアー)近事チャット用塩パニ人自

ピチシサヒ基芸づるぱで午負35春げ注図 市熱な可点事ナミカ知査ハヤテチ血山ずに間平ひ

Lorem ipsüm dolor sit ämet, consetetur sadipscing eitir, sed diam nonumy eimod temporTinvidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum Stet citta kasd gubergrån, no sea takimata sanctus est Lorem ipsum dolor sit, amet. Lurem ipsum dölor sit akmat, consetetur sadipscing eitir, sed diam nontumy egizod tempor invidunt ut lauore et dolore, anggna aliquyam text, sed diam voluptua. At vero ese at accusam et justo duo dolores et ea rebum. Stet diafta-kasd gubergran. no sea takimata sanctus est Logenupsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing eitir, sed diam nonumy eimod tempor invidunt ut latore et dolore, anggna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit amet. orem ipsom dolor sit amet, consetetur sadipscing

# With VGG Loss キリ築禁ムネ変滅にび入権す則葉日 改活モ城社に表17会ドか首国3択ミセ

Elizat ショスリントし、アタヒカセ能社安んへ探 激ラ。9加スロヘマ毎時フやざき掲訳 スムナ支年など秋6南てものぼ断用 オヤウ花重さやに存在げん相様テ煙性サ へるむ発検92009死乱イ。世やちずつ ごた人ユアジクト当るハ家理力支属とく ラカ阿京遺例せな情五切へマカ夫行スペ

ムスリ連合ソナホヤ貫&事にた緊注 注目記4社こク造二ち寄日ラあゃ て、新ノスニ問10部69処ム朝文ラコ 、サハチ熱達秋らま千思んへクベ 甘奇勘人

Arial orem ipsom dolor eit amet, consisteur sadipscing allin; sed diam nonum-yeimed tempor invidint ut abore et dolore magna alliquiyam erat, sed diam voluptus. At vero ose et accisam et justo duo dolores et ea rebum. Stet clifa kasd gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseteur sadipscing alline di diam nonumy eirmot tempor invident ut bore chore magna alliquiyam erat, ded diam vero suntus est Lorem ipsum dolor sit amet. Lorem psum dolor sit amet, conseteur sadipscing alline di diam nonumy eirmot denor hudidint ut bore. Chorem psum dolor sit amet. Lorem psum dolor sit amet, conseteur sadipscing alline ad diam nonumy eirmot tempor mididint ut labore et dolore magna siliquiyam erat, ed diam voluputa. At vero ose al accusam et justo du dolores et se rebum. Stet clifa kasd gubergren, no san takimata sanctus est. Cerem ipsum dolor sit amet.

世雲マき提出エ申撮い間切の黄 小東ロジンズジン注う細胞モナ 1 all de **上**地 0 告述ゆ祭引り読む人が光指 肌 ベクギッ図制たりに少に内 用・クラー図開たりレクト的うおこせ会34位Walk |都つ経費な報場コケフーハイ子真とがほぼ抜爆制 |都の経費な報場コケフーハイ子真とがほぼ抜爆制

精高目9号開始2221茶間かった個年5一世間通 支持地理会開始に低低人生活時間的で、 取得て世界の現地にで低下去260位にした第月 **時間**でしたの現地にで低下去60位にした第月 時間の1000元 いと適引個キ 8監ぎ地通り保報イぎ、

#### ピチシサヒ基芸づるぱで午負35春げ注意 なリ点事ナミカ知査ハヤ

em (psym, dolor sit amet, consetteur, sadipson, sad diam-nouny armod temporitydaut ut pro et dolore magne, eliquyam erat, sad diam, iptua. At vero eos et arccusam el (usoto duo no est ea rebum). Ster dita kasid gubergio takimata sanctus est Lorem ipsym dolor sit, at Lorem ipsym dolor sit, at. Lorem ipsym dolor sit, at. Lorem ipsym dolor sit, and consetteur for piscing elit, sed diam nonlimy eligio demos duo turbus. At vero eos at sociosam el just dolores et ea rebum. Ster discusa dolores elitores at. Lorem ipsym bibm. Ster discusa di guo en esta data voluptua. At vero eos at sociosam el just dolores et ea rebum. Ster discusa dolores et ea at. Lorem ipsym dolor sitametto ensistento. to sed additional samindus ear Logels juditir duois an met. Lorein jugarin duois est ander tender tender adigezing elyr, sed diam nonumy airmod tempor, witcht (i) glober et doiore angena aliquyam erat. ed dian voluptua. At vero eos et accusan et just to doiors et a rebum. Stet cital kasat gubergren oset aktimata sanctus est Lorem ipsun doior sit oset



# **Test Case 5 Results**

• Pix2Pix with ResNet:

Input



#### Learning rate 0.0001

Lorem ipsum dolor sit amet, consetetur sadipscing elit, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero esc et accusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no sea takimata sanctus est Lorem ipsum dolor sit armet. Lorem ipsum dolor sit amet, consetetur sadipscing elit, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero ces et accusam sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam: erat sadipscing elitr, sed diam nonumy eirmod tempor invidunt tabore et dolore magna aliquyam: erat sed diam voluptus. At vero ces et accusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, no se takimato senctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletar sadiam voluptus. At vero ces et accusam et justo duo dolores et ea rebum. Stet cita kasd gubergren, duo dolores et ea rebum. Stet clita kasd gubergren, no sea takimata sanctus est Lorem Ipsum dolor sit

ow, and I don't much care; for it seems to m be a little damaged in the head. But if y of Captain Ahab, of that ship there, the Pequo I you, that I know all about the loss of his leg. bout it, eh-sure you do?-all?"

figger pointed and eye levelled at the Pequ keistranger stood a moment, as if in a troubled rting a little, turned and said :--- Ye've shipp rtipg a little, turned and said: --- ' xeve smpp ames down on the papers ? Well, well, what's i; and what's to be, will be; and then again, be, after all. Any how, it's all fixed and ; and some sailors or other must go with ; as well these as any other men, God p set as shimmates morning : the ineffable hear

to ve. shipmates, morning ; the ineffable hear 

ing its community and the reputation of being Now Sidad, I am sorry to asy, had the reputation of being an incorrigible oil hunks, and in his see-going days, a bitter, hard task-master. They told use in N-stutcket, though it er-tim Sachem's Back. A transguar openng ance constant we bess of the ship, so that the insider commanded a complete vice forward.

time Sackers's field. A transguint opening meet owner we how of the ships, so that the insider commande a complete view forward. And half concereded in this queer tenement, I at length found one who by his aspect seemed to have anthority i and who, it being noon, and the ship's work supported by the second respite from the barden of command. He was stated on add-fadiood oaks chair, wriging all over why of end earving i; and the bottem of which was formed of a stout interleading of the same elastic state of which the wignum was constructed. There was nothing so very particular, perhaps, about the appearance of the identy man I saw; he was born and pilot-folds, et al. I also do same, and heavity rolled as fine and almost microcopic net-work of the minutes this consimul saki-ling in many had gales, and heavity rolled us fine and almost microcopic net-work of the minutes the communal for this causes the muscles also the grow to history and other do state and the syster to become pured together. Suck even with same advers from historic state and the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I, advassing to the door of the the Captain of the Pequed 7" said I advassing to the door of the the Captain of the Pequed 7" said I advassing to

。渡渡おけどく取属ん立望うせ安藤は念辞来ニフナイ走芸 そりサ連携キソミサ掲載ネフレ後期トざ了位ニホ夫8監ぎ ノス人所性比賞ひィ

コタト記子シサヒ塩芸つるはてから0.66に行1回紙いつゅけ近年5月開発や センウ車花ハヤテチ血山中に割平形さして任座舎待有え、展ユ电21点89/ 対応ムフマ活か約んにすり込地37を発始水・中ド振っ割助997枚収切でませ 街ちふダ中をくひん病がみはり、嗅ささ3天が注つ30歳でオサホ度以子

2時前街へし宣芸の毎祭引り編成くした指矢も80会オハレ 宣悪り毀し 記録1計時の訪問づタを 河口構入力しかに通うねこぜ完全4使催倒損化も 値た付割で得進っを受え事場よりマッヘ人 チムンはな能技能得らけ ツッネ油 かっ汁気は老くさき地震れた洗水気和1とも甲件状気に超ぶ 報一,報海5 直た付為アウ茶 ノワ素描 ウハ 新優疑パイゆこ

などのジャードロウ **5作クロク上祝テンサヒ基芸づるばで午負85巻げ注区駄いシッ チ胆兆やこ**けん市熱化り点中アビタ切面をハヤテチ魚山片に間座びるしで任味 に興丸れ高い相当をコダル開ナウマ特輪メフィマ青洋行さとセイリ氏地震スを提ぶ可能 きばつぼし育2権まぎなン残育こ公対ぼ一給密みい舎容立ト広房ノタモ楽47港ユレ

#### Learning rate 0.0002

Lorem ipsum dolor sit amet, consetetur sadipscing elltr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Siet cifita kasi gubergren, no sea takimata sanctus set Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur adicesien eite ad diam proury utered tempor. amet. Lorem ipsum dolor sit amet, consetetur sadipacing elitr, sad diam nonumy eirmod tempor invidunt ul labore et dolore magna aliquyam erat, sed diam voluptua. Al vero cos et accusam ef justo duo dolores et aa rebum. Stet citta kasd gubergren, no soa takimata sanctus est. Lorcm ipsum dolor sit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ul labore et dolore magna aliquyam erat, sed diam volubus. Al vero cos et accusam et justo sed diam voluptua. At vero eos et acc am et juista duo dolores el ea rebum. Stét clita kasd gubergren, no sea takimata sanctus est Lorem Ipsum dolor sit

ow, and I don't much care; for it seems to n end," said 1, be a little damaged in the head. But if y of Captain Ahab, of that ship there, the Pequo you, that I know all about the loss of his leg. bout it, eh-sure you do ?-all ?"

finger pointed and eye levelled at the Pequ keistranger stood a momont, as if in a troubled rting a little, turned and said :-- "Ye've shipp rung a little, turned and said :--- 's ever shoppy andes down on the papers ? Well, well, what's i; and what's to be, will be; and then again, be, after all. Any how, it's all fixed and i; and some sailors or other must go with the as well these on other men. God re ; as well these as any other men, God p to ve. shipmates, morning; the ineffable hea

Petring from factor one as the good period sector only in the sector of the sector of

ロタ上祝チシサヒ基芸づるはで作負85 体げ注図駄いッルば度材け肥兆。 テビカ知道ハヤモチ血山ずに関乎びざしで任狭寄籠寿え、奥ユ典21-18 ワ特績メフィ者からなセイ氏領定なな指法ユ哔提へ開始97者専切で注 にはりし回うひさして10株常 2七イリ民地に入さ構成ユ昨提 のみまわり、近いささスセンジョ ワ特婦メフィッ教神舎に 間間ちふ安年をくひぬう 潜むに禁東利絶好せ

マス前前人し近景の市第四の発展くお充当先台かの会オーレ発展リ服用振振技 (現在21時代の海側つぎ)の対抗(あした注張)はご常常な(健性別的)を使作別) 電助だけ(部)で身際の)を設えは対当コアへ入トイルらは能技能発きげ 福一 次/7時間 カイリ双口 ぜてごが取れたSは下をお中村以注国ぶ、採高り 夕陽便能・くイゆこ

5作クロタ上況チシサと基芸づるばで午負35春げ注図駄いッ ア胆兆やこけ 上ばつぼし育2権まぎなン残育こ公対ぼ一給密みい舎容立ト広房ノタモ案47港ユ

#### Input



#### Learning rate 0.0001

Lorem ipsum dolor sit amet, consetetúr sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilia kasd gubergren, no sea takimata sanctus est Lorem ipsum dolorsit amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilia kasd gubergren no soa takimata sanctus est Lorem ipsum dolor sit amet. Lorem ipsum dolor sit amet, conseletur sadipscing elitr, sed diam nonumy eirmod tempor invidunt ut labore et dolore magna aliauyum e am saupsang eitit, sed diam nonumy eimod temoo-invidunt ut labore et.dolore magna aliguyam e a-sod diam voluptus. At vero eos et a zcucam o, tho-duo dolores et eu rebum. Ştiet clita kasa gubergren no sea takimata sanctus est Lorem ipsum dolor sit amet.

y, and I don't much care; for it seems to n end," said 1, Captain Ahab, of that ship there, the Pequo ou, that I know all about the loss of his leg. pout it, en sure you do?- all ?"

finger pointed and eye levelled at the Pequ terstranger stood a moment, as if in a troubled ting a little, turned and said - "Ye've shipp mes down on the papers ! Well, well, what's and what's to be, will be; and then again after all. Any how, it's all fixed and and some sailors or other must go with as well these as any other men, ve. shipmates, morning; the ineffable hear

<u>A vec shipmates, morning; the ineffature near a vec shipmates, morning; the ineffature near a vector of the second se</u>

income. Now Bildad, I am sorry to say, had the reputation of being an incorrigible old humas, and in his seargoing days, a bitter, hard task-master. They stall use in Newszlett, though it see time Sachers Beach. A transguare opening incee waves use how of the ship, so that the insider commanded a complete into focus

interlacing of the same elastic stuff of which the wigeram was constructed. There was robing to every particular, perhaps, about the appearance of the sliderly must have in or was bown and prior both, out in the source of the slider of the slider prior both, out in the slider of the minutest writeful up in blue and histogravity hist start of the minutest writeful in the prior both and the slider of the minutest writeful interlacing round histogravity which must have starts from his continual sali-ing the start of the slider of the slider of the slider of the slider of the start of the slider of the slider of the slider of the slider transfer of the slider of the slight slider of the slider of

the do. "Su e tent. g it be the Captain of the Pequod, what dost thou "Supposing is a way want of him ?" he domanded.

ん、「早)せ交話は、arii 二二(北京省平床村勤百 サ掲載率フレルホト。子振一米太8話き地通り保護子

「小小川花花花(小)」 「マント夏でも、中午になっています。「白露水石」に回知に、シムになりませて開催でき。 中午からかが、シートで加加していたりでいた」「石口です」が、「一」「「中午」の 「小小口」」の「「一」」で「日本」で見加していた」、「日本」 マイサロボムシャチ生化中ドロドアシューアはロマン・104に及び10年に フトローアス会会である人では人生進まる結果に研想と加て1097年代以下で 国际バルマナル、したに対した進まる結果に研想と加て1097年代以下です。 (ウストード4代12月3月)

の面面のないにたいかっていっていたれたこれの、イベレーの形型構成地から またようから教育になな、みる話がれたいにはなまた、世界など開始可た40代表引 値にしたってで作者、5条のよりと思ったで、人くならな様様体が行きし、他ー 「あっ」、マット」の目をし、この美人について発起でも単作技を見られ、示範の DE STATE

ム、認定は中位活為目的増加なから使用の等地準に一定活用の一九谷(広 の用人になっる時時間な風味の時間ですませた時期目は、単々な大学時間 見ませたサキ、彼然、可たった中、一世」2000年によっ引用産業現合い 島づいば「環境業賃貸」よい認知に、「はては、月度等や中産業件やな 素によい。こんなに実地的で発行で、の準定化で1月14日、人名日日に行 同月個キ変犯ざい止

5作クロタ上況チシッサヒ基芸 つるばで午負55春げ注図駄いッ

T担実を二日本市急なリた中ナキガ加会ーーー・美田市営業のSL 車項11点の用サシロ採用ナウフ持届メラマ青年行ち七イ川切地実えき構 次切では世がた政・人々開設な医院になったなどの統領本実体、現任さ 聞る のホル反ニ(m79 & アンチル 説信ひと集張村総(Fu)、聞る国意メケ文持り分野 つぼし育2権まぎなン残育こ公対ほー給密みい含容立F広房ノタモ条47港ユ

#### Learning rate 0.0002

Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy elimod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero eos et accusam et justo duo dolores et ea rebum. Stet cilta kasi gubergren. n sea takimata sanctus est Lorem ipsum dolor.sit n no amet. Lorem ipsum dolor sit amet, consetetur sadipscing elitr, sed diam nonumy eirmod tempor sadipacing eith, sed dahn nortunny einmod tempor invidunt ul tabore el dolore magna aliauvam erat, sed diam voluptua. At vero eos et accusam et justo dou odores et ea rebum. Stet alita kasd gubergren no sea takimata sanctus est Lorom ipsum dolor si ment. Lorem ipsum odor si arnet, conseletur sadipscing elltr, sed diam nonumy eirmod tempo sacipeoing end, sed diamonitry enhouse ter co-invidunt ut labore et don'non magna affert un a -sed diam voluptus. At vero eos et accusar i ju-duo dolores et es rebum. Stet citta kasti gubergrei-no sea takimata sanctus est Lorem ipsum dolor sit

.J." said 1. and I don't much care; for it seems to n a little damaged in the head. But if y zeprain Ahab, of that ship there, the Pequo ou that I know all about the loss of his leg. out it eh sure you do?--all?"

ger pointed and eye levelled at the Pequ e stranger stood a moment, as if in a troubled ing a little, turned and said :--- "Ye've shipp nes down on the papers 1 Well, what and what's to be will be ; and then again, after all. Any how, it's all fixed and and some sailors or other must go with as well these as any other men, ve. shipmates, morning; the ineffable hear

as were interested in the interfailed field of the interfailed field field interfailed field field interfailed field field interfailed field field field interfailed field f

income. Now Bilad, I am sorry to say, had the reputation of being an incorrigible old humks, and in his sea going days, a bitter, hard task-master. They wild use in N-statickt, though it or time Stachen's Beal. A trungue openog nace oversa some bows of the ship, so that the insider commanded a complete time forward.

tores or that ships so that the number determines the format. And had accoseded in this queue tearment, I at length found one who Iv his aspect accound to have authority; and who, it being mon, and the ahip's work supposed, was now withying registe from the burdler of command. He areas sended on an old-failshoued oasen chain, wrigging all over with enrices earing; and the bottom of which was formed of a stout interlating or the same chains to still who wight more southweating.

interlating of the same elastic suff of which the wignant was constructed. Three was nothing so very particular, pethops, about the appearance of the delorly man I saw; he was brown and brawny, like most old assume, and hardi'r rolled up in blas pilot-cloth, cut in the Quaker style; only there was a fine and almost microscopie net-work of the minutest wrinklikes interlatings rough his eyes, which must have arises from his constrainal sali-ings in many hard gales, and always looking to windward;—-for this causes the muscles shout the eyes to become pursed to his causes the muscles shout the eyes to become pursed to his door of the total.

the do. "Sr ne tent. ng it be the Captain of the Pequod, what dost thou want of him 1" he demanded.

立分けせるかれる計水二パー(北京香ヤ球目的日 地域率パー伝示トを子位二米表8にき地通り採用子

56100

9作クロタ上況チシサヒ基芸づるばで午負85春げ注図駄いッパ ▶ 胆兆やここでおはなり、ポートというへいとし、 に現21(189出日をンロが用ナウマ特殊くフィンドについ 次切てほぜのすが、ケート利品支部間を広く安全への最 開る国点 ・パ サホモ反ユ曲79 え チェ 説前ひよ鳥東村進は北 聞る国点メケ文持り分御 しばつぼし育2権まきなン残育こ公対ぼ一給密みい舎容立ト広房ノタモ案47港ユ

# • CycleGAN with ResNet:



#### Learning rate 0.0001

Anal Lorem ipsum dolor sit amet, consetetur sadipscing ellr, sed diam nonumv eirmod tempor invidunt ut labore et dolore magna aliquyam erat, sed diam voluptua. At vero esc et accusam et justo duo dolores et ea rebum. Stet dilla kasd gubergren, no sea lakimata sanctus est Lorem ipsum dolorsil amet. Lorem nerum dolor sit amet, avonsetetur adolocing oliti, sed diam nonumy elimod tempor invidunt ut labore et dolore magna, anquyata. erat, sed diam voluptua. At vero esc et accusam et justo dio dolores et e arbum. She tilla kasd gubergren, no soo takimato zancuu, cet Lorem ipsum dolor sit met. Lorem neum outo sit amet, consetetur sadipscing ellir, sed diam nonumy firmod tempor invidunt ut labore et.dolore magna sittijuyan erat sacipscing eitif, sed diatu nanumyreirmad tempor nividunt ut labore et.dolore magna aitfulyne est sed diam voluptus, At vero ess et accurant bi jutato duo dolores et es rebum; Stät čita kasta guargieu, no sea takimata sanctus est Lorem ipsung dolog sit

hdp, said I, when the care; for it seems to m now, and I don't much care; for it seems to m and a hitle damaged in the head. But if y of Captanu-Ahab, of that ship there, the Pequo J. Captanu-Ahab, of that ship there, the Pequo I. box, that I know all about the loss of his leg. abrut its en-sure you do?-all ?"

firger pointed and eye levelled at the Pequ ke stranger, stood a moment, as if in a troubled rules a little, turned and said :- " Xeveshipp airles down on the papers ! . Well, well, what d; | and what's to-be; will be; and then again,

in, after all. Any how, it's all fixed and it and some sailors or other must go with 1 as well these as any other men, God F

11 Bay well wells, morning; the methalole new years years wells wells wells wells wells wells wells and the second sec

Now Bidad, I am sorry to say, had the reputation of being an incorrigible of bunks, and in his searging days, a bliter, hard task-mate. They add main. Nanadetet though I ever times Sachers Bead. A rungurgue oppongunce constraints bows of runzahly, so that the/naider commanded a complete view forward.

amo statem's from. A runninguar opening more source and being of riterability to that the rules communido a complete view forward. And half concealed in this queer tenennest, I at length found-one who by his mapet second to have and the rule of the "explose found the abdy's not support. He was easily found explose found the abdy's not support. He was easily on a state found the bottom of winging all over with carious models, and the bottom of winging all over with carious producting of the same eakies staff of which the wigners was constructed. The bottom of winging all over with carious producting of the bottom of winging all over with carious producting of the same eakies staff of which the wigners was different to bottom of winging all over with carious producting of the bottom of winging all over with carious producting of the same eakies staff of which the wigners was different to bottom of winging all over the staff of the production of the bottom of winging all over the staff of the production of the same and haven't product in the and half over the bottom of the same from the control all all length in mary hand gives and fliving holes are been approxed of the staff of the marks and the regular wing the staff of the more forth of the same of the same from the control all all for this caused the more of staff of the staff of the staff. The staff of the the control of the same of the staff of the more forth of the staff of the regular of the staff of the st

$$\label{eq:states} \begin{split} & (1+2) + (1+2)$$

作クロタ上況テシンドにませるばった「食い素料「注図数い」の 加北マートルのなりた中で、知道、小学をの止点は用花がしてはた でのないの目です。そのの目的など、「かやの」の原花がしてはた 線行りた中小で大加度っ七字で適力に加ま開催ます。 いる用中が公開協会かっ指指行されなり10世界であり

DIEじ育2福まぎなジ残筲C公対ぼ一給密みい含容立ト広房ノタモ来47港ユ

#### Learning rate 0.0002

Апа

Lorem ipsum doior sit amot, consetetur sadipscing ellir, sed diam nonumv elimiod lemicor invidunt ut labore et dolore magna aliquyam erat. sed diam voluptia. At vero eso et accusam et justo duo dolores et ea rebum. Site cilita kasti gueergren, no sea takimata sanctus est Lorem ipsum dolorsit amet Lorem ipsum dolor sit amet aposetetur sadipscing ellir, sed diam nonumy erimot tampor invidunt ut labore et dolore magna elliquyate entity sed diam voluptia. At vero eso et accusau ef justo duo dolores et ea rebum. Ellir sad gubergren, no sco takimato sancto set Lorem inoum dolor sit amet. Lorem inoum dolor sit metic, consellaritor sadipscing ellir, sed diam nonum - munet invidunt ut labore et dolore magne atri cod diam voluptua. At vera c.c.o.c. atri duo dolores et es rebum. Ster cira kas no sea takimata sanctus est lorem ipen ama

or, and I don't much care; for it seems to T or, and I don't much care; for it seems to T be a little damaged in the head. But if y Captoin-Ahab, of that ship there, the Pequo Captoin-Ahab, or this bout the loss of his leg. You's that I know all about the loss of his leg.

ger pointed and eye levelled at the Pequ stranger.stood a moment, as if in a troubled ing a little, turned and said :-- " Yeve shipp es down on the papers ! Well, well, what's and what's to-be, will be; and then again, after all. Any how, it's all fixed and and some sailors or other must go with as well these as any other men, ve. shipmates, morning; the ineffable hea

Where shipmates, morning ; the ineffable here sufficient of the immutibations, was then some lack of commo-consumbary bound worth Cyclin Poler. Thoogh relation, from consumer that worth Cyclin Poler. Thoogh relation, and hough arrays here here areas against land meaders, yet the straight of the second second second second bound in the source of the second second second consumer that we have a second second second second here the second bound second second second second second and these there are the second second

neome. Now Bildad, I am sorry to say, had the reputation of being an incorrigible old hunks, and in his seergoing days, a bitter, and task-master. They told me in Nanueles, hough it cer-

an incorrichle old hunks and in his energoing days, a hit hard taok-master. They takk main hards the hold hards and have statements. They takk main hards the hold hards have dotteships in that the fraided communical a strapping war forward. And half-concretely in this quest examend, it at length found one widdly the mapsets second to have authority and who, it follows and hards and the hards and the strapping of the strapping of the strapping of commonds, was now enjoying main hards hards a concerned. In the second of the strapping of the forward wider was formed of a statu-te paradoxid hards have and which the strappened of a forward hard the forten one wider was formed of a statu-te of the strapping of the hold hards of the hold hards of the forward hards of the status of the hold hards of the hold hards of the forten set on ching as set participant, for hard-status the

for large the same classic start of which the waymen was constructed. There was nothing so-very particular, being possible to the parameter of the delayer, start in the sea Aroon and bravery, this most old segment, and head's for the particular possible to the search problem in the search problem is the search problem in the search problem in the search problem in the search problem is the search problem in the search problem in the search problem is the search problem in the search problem in the search problem is the search problem in the search problem in the search problem is the search problem in the search problem in the search problem is the search problem in the se





#### Learning rate 0.0001

Loram lipsum dolar sil amet, conseletur sadipacing elilir, sed diam nonumv elimod lempor invidunt li aldone el dolare magna alquyam erat, sed diam voluptus. Al vero eos el accusam el justo du sed las elementas de la consenta la consenta a sea la come presima dolar da la kasa gubergren, no sea la kimata sancius esi Lorem insum doloral. amet i noram justim dolora itamet, annestenuror mividunt ul labore el dolare magna, auquyata-erat, o intro noincre e la magna, auquyata-erat, o intro noincre e la magna, auquyata-erat, o intro noincre e la magna, auquiata-erat, o intro noincre e la magna, auquiata-erat, o intro noincre el magna, auquiata-erat, o intro noincre el adolare magna, auquiata-erat, o intro noincre el adolare magna, alquiata-erat, o intro noincre el adolare magna alterarenta famo sadipacing elitir, sed dian-aragina alterarenta famo cod diam voluptus. Al vero cos el accusar: juitato ou duros esi e reloum. Silo illa kasad Guergreu, no sea takimata sancius esi l orem iostun dola sil-

nui, said I, when here are; for it seems to n oy, and I don't much care; for it seems to n by a little damaged in the head. But if y of Apton-Alab, of that ship there, the Pequo is out that I know, all about the loss of his leg. and," said 1 dut it, eh-sure/you do?-all ?"

firger pointed and eye levelled at the Pequ g stranger stood a moment, as if in a troubled ing a little, turned and said :-- "Yeveshipp ardes down on the papers ! 'Yell,-well, what's 1 : and what's to be, will be - and then again, r, after all. Any how, it's all fixed and and some sailors or other must go with as well these as any other men, to ve, shipmates, morning; the ineffable hea

3) a set set and the set of th

some. Now Bildad, I am sorry to say, had the reputation of being incorrigible old hunks, and in his sea-going days, a bitter, ncorrigibie oid numes, and in nis scargoing only, a trak-mastes. <u>They teld num in Nantuelest</u> though 6 Sachem's head. <u>A synagounar opening need owners</u> of whiteshipf: 56 that the sinsider: commanded a offupl though it cer

and the start index of the start of the star

e tent. g if be the Captain of the Pequod, what don't thou

- 10 geo sucurititationesta invationen 1946 - A

17 H 8 1 2 2

作クロタ上況テシカビニ基づるばで「1月35春げ注図駄いッ F開発やし、これはAFFを作ったのか、ことできましたので構成である。 一世の行動の目中でににお用すやなり構成とすった行う、私なりしたのであまし、 の法にはため、ため構成の問題であるが中心での最新に示認れてれていた。 1.はつぼじ首2権まぎなジ残省に公対低一給密みい含容立ト広房ノタモデ47港ユ

#### Learning rate 0.0002

Lorem losum dolor sit amat, consetetur sadipscina alfir, sed diam nonumy erimod lemoor invidunt ul abore el dolore magna aliquyem erat, sed riam notusuta, -ri vero eos e i accusam el tusto ou dolores et es reutum, site di la kevi oubergran. •\* takimata sanctus est Lorem losum dolor,sei men, ornan i cumi dolor si amati i unsetetur uadiocona eutr, sod diam nonumy ofm. - J temos, mindunt ul labore el dolore magna antorum est diam volubita. Al vero eos elvacousam el tusto dina dolorse el dolore magna antorum est diam volubita, al vero ess elvacousam el tusto dina dolorse el archum "Stat etta?kased nuterorise antel, corren risum i al atteit, couser-au are el la tusto. antel, corren risum i atteito dolor se antel, corren risum e di atteito coste-en are el la tusto. indian in habore distance majne en application of the state of the set of th i jia

, and I don't much care; for it se nd," said I oms to D a little damaged in the head. But it y Captau Ahab, of that ship there, the Pequo hour that I know all about the loss of his leg. att in eh-sure you do?\_all ?"

lger pointed and eve levelled at the istranger stood a moment, as it in a troubled is a little, turned and said :- " Ye've shipp ies down on the papers? 'Vell, well, what and what's to be, will be: and then again,

after all. Any how, it's all tred and and some sailors or other must go with as well these as any other men, 'rod P to ve, shipmates, morning; the meffable hea

1. By well the set of the set

come. Now Bildad, I am sorry to say, had the reputation of being i incorrigible old hunks, and in his sea-going days, a bitter. an moorngoble old hunns, and in me seargoing only, a hard task-mastes. <u>Ther told uso in Nantuelett</u>-though times exclem's head. A thanguan oneming accel owards bows of that ship, so that the noticer commanded a compl igh it cei

bees afraheably, s0 that the answers. And half concealed in this queer tensement. I at length found way who are in assessed seemed to have anthority; and who, it which are assessed as the assessment of the second second an odd-fabload seam chair, wrigging it over a state arring; and the focus of similar and the second second intellation of the same chaits used of which the significant are constructed.

indications: out the same classifies starf of which the wignmin was constructed. "Ghere was nothing so very markediant, behards-object the appearance of the blochry must rake an above, and braway, the most old seamen, and heavily reflexiva in blac-dimost microscopic net-work of the mititate variants interlation around his space. Which must have around from this south length a money have block and the search of the south length as a space of the south search of the south length of prime search of the space of the south search of the south length of the south of the south search of the south search of the space of the south search of the south search of the space of the south search of the south search of the search of the Popular in said 1, advanced the search of the the manufacture of the Popular was and the south of the search of the terms of the the Popular in the Popular in the search of the search of the terms of the search of the Popular in the the classion of the search of t

want of him 17 he demanded.

The Trans of the and the second firsts الاندر بناید. - «الاندر بناید : - «ا = د ...د∆رای آواد. 

 $\label{eq:second} \begin{array}{ccc} & \mbox{constraints} & \$ ا بطر ما<sup>رد</sup> (ارد اورامی ا

作クロタ上況ラ.。 づるは一午台35春げ注図歌いッ 

1.0021-m70000 またして、中国のあたし国の国際ダダンパワガロ 1はつぼし育2権まざなジ铁管に公対ほー給留みい言容立ト広房ノタモ、17港ユ

# **Test Case 6 Results**

#### • Pix2Pix with ResNet:



#### With L1 Loss

vitae, dignissim lacus. Donec lectus leo, tempor at la on laoreet sapien gravida a. Ut auctor shoncus n dictum, dolor sit amet eleipend lobortis, magna commodo sodales gravida. Integer tincidunt risus lorem id. porttitor orci. Integer porta purus eque, nec placenat lectus est id metus. Monbi et ton eget ipsum at, semper uttricies risus. Phasellus f

Arial Lorem ipsum dolog sit amet, consetetur sadipscing elit, seq dam no, wit eimod tempor, invidunt ut labore et dolore num's aliquyam erat, sed diam voluptius, at ver hagn et accusam et justo duo dolores et areb, sog et accusam et justo duo amet. Logen ipsung est Lorem ipsung dolor sit and Logen ipsung est Lorem and tempor invidmut it labore et dolore magna aliquyam erat, sed diam voluptius at vero eos et accusam et justo duo dolores et as, pours. Est clina kasa dubergren, no sea takimata sengus est Lorem ipsung dolor sit andet. Logen diam nonumy eimod tempor invidmut it labore et dolore magna aliquyam erat, sed diam voluptius at vero eos et accusam et justo duo dolores et as, pours. Est clina kasa dubergren, no sea takimata sengus est Lorem ipsum dolor sit amet. Logen baore et dolore magna aliquyam erat, sed diam, voluptus et dore magna aliquyam erat, sed diam, voluptus et dores en agna aliquyam erat, sed diam, voluptus et dores en agna aliquedren, no sea takimata sengulus est clita kasad gubergren, no sea takimata sengulus est corem ipsum dolor sit amet. 

ther not sleep with your own brother. is, but people like to be private when nd when it comes to sleeping with an strange inn, in a strange town, and that then your objections indefinitely multip earthly reason why I as a sailor should re than anybody else; for sailors no n at sea, than bachelor Kings do ashore sleep together in one apartment, but ammock, and cover yourself with y sleep in your own skin.

I pondered over this harpooneer, th I pondered over this harpooneer, the the theorem of the second over the second over the second readful late, you had better be turning flukes—it's a rise here's plenty room for two to kick about in that bed; it mightly big bed that. Why, afore we give it up, Sal use here's plenty room for two to kick about in that bed; it our Sam and little Johnny in the foot of it. But I g saming and sprawling about one night, and somehow, Sam ched on the floor, and came near breaking his arm. A it sal said it wouldn't do. Come along here, I'll give y n in a jift; " and so saying he lighted a camele and hed ards me, offering to lead the way. But I stood irresolut in looking at a clock in the corner, he exclaimed "I vum i day—you won't see that harpooneer to-night; he's come day—you won't see that harpooneer to night; he's com or somewhere—come along then; do come; won't

e '' considered the matter a moment, and then up stairs y , and I was ushered into a small room, cold as a clam, an shed, sure enough, with a prodigious bed, almost bi gh indeed for any four harpooneers to sleen abrass the statement of the stat igna condimentum lobortis. Quisque impe n molestie urna, vitae vestibulum turpis va ndum varius, velit sapien blandit metus, tincidunt in dapibus nec, vehicula eu dui penatibus et magnis dis parturient mont es est, non rhoncus mauris congue porta. um sociis natoque penatibus et magnis di teger bibendum sagittis ligula, non faucib is et magnis dis parturient montes, nasce accumsan. Cum sociis natoque penatibus lus mus. Vestibulum vitae ipsum nec arcu pellentesque orci. Morbi congue viverra n ommodo, lacus arcu elementum elit, at n

#### Without L1 Loss eu mauris qua

vitae, dignissim lacus. Donec lectus leo, tempor at l on laoreet sapien gravida a. Ut auctor shoncus n dictum, dolor sit amet elexpend lobortis, magna n commodo sodales gravida. Integer tincidunt resus xs lorem id porttitor orci. Integer porta p negue, nec placenat lectus est id metus. Morbi et to eget ipsum at semper uttricies risus. Phasellus fo

#### Aria

Arial Lorem lipsym dolog, st amet. consetetur sadipscing elits, seq gam non, st, weimod tempor, invidunt ut labore et dolore num a lipsymmetry a liguyam erat, sed diam voluptus, at var hadna set accusam et justo duo dolores et aclore statum, stet clita kasa gubergren, no sea takingat sandum, stet clita kasa gubergren, no sea takingat sandum sea. sed diam voluptus ti vero eos et accusam et justo duo dolores et a fugurum. Stet clita kasa gubergren, no sea takingat sandum sea. sed diam voluptus ti vero eos et accusam et justo duo dolores et a fugurum. Stet clita kasa gubergren, no sea takingat sandum, set clita kasa gubergren, no sea takingat sandum, set clita kasa gubergren, no sea takingat sandum, set clita kasa gubergren, no sea takingat sandum set. sed diam voluptus ti vero eos et accusam et justo uo dolores et a fugurum. Stet clita kasa gubergren, no sea takingat sandum set. set clita kasa gubergren, no sea takingat sandum set. set clita kasa gubergren, no sea takingat sandum set. set clita kasa gubergren, set clita kasa gubergren, set clita kasa gubergren, set sandum set. set sandum set. set sandum set sandum set. set sandum set. set sandum set sandum set. set sandum set sandum set. set

......

ther not sleep with your own brother is, but people like to be private when d when it comes to sleeping with an strange inn, in a strange town, and that then your objections indefinitely multip earthly reason why I as a sailor should re than anybody else; for sailors no i at sea, than bachelor Kings do ashor sleep together in one apartment, but ammock, and cover yourself with y leep in your own skin.

I pondered over this harpooneer, th I pondered over this harpooneer, the the theorem of the second second second second second readful late, you had better be turning flukes—it's a mice large second second second second second second second particle second second second second second second rightly big bed that. Why, afore we give it up, Sal use to ur Sam and little Johnsyn in the foot of it. But I g saming and sprawling about one night, and somehow, Sam ched on the floor, and came near breaking his arm. A ty Sal said it wouldn't do. Come along here, I'll give y n in a jifty;" and so saying he lighted a candle and held and second second second second second second and second second second second second second n looking at a clock in the corner, he exclaimed 'I vun for somewhere—come along then; do come; second second

nsidered the matter a moment, and then up stairs considered the matter a moment, and then up states, and I was ushered into a small room, cold as a clam, are shed, sure enough, with a prodigious bed, almost bi gh indeed for any four harponeers to sleen abreast igna condimentum lobortis. Quisque imp molestie urna, vitae vestibulum turpis v ndum varius, velit sapien blandit metus tincidunt in dapibus nec, vehicula eu dui penatibus et magnis dis parturient mon es est, non rhoncus mauris congue porta um sociis natoque penatibus et magnis di eger bibendum sagittis ligula, non faucib is et magnis dis parturient montes, nasce accumsan. Cum sociis natoque nenatibus us mus. Vestibulum vitae ipsum nec arcu pellentesque orci, Morbi congue viverra r ommodo, lacus arcu elementum elit, at n



ommodo, lacus arcu elementum elit, at m

#### With L1 Loss

vitae, dignissim lacus. Donec lectus leo, tempor at la et sapien gravida a llt auctor shoncus m dictum, dolor sit amet eleijend lobortis, magna arcu commodo sodales gravida Integer tincidunt risus v as lonem id porttitor orci. Integer porta p neque, nec placenat lectus est id metus. Morbi et tor

eget ipsum at semper uttricies risus. Phasellus fo

Arial Lorem ligsum dolg, sit amet. consetetur sadipscing elits. seq dolore hurny aliatuyam erat. sed diam voluptua, At vero evod et accusam et justo duo voluptua, At vero evod et accusam et justo duo sea tafkingta sanche et alianet, consetetur sadipscing alitt. seq diam notumy elimod tempor invidunt ut labore et alianet, consetetur sadipscing alitt. seq diam notumy elimod tempor invidunt ut labore et alianet, consetetur sadipscing alitt. seq diam notumy elimod tempor invidunt ut labore et alianet accusam et liupton duo dologes et eta, juus est Lorem ipsum dolor sit sadipscing alitt, seq diam nonumy elimod tempor invidunt ut labore eta alianet, consetetur sadipscing alitt, seq diam nonumy elimod tempor invidunt ut labore eta diam encumy elimod tempor invidunt ut labore eta diam segna aliquyam erat. sed diam voluptus eta diam esaga aliquyam erat. sed diam voluptus eta diam segna aliquyam erat. sed taim voluptus eta diam esaga aliquyatum erat. sed taim voluptus eta al taimat. Consentetur satus eta kutta 'angritus est Lorem ipsum dolor sit amet.

he not sleep with 'on UT OFOLUS is put people il 001/81 0.06 a when it comes is sigeding with an strange inn, in a strange town, and that then your objections indefinitely multip earthly reason why I as a sailor should re than anybody else; for sailors no n a sea than bachelor Kings do ashor sleep together in one apartment, but ammock, and cover yourself with y

icev m your own skin.

pondered over this narpooneer th onind him. It .

the thousant for the prime in the first lite one is a single set of the set o

onsidered the matter a moment, and then up stairs s and I was ushered into a small room, oold as a clam, an shed, sure enough, with a prodigious bed, almost bi gh indeed for any four harpooneers to sleen abroast

gna condimentum lobortis. Quisque imp molestie urna, vitae vestibulum turpis va endum varius, velit sapien blandit metus, tincidunt in dapibus nec, vehicula eu dui e penatibus et magnis dis parturient mon es est, non rhoncus mauris congue porta. um sociis natoque penatibus et magnis di teger bibendum sagittis ligula, non faucib us et magnis dis parturient montes, nasce accumsan. Cum sociis natoque nenatibus lus mus. Vestibulum vitae ipsum nec arcu pellentesque orci. Morbi congue viverra i ommodo, lacus arcu elementum elit, at m

#### Without L1 Loss rabitur eu mauris quae

vitae, dianissim lacus. Donec lectus leo, tempor at l eet sapien gravida a, Ut auctor sho . 1. dictum, dolor sit amet eleifend lobortis, magna arc commodo sodales gravida Integer tincidunt risus i lorem id. portitor orci Integer porta pu reque, nec placenat lectus est id metus. Morbi et ton

eget ipsum at semper uttricies risus. Phasellus fo

#### Aria

Ariai Lorem ibsum dolog at amet, consetetur sadipscing elitr, seq diam non-uny ainduvam erat, sed diam voliptita, At ven tang ainduvam erat, sed diam voliptita, At ven tang ainduvam erat, sed diam voliptita, At ven tang ainduvam erat, sed diam sea takingat senting set Lorem ibsum dolor sit amet. Lorem ibsum dolor sit amet. Lorem ibsum dolor sit adipscing all's adiptication anoumy eimod tempor invidunt ut labore si dolore magna allquvam erat, sed diam, voluptita, ty veno es et accusan et lusto duo dolores et ea repum. Stet cita kasa gubergren, no sed taking taket. Lorem ibsum dolor sit amet. Lorem ibsum dolor sit amet. Lorem ibsum dolor sit adipscing all's adiptication anoumy eimod tempor invidunt ut labore si dolore magna allquvam erat, sed diam, voluptita, ty veno es et accusan et lusto duo dolores et ea repum. Stet cita kasa gubergren, sed diam, voluptita, ty veno es et accusan et lusto duo dolores et ea repum. Stet cita kasa gubergren et alloren invidunt ut labore si dolore magna allquvam erat, sed diam, voluptita, ty veno es et accusan et lusto duo dolores et ea repum. Stet cita kasa gubergren o sea takinata sanglus set Lorem ibsum dolor sit amet. Lorem ibsum dolor sit amet. Lorem ibsum dolor sit amet, sed allan voluptita ty veno es et accusan et lusto duo dolores et ea reputs et lorem ibsum dolor sit amet.

her not sleep with

is put neople like w

a when it comes to steeping with an strange inn, in-a-strange town, and that then your objections indefinitely multi earthly reason why I as a sailor should e than anybody else; for sailors no n at sea, than backelor Kings do ashor ieen weether in one apartment, but mmock, and cover syourself with y leen m vour own skin.

onaered over this harpooneer. th

Hearth Law you had better be turning flukes-it's get weathul late, you had better be turning flukes-it's a nice all and me slept in that ere bed the night we were spl had be slept in your for two to kick about in that bed ; it All and me slept in that ere bed the night we were sp here's plenty room for two to kick about in that bed; it mighty big bed that. Why, afore we give it up, Sal use t our Sam and little Johnny in the foot of it. But I g our Sam and little Jonnny in the toot of it. But 1 g uming and sprawling about one night, and somehow, Sam hed on the floor, and came near breaking his arm. A Sal said it wouldn't do. Come along here, I'll give y i, Sai said it wouldn't do. - Come along here, I'll give y n in a jiffy ?'\_and so saying he lighted a candle and hel ards me, offering to lead the way. But I stood irresolu n looking at a clock in the corner, he exclaimed "I uum day—you won't see that harpooneer to-night; he's come or somewhere—come along then; do come; won't g?"

nsidered the matter a moment, and then up stairs consucera an inter a moment, and anen up stains , and I was ushered into a small room, cold as a clam, an shed, sure enough, with a prodigious bed, almost bi gh indeed for any four harpooneers to sleen abreast

igna condimentum lobortis. Quisque imp molestie urna, vitae vestibulum turpis v endum varius, velit sapien blandit metus, tincidunt in dapibus nec, vehicula eu dui penatibus et magnis dis parturient mon es est, non rhoncus mauris congue porta um sociis natoque penatibus et magnis di teger bibendum sagittis ligula, non faucib is et magnis dis parturient montes, nasce accumsan. Cum sociis natoque penatibus lus mus. Vestibulum vitae ipsum nec arcu ellentesque orci. Morbi congue viverra r ommodo, lacus arcu elementum elit, at m

# C. Code Appendix

All the codes presented in the thesis appendix have undergone a refactoring process to enhance their readability. The aim of this was to ensure that any reader, regardless of their programming background, can easily follow and understand the code.

# **Pix2Pix Implementation**

# Pix2Pix Generator (U-Net)

```
class Generator(nn.Module):
1
     def __init__(self, in_channels, out_channels):
2
3
       super().__init__()
       # encoder
4
5
       encoder_layers = [
6
         # The first downsample-block with the specified parameters (128x128x64)
7
         Downsample_Block_1(in_channels=3,out_channels=64,kernel_size=4,stride=2,
         padding=1),
8
         # The second downsample-block with the specified parameters (64x64x128)
9
         Downsample_Block_2(in_channels=64,out_channels=128,kernel_size=4,
10
         stride=2,padding=1),
11
12
         # The third downsample-block with the specified parameters (32x32x256)
         Downsample_Block_3(in_channels=128,out_channels=256,kernel_size=4,
13
14
         stride=2,padding=1),
         # The fourth downsample-block with the specified parameters (16x16x512)
15
         Downsample_Block_4(in_channels=256,out_channels=512,kernel_size=4,
16
         stride=2,padding=1),
17
         # The fifth downsample-block with the specified parameters (8x8x512)
18
19
         Downsample_Block_5(in_channels=512,out_channels=512,kernel_size=4,
         stride=2,padding=1),
20
         # The sixth downsample-block with the specified parameters (4x4x512)
21
         Downsample_Block_6(in_channels=512,out_channels=512,kernel_size=4,
22
23
         stride=2,padding=1),
         # The seventh downsample-block with the specified parameters (2x2x512)
24
25
         Downsample_Block_7(in_channels=512,out_channels=512,kernel_size=4,
         stride=2,padding=1),
26
         # The eighth downsample-block with the specified parameters (1x1x512)
27
         Downsample_Block_7(in_channels=512,out_channels=512,kernel_size=4,
28
         stride=2,padding=1)]
29
```

```
# decoder
30
       decoder_layers = [
31
         # The first upsample-block with the specified parameters (2x2x512)
32
         Upsample_Block_1(in_channels=512,out_channels=512,kernel_size=4,
33
         stride=2, padding=1),
34
         # The second upsample-block with the specified parameters (4x4x512)
35
         Upsample_Block_2(in_channels=1024,out_channels=512,kernel_size=4,
36
37
         stride=2, padding=1),
         # The third upsample-block with the specified parameters (8x8x512)
38
         Upsample_Block_3(in_channels=1024,out_channels=512,kernel_size=4,
39
         stride=2, padding=1),
40
         # The fourth upsample-block with the specified parameters (16x16x512)
41
         Upsample_Block_4(in_channels=1024,out_channels=512,kernel_size=4,
42
         stride=2, padding=1),
43
         # The fifth upsample-block with the specified parameters (32x32x256)
44
         Upsample_Block_5(in_channels=1024, out_channels=256,kernel_size=4,
45
         stride=2, padding=1),
46
         # The sixth upsample-block with the specified parameters (64x64x128)
47
48
         Upsample_Block_6(in_channels=512,out_channels=128,kernel_size=4,
         stride=2, padding=1),
49
         # The seventh upsample-block with the specified parameters (128x128x64)
50
51
         Upsample_Block_7(in_channels=256,out_channels=64,kernel_size=4,
         stride=2, padding=1),
52
         # The eighth upsample-block with the specified parameters (256x256x3)
53
54
         Upsample_Block_8(in_channels=64, out_channels=3,kernel_size=4,
         stride=2, padding=1)]
55
56
       self.encoder = nn.ModuleList(*encoder_layers)
57
       self.decoder = nn.ModuleList(*decoder_layers)
58
59
     def forward(self, x):
60
       skips_cons = []
61
       for encoder_block in self.encoder:
62
         x = encoder_block(x)
63
         skips_cons.append(x)
64
       skips_cons = list(reversed(skips_cons[:-1]))
65
       decoders = self.decoder[:-1]
66
       for decoder_block, skip in zip(decoders, skips_cons):
67
         x = decoder_block(x)
68
         x = torch.cat((x, skip), axis=1)
69
70
       return x
```

Listing C.1: Pix2Pix generator

Note that both downsample-block and upsample-block are detailed in section 4.2.4.1.

# **Pix2Pix Discriminator**

```
1 class Discriminator(nn.Module):
2
3
     def __init__(self, in_channels, out_channels, n_blocks):
4
       super().__init__()
       discriminator_layers = []
5
6
       discriminator_layers.extend([
7
         # The first block with the specified parameters
8
9
         nn.Conv2d(in_channels=6,out_channels=64,kernel_size=4,stride=2,
         padding=1), nn.LeakyReLU(0.2, True),
10
         # The second block with the specified parameters
11
         nn.Conv2d(in_channels=64,out_channels=128,kernel_size=4,stride=2,
12
         padding=1), nn.BatchNorm2d(128), nn.LeakyReLU(0.2, True),
13
         # The thierd block with the specified parameters
14
15
         nn.Conv2d(in_channels=128,out_channels=256,kernel_size=4,stride=2,
         padding=1), nn.BatchNorm2d(256), nn.LeakyReLU(0.2, True),
16
         # The fourth block with the specified parameters
17
         nn.Conv2d(in_channels=256,out_channels=512,kernel_size=4,stride=1,
18
         padding=1), nn.BatchNorm2d(512), nn.LeakyReLU(0.2, True),
19
         # The final block with the specified parameters
20
21
         nn.Conv2d(in_channels=512,out_channels=1,kernel_size=4,stride=1,
         padding=1)
22
       ])
23
24
25
       self.model = nn.Sequential(*discriminator_layers)
26
     def forward(self, input, target):
27
       x = torch.cat([input, target], axis=1)
28
       return self.model(x)
29
30
```

Listing C.2: Pix2Pix discriminator layers

# **CycleGAN Implementation**

**CycleGAN Generator (ResNet)** 

```
1 class Generator(nn.Module):
2
     def __init__(self, in_channels, out_channels, n_blocks):
3
       super().__init__()
4
5
       generator_layers = []
       # Add the encoder layers
6
       generator_layers.extend([
7
         # The first downsample-block with the specified parameters
8
         nn.ReflectionPad2d(3),
9
         nn.Conv2d(in_channels=3,out_channels=64,kernel_size=7,stride=1,
10
         padding=0), nn.InstanceNorm2d(64), nn.ReLU(True),
11
12
         # The second downsample-block with the specified parameters
         nn.Conv2d(in_channels=64,out_channels=128,kernel_size=3,stride=2,
13
         padding=1), nn.InstanceNorm2d(128), nn.ReLU(True),
14
         # The thierd downsample-block with the specified parameters
15
         nn.Conv2d(in_channels=128,out_channels=256,kernel_size=3,stride=2,
16
17
         padding=1), nn.InstanceNorm2d(256), nn.ReLU(True)])
18
19
       # The residual blocks
20
       for i in range(n_blocks):
         generator_layers.append(ResidualBlock(256,))
21
22
       # The decoder layers
23
       generator_layers.extend([
24
         # The first upsample-block with the specified parameters
25
         nn.ConvTranspose2d(in_channels=256,out_channels=128,kernel_size=3,
26
27
         stride=2,padding=1), nn.InstanceNorm2d(128), nn.ReLU(True),
         # The second upsample-block with the specified parameters
28
         nn.ConvTranspose2d(in_channels=128,out_channels=64,kernel_size=3,
29
         stride=2,padding=1), nn.InstanceNorm2d(64), nn.ReLU(True),
30
         # The final block with the specified parameters
31
         nn.ReflectionPad2d(3),
32
         nn.Conv2d(in_channels=64,out_channels=3,kernel_size=7,stride=1,
33
         padding=0), nn.Tanh()])
34
35
       self.model = nn.Sequential(*generator_layers)
36
37
     def forward(self, input):
38
39
       return self.model(input)
```

Listing C.3: CycleGAN generator

# **CycleGAN Discriminator**

```
1 class Discriminator(nn.Module):
2
3
     def __init__(self, in_channels, out_channels, n_blocks):
       super().__init__()
4
       discriminator_layers = []
5
6
       discriminator_layers.extend([
7
8
         # The first block with the specified parameters
9
         nn.Conv2d(in_channels=3,out_channels=64,kernel_size=4,stride=2,
10
         padding=1), nn.LeakyReLU(0.2, True),
         # The second block with the specified parameters
11
         nn.Conv2d(in_channels=64,out_channels=128,kernel_size=4,stride=2,
12
13
         padding=1), nn.InstanceNorm2d(128), nn.LeakyReLU(0.2, True),
         # The thierd block with the specified parameters
14
15
         nn.Conv2d(in_channels=128,out_channels=256,kernel_size=4,stride=2,
16
         padding=1), nn.InstanceNorm2d(256), nn.LeakyReLU(0.2, True),
         # The fourth block with the specified parameters
17
         nn.Conv2d(in_channels=256,out_channels=512,kernel_size=4,stride=1,
18
19
         padding=1), nn.InstanceNorm2d(512), nn.LeakyReLU(0.2, True),
         # The final block with the specified parameters
20
         nn.Conv2d(in_channels=512,out_channels=1,kernel_size=4,stride=1,
21
22
         padding=1)
       ])
23
24
       self.model = nn.Sequential(*discriminator_layers)
25
26
     def forward(self, input):
27
       return self.model(input)
28
29
```

Listing C.4: CycleGAN discriminator layers